

Suite 410, 900 Howe Street Vancouver, BC Canada V6Z 2N3 bcuc.com P: 604.660.4700
TF: 1.800.663.1385
F: 604.660.1102

Creative Energy Vancouver Platforms Inc.

2019-2020 Revenue Requirements Application for the Core Steam System and Northeast False Creek Service Areas

Decision and Order G-227-20

September 2, 2020

Before: A. K. Fung, QC, Panel Chair K. A. Keilty, Commissioner E. B. Lockhart, Commissioner

TABLE OF CONTENTS

1.0	Introdu	uction1
	1.1	The Applicant1
	1.2	Key Issues1
	1.3	Organization of the Decision2
	1.4	Background2
	1.5	Legislative Authority
	1.6	Regulatory Process and Participants
2.0	Setting	2019 Rates for the Core Steam System4
	2.1	Background4
	2.2	Creative Energy's Proposal for 2019 Rates6
3.0		2020 Rates for the Core Steam System
5.0	-	
	3.1	2020 Proposed Revenue Requirement16
	3.2	Deferral Accounts27
		3.2.1 Water Cost Deferral Account
		3.2.2 Property Tax Deferral Account
		3.2.3 Third Party Regulatory Costs Deferral Account
		3.2.4 Pension Expense Deferral Account
		3.2.5 DARR Mechanism
	3.3	Proposals regarding the Fuel Switch Study and LTRP Deferral Account
		3.3.1 Background
		3.3.2 Creative Energy's Proposal
		3.3.3 Discussion of the Issues
	3.4	Setting 2020 Steam Rates41
4.0	Setting	2019 Rates for NEFC41
	4.1	NEFC Hot Water Rates41
	4.2	Underreporting of PARQ Hot Water Plant Steam Meter44
5.0	Setting	2020 Rates for NEFC47
	5.1	2020 Hot Water Rates

	5.2	Proposed Fuel Cost Adjustment Charge Rate Rider	.51
6.0	Additio	nal BCUC Directives and Recommendations	.53
	6.1	Historical Pension Accounting Discrepancies	.53
	6.2	Panel Recommendations for Future Filings	.55
7.0	Summa	ary of Approvals and Directives	.57
сомм	ISSION	DRDER G-227-20	

APPENDIX A List of Acronyms APPENDIX B Exhibit List

Executive Summary

On December 19, 2019, Creative Energy Vancouver Platforms Inc. (Creative Energy) filed its 2019-2020 Revenue Requirements Application (RRA) with the British Columbia Utilities Commission (BCUC) seeking approval to establish permanent rates for 2019 and 2020 for its core steam system (Core Steam System) and Northeast False Creek (NEFC) service areas (Application).

Rates for the Core Steam System and NEFC are currently set on an interim basis for 2019 and 2020. For the Core Steam System, rates (Steam Rates) have been established under a traditional cost of service approach to utility rate setting. For the NEFC, rates (Hot Water Rates) have been established at a levelized rate to recover the costs incurred to install the required infrastructure to serve NEFC customer load while recognizing the revenues from the buildout of that load over time.

As part of this Application, Creative Energy has also applied for related approvals in connection with setting rates, including but not limited to, approval to:

- i) Establish or continue certain deferral accounts;
- ii) Establish a rate rider mechanism to recover variances in specified deferral accounts from Core Steam System customers; and
- Charge NEFC customers a Fuel Cost Adjustment Charge Rate Rider. Creative Energy also seeks BCUC approval of requests related to expenses which have been captured in its Fuel Switch Study and Long-Term Resource Plan (LTRP) Deferral Account.

Key issues

With regards to the Core Steam System and NEFC, the Panel identified the following key issues:

- Considering whether the use of 2019 actual results for the proposed revenue requirement is appropriate within the context of the rate setting background for the Core Steam System and NEFC, and how variances recorded in various deferral accounts should be treated.
- Assessing whether the additional steam supplied by the Core Steam System to NEFC at the PARQ Hot Water Plant meter, which was underreporting, should be recognised, and if so, how and when it should be recognised.
- Determining whether the forecast revenue requirements and forecast load and revenues for the Core Steam System and NEFC are reasonable.
- Evaluating whether a Deferral Account Rate Rider (DARR) mechanism is appropriate to recover variances captured in certain deferral accounts for the Core Steam System, and if so, the appropriate level of that rate rider for this test period.
- Reviewing whether the expenses captured in the Fuel Switch Study and LTRP Deferral Account appropriately relate to the Fuel Switch Study and LTRP and determine how much, if anything, should be recovered from Core Steam System customers.
- Reviewing the treatment of NEFC's fuel costs and determining if it is appropriate to recover directly from NEFC customers those amounts approved to be recovered from steam customers through a Fuel Cost Adjustment Charge (FCAC) Rate Rider.

2019 Rates for the Core Steam System

Creative Energy's 2019 interim Steam Rates for the Core Steam System (which are maintained at 2018 Steam Rates) are approved as permanent. In general, the Panel does not agree that it is appropriate to set utility rates based on hindsight, using actual results. Rates should be set prospectively based on a reasonable forecast and the utility should be held accountable for its forecast with the exception of those revenue requirement items that the BCUC approves as being appropriate for deferral treatment. The Panel finds it appropriate to maintain the BCUC-approved 2018 Steam Rates and approved forecast for 2018 for setting 2019 permanent rates, including the continuation of variance treatment for Regulatory Costs and Pension Expense. Based on the BCUC's previous determinations, Creative Energy is also directed to reduce its allowed return for 2019 by \$14,862 as it relates to the After-tax Regulatory Pension Asset Account.

2020 Rates for the Core Steam System

Creative Energy's request to increase Steam Rates by 4.2 percent, effective January 1, 2020, is denied. Instead, the Panel approves 2020 Steam Rate on a permanent basis with various adjustments.

The Panel finds that the inclusion of NEFC in the Massachusetts Formula cost allocation is a practical approach to estimating the NEFC's costs in serving a separate energy system. However, Creative Energy's proposed two-factor Massachusetts Formula is denied. Creative Energy is instead directed to use a three-factor Massachusetts Formula which reflects the industry standard.

The Panel disagrees with Creative Energy's submissions that the proposed DARR mechanism to recover forecast variances in operating and maintenance (O&M) expenses is a "straightforward and accepted means" to recover forecast variances in O&M costs and that it supports transparency. While the Panel agrees that the Fuel Cost Stabilization Account and related rate rider mechanism are a typical and understood approach used to capture and recover fuel cost variances, there is no evidence to support such an approach for pension, regulatory and other recurring non-controllable O&M costs. Since deferral account treatment for the latter group of recurring non-controllable O&M costs is allowed, variances relating to these costs may be recovered in subsequent revenue requirements considering that these applications are filed regularly by Creative Energy.

Creative Energy is directed to reduce its allowed return for 2020 by \$21,503 as it relates to the After-tax Regulatory Pension Asset Account pursuant to previous BCUC determinations.

The Panel finds that variance treatment for third-party regulatory costs and pension expense is consistent with previous BCUC decisions and approves the continuation of the TPRCDA and Pension Expense Deferral Account, respectively. However, the Panel considers that recognizing variances over a one-year period, rather than two years as proposed by Creative Energy, is more consistent with past practice and that 2018 and 2019 pension remeasurement amounts should be offset.

The Panel also approves the establishment of a non-rate base Water Cost Deferral Account to capture water cost forecast variances for the 2020 Test Year only and a non-rate base Property Tax Deferral Account to capture property tax forecast variances, as these were demonstrated to be primarily outside of the utility's control and have been difficult to forecast.

Finally, related to expenses which have been captured in a Fuel Switch Study and LTRP Deferral Account, the Panel approves for Creative Energy to write off \$103,536 from the balance in the Fuel Switch Study and LTRP Deferral Account as proposed by Creative Energy. However, the Panel denies Creative Energy's request for approval to recover from customers of the Core Steam System the \$214,185 spent in 2016 on enabling low-carbon development and to maintain a balance of \$417,502 in the deferral account until such time as a

Certificate of Public Convenience and Necessity for a low-carbon energy project is submitted and an assessment of how much of the balance can be capitalized can be made. The Panel finds that it would be inappropriate to allow Creative Energy to recover these costs from current or future ratepayers as this would amount to retroactive ratemaking. Instead, the Panel directs that they be written off from the balance in the Fuel Switch Study and LTRP Deferral Account and the deferral account be closed thereafter.

2019 Rates for NEFC

Creative Energy's 2019 interim Hot Water Rates for the NEFC are approved as permanent, effective January 1, 2019. The Panel understands these rates align with the BCUC's previous decisions, in which levelized rates were set to remain in place until commencing recovery of the RDDA in 2020. However, the Panel disagrees with the use of actuals to determine the amount of variances to be recognized in the RDDA and Variance Deferral Account for 2019, as proposed by Creative Energy. Creative Energy is directed to recalculate the allocation of variances between the approved deferral accounts based on the 2017 approved forecast.

With respect to underreported steam supplied by the Core Steam System to NEFC at the PARQ Hot Water Plant meter, the Panel finds that existing mechanisms apply to variances between forecast and actual Steam Service Rates and Fuel Cost charged to NEFC. The Panel notes it is Creative Energy's preference that NEFC's customers are not charged for the amount related to underreported steam services due to the 'controversial impact' of a large one-time surcharge on customers. The Panel acknowledges these concerns and finds Creative Energy's proposal to address the matter prospectively in 2020 by recognizing additional steam in the 2020 steam load forecast and to make no adjustments for 2019 to be reasonable for the underreported steam service. However, Creative Energy is directed to account for the NEFC underreported fuel cost charges in the Core Steam System's Fuel Cost Stabilization Account and to also record the additional fuel costs in the NEFC Variance Deferral Account.

2020 Rates for NEFC

Creative Energy's request to increase its 2020 Hot Water Rates in the NEFC service area by 3.7 percent, is denied. Instead, the Panel approves increases to Hot Water Rates by 10 percent, effective January 1, 2020, because the balance in the RDDA now exceeds approximately \$1.4 million. Creative Energy is also directed to:

- Recalculate its Massachusetts Formula-allocated costs to the Core Steam System and NEFC, respectively; and
- Continue utilization of its RDDA and Variance Deferral Account.

The Panel acknowledges Creative Energy's submission that it currently does not have sufficient information to determine the longer-term levelized rate increases needed to recover the RDDA over a reasonable time frame and accepts deferring consideration of the recovery of the RDDA until 2021, as proposed by Creative Energy. Until a comprehensive rate design application is submitted, the Panel finds it appropriate to maintain the existing rate structure. Accordingly, the Panel approves the continuation of the RDDA and Variance Deferral Account for 2020 and denies the request for the proposed NEFC FCAC Rate Rider. Given the Panel's rejection of the NEFC FCAC Rate Rider, which would have otherwise resulted in an overall rate impact of a 23.7 percent rate increase if approved, the Panel considers that an appropriate rate increase for 2020 is 10 percent. This level of increase will reduce the addition to the RDDA while keeping the rate below the level typically associated with rate shock.

1.0 Introduction

On December 19, 2019, Creative Energy Vancouver Platforms Inc. (Creative Energy) filed with the British Columbia Utilities Commission (BCUC) a 2019-2020 Revenue Requirements Application (RRA) for the core steam system (Core Steam System) and Northeast False Creek (NEFC) service areas (Application). This Application seeks approval from the BCUC to establish permanent rates for 2019 and 2020 for Creative Energy's Core Steam and NEFC system service areas.

This document sets out the key issues to be decided by the Panel, provides an overview of the relevant evidence, considers the positions of the parties and outlines the reasons for the Panel's decision (Decision).

1.1 The Applicant

Creative Energy is a thermal energy utility that uses gas boilers to produce steam, which is distributed to individual buildings through an underground network of distribution pipes. Creative Energy's Core Steam System serves approximately 215 customer buildings¹ in downtown Vancouver.

The Core Steam System also supplies thermal energy to Creative Energy's NEFC hot water system, which serves four buildings in the NEFC neighbourhood of Vancouver. The NEFC system receives steam from the Core Steam System to produce hot water. However, Creative Energy treats steam customers and hot water customers as two separate classes of service (Steam Service and NEFC Service).² The NEFC system is thus both a customer of the Core Steam System and a separate service area with its own BCUC-approved revenue requirements and rates for hot water service.³

Creative Energy is a wholly-owned subsidiary of Creative Energy Canada Platforms Corp. (Creative Energy Canada). In 2014, Creative Energy Canada acquired all the shares of Central Heat Distribution Ltd. (CHDL) with BCUC approval and renamed the company Creative Energy Vancouver Platforms Inc.⁴

Creative Energy also owns and operates several Thermal Energy Systems (TES) that are not connected to the Core Steam or NEFC systems, some of which are registered as Stream A utilities under the BCUC's TES Regulatory Framework Guidelines⁵ and some of which are regulated as Stream B TES.

1.2 Key Issues

The Panel has identified the following key issues in relation to both the Core Steam System and NEFC:

- Considering whether the use of 2019 actual results for the proposed revenue requirement is appropriate within the context of the rate setting background for the Core Steam System and NEFC, and how variances recorded in various deferral accounts should be treated.
- Assessing whether the additional steam supplied by the Core Steam System to NEFC at the PARQ Hot
 Water Plant meter, which was underreporting, should be recognised, and, if so, how and when it should
 be recognised.

¹ Retrieved from: <u>http://creative.energy/about</u> and <u>https://instaragf.com/2018/news/metro-vancouvers-vibrant-district-energy-market-</u> 2/.

 <sup>2/.
 &</sup>lt;sup>2</sup> Creative Energy 2016-2017 Revenue Requirements Application and Rate Design for NEFC Hot Water Service Decision and Order G-167-16 dated November 18, 2016 (2016-2017 Core Steam System and NEFC Decision), p. 1.

³ Exhibit B-1, Application, p. 14.

⁴ Exhibit B-1, p. 14.

⁵ Established by Order G-127-14 and revised by Order G-27-15.

- Determining whether the forecast revenue requirements and forecast load and revenues for the Core Steam System and NEFC are reasonable.
- Evaluating whether a Deferral Account Rate Rider (DARR) mechanism is appropriate to recover variances captured in certain deferral accounts for the Core Steam System, and if so, what level (DARR rate) should be set for this test period.
- Reviewing whether the expenses captured in the Fuel Switch Study and long-term resource plan (LTRP) Deferral Account appropriately relate to the Fuel Switch Study and LTRP and determine how much, if anything, should be recovered from Core steam customers.
- Reviewing the treatment of NEFC's fuel costs and determining if it is appropriate to recover directly from NEFC customers those amounts approved to be recovered from steam customers through a FCAC Rate Rider.

1.3 Organization of the Decision

The remainder of this Decision provides the Panel's approvals and determinations with respect to Creative Energy's revenue requirements and establishes permanent rates for the Core Steam System (Steam Rates) and NEFC (Hot Water Rates) service areas for the years 2019 and 2020.

Section 2.0 of this Decision addresses issues related to setting 2019 rates for the Core Steam System.

Section 3.0 examines the issues related to setting 2020 rates for the Core Steam System, including steam load, operating and maintenance (O&M) expenses, capital additions, the After-tax Regulatory Pension Asset account and related returns, deferral accounts and proposals regarding the Fuel Switch Study and LTRP Deferral Account.

Section 4.0 focuses on the issues related to setting 2019 rates for NEFC, including the underreporting of the PARQ Hot Water Plant steam meter.

Section 5.0 addresses issues related to setting 2020 rates for NEFC, including hot water load, O&M expenses and the proposed FCAC rate rider.

Lastly, section 6.0 discusses additional BCUC directives and recommendations, including historical pension accounting discrepancies and Panel recommendations for future filings.

Given that Creative Energy's proposals in this Application include differing approaches to setting rates for each year of the test period, we have organized this Decision by test year (i.e., 2019 or 2020) and service area (i.e. Core Steam System or NEFC) rather than by key issues collectively for both service areas and the entire test period.

1.4 Background

The BCUC has historically established Creative Energy's Steam Rates according to a traditional cost of service approach to utility rate setting for a considerable time.⁶ In 2017, Creative Energy applied for a multi-year index based ratemaking mechanism for the 2018-2022 period. After an 11-month long proceeding, the BCUC denied Creative Energy's multi-year application on the basis that Creative Energy's proposed mechanism had not been adequately designed and sufficiently thought through to persuade the BCUC that approval would be in the interests of its ratepayers. Instead, the BCUC approved permanent 2018 Steam Rates.⁷

⁶ 2016-2017 Core Steam System and NEFC Decision, p. 9.

⁷ Creative Energy 2018-2022 Revenue Requirements Application Decision and Order G-205-18 dated October 25, 2018 (2018-2022 Core Steam System Decision), pp. 16, 21.

On December 14, 2018, Creative Energy applied to the BCUC for approval to set interim 2019 Steam Rates. In that application, Creative Energy requested approval to maintain Steam Rates at existing 2018 rates on an interim basis, effective January 1, 2019. The BCUC approved Creative Energy's proposal, with the expectation that Creative Energy was to file a permanent rate application shortly thereafter.⁸

The 2016-2017 Core Steam System and NEFC Decision⁹ established the revenue requirements, rate design, and rates for the NEFC service area. This rate and rate design approval followed the BCUC's approval of a Certificate of Public Necessity and Convenience for Creative Energy to serve new developments in the NEFC neighbourhood of the City of Vancouver.¹⁰ NEFC rates have been established on a levelized rate design basis with the Revenue Deficiency Deferral Account (RDDA) in place to record the impact of timing differences between the costs incurred to install the required infrastructure to serve hot water load, and the revenues from the buildout of customer load over time.¹¹

1.5 Legislative Authority

This Application is filed by Creative Energy, and is reviewed by the BCUC, pursuant to sections 58 to 60 of the *Utilities Commission Act* (UCA). In particular, section 59(5) of the UCA defines what is an "unjust" or "unreasonable" rate while section 59(4) states that the determination of what is "unjust" or "unreasonable" is a question of fact of which the BCUC is the sole judge. Section 60 provides the BCUC the authority to establish rates and includes mandatory considerations, including the requirement that rates not be "unjust, unreasonable, unduly discriminatory or unduly preferential."

Section 60(1)(b.1) of the UCA states that in setting a rate, the BCUC may use "any mechanism, formula or other method of setting the rate that it considers advisable, and may order that the rate derived from such a mechanism, formula or other method is to remain in effect for a specified period." The Panel conducts its review of this Application pursuant to this legislative authority.

1.6 Regulatory Process and Participants

By Order G-7-20A dated January 15, 2020, the BCUC established, among other things, interim rates for the Core Steam System and NEFC, a DARR of \$0.29 per thousand pounds of steam for Core Steam System customers and a FCAC Rate Rider of \$16.15 per megawatt hour (MWh) to NEFC customers, effective January 1, 2020, on an interim and refundable basis.

Order G-7-20A also established a regulatory timetable for the review of the Application, which included Creative Energy filing an evidentiary update and a proposal for the potential recovery of the balance in its Fuel Switch Study and LTRP Deferral Account, two rounds of information requests (IRs) by the BCUC and interveners, written final arguments and a written reply argument from Creative Energy.

The Commercial Energy Consumers Association of British Columbia (the CEC) registered and participated in the proceeding. The Panel also acknowledges FortisBC Alternative Energy Services Inc. and FortisBC Energy Inc. which registered as an intervener and interested party in the proceeding, respectively.

The regulatory timetable was amended by Orders G-29-20 and G-103-20 dated February 21, 2020 and May 1, 2020, respectively, due to two extension requests by Creative Energy.

⁸ Order G-248-18.

⁹ Creative Energy 2016-2017 Revenue Requirements Application and Rate Design for NEFC Hot Water Service Decision and Order G-167-16 dated November 18, 2016 (2016-2017 Core Steam System and NEFC Decision)

¹⁰ Order C-1-19. ¹¹ Creative Energy Final Argument, p. 25.

¹¹ Creative Energy Final Argument, p. 25.

By letter dated May 12, 2020, the Panel noted several corrections to the evidence resulting from Creative Energy's responses to IRs, impacting various financial schedules for Core and NEFC, and requested Creative Energy to provide a summary of:

- 1. Approvals sought for each of the Core and NEFC service areas; and
- 2. The under-reported steam consumption at the PARQ hot water plant steam meter, including: the cause(s) of the under-reporting; impacts of the under-reported steam on the Core Steam System and NEFC's 2019 revenue deficiency/surplus; and Creative Energy's positions with respect to alternatives.

In addition to any other submissions, the Panel invited the parties to discuss the following factors in their final arguments:

- 1. The merits of amending the Massachusetts Formula method for allocating general and administrative costs (e.g. the inclusion of NEFC and revising the number and/or type of factors applied) in this proceeding; and
- 2. Whether a deferral account to capture steam load variances for the Core Steam System may be warranted considering recent variances between forecast and actual load and the ongoing COVID-19 global pandemic.

In accordance with the amended regulatory timetable established by Order G-103-20 the BCUC received written final arguments from the CEC on May 28, 2020 and Creative Energy's written reply argument on June 4, 2020.

2.0 Setting 2019 Rates for the Core Steam System

As context for setting 2019 Steam Rates, the Panel first reviews the rate setting background for the Core Steam System. The Panel then considers Creative Energy's proposals for 2019 Core Steam Rates, including its proposals related to 2018 amounts recorded in the Third Party Regulatory Costs Deferral Account (TPRCDA) and the Pension Expense Deferral Account, and the updates to the After-tax Regulatory Pension Asset Account and related returns.

2.1 Background

As noted above, the BCUC approved interim and refundable rates for the Core Steam System, effective January 1, 2019.¹² In that application for interim rates, Creative Energy proposed to maintain 2019 rates at the approved 2018 rates on an interim basis. While Creative Energy proposed interim rates consistent with the 2018 approved rates, it submitted that the interim revenue requirement for 2019 differed from the 2018 approved revenue requirement. These differences included an inflationary increase of 1.84 percent to certain costs and removal of the amortization expense related to specific deferral accounts. The net impact of the proposed adjustments resulted in a zero percent change in rates.¹³

In its approval of 2019 interim rates, the BCUC found that Creative Energy had not provided adequate evidence to support its requested interim revenue requirement for 2019 and directed Creative Energy to maintain Core Steam Rates at the existing approved rates (2018 approved rates) on an interim basis pending a full rate application for 2019 rates. While the BCUC acknowledged that Creative Energy stated in the interim rates application that it intended to file a full rate application for the Core Steam System in 2019, the BCUC stated that an interim rate application "should generally seek to maintain the status quo unless the utility is able to provide compelling evidence for changes to be made. No such evidence was provided in this case."¹⁴

¹² Order G-248-18.

¹³ BCUC reasons for decision on Creative Energy's Application for 2019 Interim Rates for Core Steam and Northeast False Creek Operations dated December 14, 2018, Order G-248-18, Appendix A, p. 1.

¹⁴ BCUC reasons for decision on Creative Energy's Application for 2019 Interim Rates for Core Steam and Northeast False Creek Operations dated December 14, 2018, Order G-248-18, Appendix A, p. 2.

A summary of the 2018 approved revenue requirement which the BCUC used to set 2018 approved rates is as follows:

	2018 App	roved
Operation and Maintenance Expenses		
Wages and Benefits	Ş	2,962,446
Water-related and Electricity Expenses		776,350
Maintenance (including parts, supplies, safety and vehicles)		487,994
Special Services		
Audit	28,590	
Legal	17,869	
Outside Services	51,205	
Regulatory	135,651	233,315
Other General & Administration, Sales Expense (e.g. insurance,		
office expenses)		285,152
Total Operation and Maintenance Expenses	_	4,745,257
Municipal Taxes		258,887
Non-Operation and Other		
Property Taxes		595,160
Income Tax Expense		275,579
Depreciation Expense		1,028,015
Depreciation Expense Allocated to Non Reg		(706)
Amortization of CIAC		(17,158)
Amortization of Rate Base Deferred Expenses		0
Amortization of Non-Rate Base Deferred Expenses		133,837
Actual/Proposed Return on Rate Base		1,653,882
Total Non-Operation and Other	_	3,668,609
Total Revenue Requirement	Ş	8,672,753

Table 1 – 2018 Approved Revenue Requirement¹⁵

The 2018 approved load forecast used to calculate 2018 approved rates was 1,098,514 thousand pounds of steam.¹⁶ Additionally, in the 2018-2022 Core Steam System Decision, the BCUC approved deferral treatment for forecast variances related to regulatory costs and pension expense for 2018.¹⁷

The BCUC's 2018 approval of forecast variance treatment for regulatory costs was consistent with the determinations made in the Creative Energy 2016-2017 Core Steam System and NEFC Decision in which the BCUC approved the establishment of a Third Party Regulatory Cost Deferral Account (TPRCDA). This variance account was approved with a one-year amortization period and a short-term debt carrying cost for a period of five years, at the end of which, Creative Energy would have to apply to the BCUC for renewal of the TPRCDA.¹⁸

¹⁵ Exhibit B-1-1, Evidentiary Update, Appendix 1, pp. 4–5, Tables 15 and 27 Update; Exhibit B-9, Schedules for RRA Filing – IR Round 2, Core Schedule 1.

¹⁶ Creative Energy 2018-2022 Revenue Requirements Application Decision and Order G-205-18 dated October 25, 2018 (2018-2022 Core Steam System Decision), p. 45.

¹⁷ 2018-2022 Core Steam System Decision, pp. 46, 48–49.

¹⁸ 2016-2017 Core Steam System and NEFC Decision, p. 48.

Regulatory costs include third-party expenses related to regulatory filings and proceedings required under the UCA. The 2018 approved forecast for regulatory costs was \$135,651.¹⁹

The approval of deferral treatment for pension expense forecast variances in 2018 was also consistent with the BCUC's determinations in the Creative Energy 2015-2017 Core Steam System Decision in which the BCUC approved the creation of a Pension Expense Deferral Account to capture the annual variance between forecast pension expenses recovered in rates and pension expenses reported in Creative Energy's financial statements. This variance account was approved with a one-year amortization period and a carrying cost equal to Creative Energy's short-term debt rate.²⁰ Pension expense relates to Creative Energy's pension plan covering all unionized employees, including plant and distribution team employees, non-unionized management and administrative staff that started working for Creative Energy before July 2018. The pension plan was closed to non-unionized staff in 2018. Required pension plan contributions are based on actual evaluations as periodically remeasured.²¹ The 2018 approved forecast for pension expense was \$190,284 (included in Wages and Benefits in Table 1 above).²²

2.2 Creative Energy's Proposal for 2019 Rates

Use of 2019 Actual Results for Proposed Revenue Requirement

Creative Energy states that its proposals for permanent 2019 Core Steam System Rates are based on:

- 12 months of 2019 actual results with respect to load, O&M expenses, municipal access fees and property taxes; and
- 9 months of 2019 actual results plus 3 months of forecast information with respect to plant in service and rate base, which impact depreciation, income taxes and the return on rate base.²³

Using 2019 actual results, Creative Energy proposes to maintain the approved 2019 interim Core Steam Rates on a permanent basis and to add the difference of \$103,241 between its updated 2019 revenue requirement and the 2019 actual revenues collected to the TPRCDA for future recovery.²⁴ Specifically, Creative Energy proposes to recover the amount, among others, through a DARR beginning in 2020. The Panel addresses Creative Energy's proposals with respect to the use of a DARR in subsection 3.2.5. The specific amounts recorded in the respective deferral accounts and the related amortization periods are addressed below.

On the use of 2019 actual results, Creative Energy states:

Although the timing of this filing for final 2019 rates is uncommon in practice before the BCUC, Creative Energy does not consider that review and approval of final 2019 rates presents any extraordinary considerations for the Panel outside of its existing mandate under the UCA to ensure that Creative Energy's ratepayers receive safe, reliable and non-discriminatory energy services at just and reasonable rates, and that shareholders are afforded a reasonable opportunity to earn a fair return on their invested capital.²⁵

Given the timing of the Application, Creative Energy explains that it is not seeking an increase in 2019 rates, which may have otherwise been justified on the basis of recovering its updated 2019 revenue requirement.²⁶ In

¹⁹ Exhibit B-1, p. 48.

²⁰ Creative Energy 2015-2017 Revenue Requirements Application Decision and Order G-98-15 dated June 9, 2015 (2015-2017 Core Steam System Decision), Directive 1(d)(iii).

²¹ Creative Energy Final Argument, p. 11.

²² This amount was determined by adding the following pension amounts (\$101,972 + \$28,641 + \$59,671 = \$190,284) from Exhibit B-1-1, Appendix 1, pp. 4–5, Tables 17, 19 and 27 Updates.

²³ Exhibit B-4, BCUC IR 1.1.

²⁴ Creative Energy Final Argument, p. 5.

²⁵ Creative Energy Final Argument, p. 14.

²⁶ Exhibit B-4, BCUC IR 2.5.

its view, a better and far less controversial approach is to record the proposed amount to the TPRCDA in 2019 for prospective recovery.²⁷

Creative Energy states that approximately 87 percent of the increase in 2019 proposed revenue requirement compared to the 2018 approved forecast relates to externally driven cost pressures largely outside of its control, as follows:²⁸

- Actual regulatory costs in 2019 compared to 2018 approved forecast²⁹ increased by \$177,349;³⁰ and
- Actual water costs from the City of Vancouver increased by \$193,945 in 2019 compared to the 2018 approved forecast.³¹

Creative Energy submits that the regulatory costs were reasonably incurred and required to support its applications before the BCUC as a regulated public utility under the UCA. Therefore, they ought to be recoverable.³² However, it proposed to add only \$103,241 of the \$177,340 increased regulatory costs to the TPRCDA, representing the difference between Creative Energy's updated 2019 revenue requirement and the 2019 actual revenues collected. Creative Energy explains this approach provides a beneficial sharing with customers of the higher actual load in 2019 compared to the lower load for an average year, which Creative Energy states it would have forecasted for 2019 (i.e. similar to the 2018 approved load forecast).³³ The 2019 actual steam load was 1,126,060 thousand pounds of steam.³⁴

2018 Regulatory and Pension Expense Variances

With respect to the BCUC-approved deferral treatment for any forecast variances related to regulatory costs and pension expense for 2018, Creative Energy seeks approval of: \$207,659³⁵ to be added to the TPRCDA; and a net loss of \$377,966 to be added to the Pension Expense Deferral Account, as follows:

Description	2018 Approved Forecast	2018 Actual	Variance
Current Service Costs:			
Plant	\$101,972	\$98,280	\$3,692
Service	\$28,641	\$27,181	\$1,460
Management	\$59,671	\$39,088	\$20,583
Total	\$190,284	\$164,549	\$25,735
Remeasurement Gain/(Loss)	<u>\$0³⁷</u>	<u>\$403,700</u>	<u>(\$403,700)</u>
Total	\$190,284	\$568,249	(\$377,966)

Table 2 – 2018 Pension Expense Variances³⁶

Creative Energy explains that the variance between 2018 actual and approved forecast regulatory costs was primarily due to BCUC, intervenor, consulting and legal costs for the review of the Creative Energy 2018-2022 Revenue Requirements Application and BCUC fees for the Creative Energy Application for a Certificate of Public

²⁷ Creative Energy Final Argument, p. 15.

²⁸ Creative Energy Final Argument, p. 15.

²⁹ 2018 Forecast was the last forecast approved by the BCUC.

³⁰ Calculated as the difference between 2018 Approved Forecast of \$135,651 and 2019 Actual of \$312,991 as shown in Exhibit B-1-1, Appendix 1, p. 5, Table 27 Update.

³¹ Calculated as the difference between 2018 Approved Forecast of \$502,200and 2019 Actual of \$696,145 as shown in Exhibit B-1-1, Appendix 1, p. 5, Table 23 Update.

³² Exhibit B-4, BCUC IRs 22.6, 22.6.1.

³³ Creative Energy Final Argument, p. 15; Exhibit B-4, BCUC IR 2.8.

³⁴ Creative Energy Final Argument, p. 15.

³⁵ Exhibit B-1, p. 48, Table 52.

³⁶ Exhibit B-1, p. 48, Table 51; Exhibit B-4, BCUC IR 21.1.

³⁷ In response to BCUC IR 21.3, Creative Energy states that pension remeasurement gains/losses cannot be forecasted in advance.

Convenience and Necessity for Beatty-Expo Plants and Reorganization (Beatty-Expo CPCN).³⁸ For pension-related costs, Creative Energy states that current service costs in 2018 were lower than the 2018 approved forecast. However, there was a revaluation of the pension asset resulting in a remeasurement loss of \$403,700 as the return on plan asset was not sufficient to cover the interest expense on pension liability.³⁹

In addition to the \$103,241 difference between its updated 2019 revenue requirement and the 2019 actual revenues collected discussed above, Creative Energy proposes to amortize the \$207,659 regulatory cost variance and the \$377,966 pension expense variance, including interest, over two years to be collected from customers by way of the DARR mechanism, commencing January 1, 2020.⁴⁰ Creative Energy states that the proposed two-year amortization period achieves an overall rate impact in 2020 that is less than 10 percent. Creative Energy states that a shorter amortization period (e.g. a one-year amortization period) would result in an overall 2020 rate impact that would exceed 10 percent.⁴¹

Subsequent to filing the Application and the Evidentiary Update, Creative Energy stated that the 2019 actual pension remeasurement gain amount of \$446,700 became known.⁴² The following table shows a breakdown of the 2019 pension remeasurement gain and the historical gains/(losses) from 2015 to 2019. The Panel notes that the 2019 pension remeasurement gain is \$446,200 in this table instead of \$446,700:

	2015	2016	2017	2018	2019
Interest on defined benefit obligations	(241,400)	(247,100)	(258,000)	(269,100)	(280,300)
Return (loss) on Plan Assets	177,800	335,000	343,700	(59,500)	792,400
Finance Costs	(31,600)	(28,300)	(29,600)	(25,100)	(15,500)
Administrative Costs	*	*	(50,000)	(50,000)	(50,400)
Actuarial Loss	0	0	(252,500)	0	0
	(95,200)	59,600	(246,400)	(403,700)	446,200

Table 3 – 2015 to 2019 Pension Remeasurement Gains/(Losses)⁴³

With respect to how the 2019 actual remeasurement gain should be treated, Creative Energy explained:

The remeasurement gain will be recorded as a reduction in the deferral account balance. The Deferral Account Rate Rider for 2020 and 2021 is currently based on the 2018 balance and the remeasurement loss from that year. Under this lagged approach, the remeasurement gain for 2019 will be refunded through the Deferral Account Rate Rider starting in 2021 and through 2022.⁴⁴

In its final argument, Creative Energy states it would be amenable to the BCUC directing that the DARR be amended for the remeasurement gain in 2019 to remove the timing difference.⁴⁵

2019 After-tax Regulatory Pension Asset Account and Related Return

³⁸ Exhibit B-1, p. 49.

³⁹ Exhibit B-1, p. 48.

⁴⁰ Exhibit B-1, p. 47; Exhibit B-4, BCUC IR 23.10.

⁴¹ Creative Energy Final Argument, p. 20.

⁴² Exhibit B-4, IR 21.3.

⁴³ Exhibit B-9, BCUC IR 54.8.

⁴⁴ Exhibit B-9, BCUC IR 54.4.

⁴⁵ Creative Energy Final Argument, p. 12.

Background

In addition to the Pension Expense Deferral Account, in the 2015-2017 Core Steam System Decision, the BCUC approved the establishment of a rate base mid-year After-tax Regulatory Pension Asset account and determined that the balance for 2015 was \$414,012.⁴⁶ The BCUC further stated:

... the Panel directs that when determining the mid-year After-tax Regulatory Pension Asset in future test periods the opening balance will equal the previous year's December 31 Pension Asset (after-tax) reported on the audited financial statements.

[emphasis in original]

The BCUC determined that the opening balance (i.e. the previous year's December 31 Pension Asset (after-tax) reported on the audited financial statements) is then adjusted for the after-tax variance between forecast employer contributions and forecast pension expense for the test period. The mid-year amount is then determined as an average of the opening and closing balances.⁴⁷

In the subsequent 2016-2017 Core Steam System and NEFC proceeding, which followed the 2015-2017 Decision approving only a one-year revenue requirement, Creative Energy explained that the After-tax Regulatory Pension Asset account balance was not updated for 2016 because the company's financial statements were not finalized at the time that application was filed.⁴⁸

For 2018, the After-Tax Regulatory Pension Asset was not addressed in the 2018-2022 Core Steam System Decision and the balance remained at \$414,012 for the one-year revenue requirement that the BCUC approved. There was no forecast amount for contributions provided in 2018 and the approved 2018 forecast pension expense was \$190,284. Since Creative Energy did not forecast contributions for 2018 and did not request adjustments to the After-tax Regulatory Pension Asset, in the Panel's view, this has the effect of setting the 2018 forecast contributions equal to the forecast pension expense. As a result, no adjustments to the After-Tax Regulatory Pension Asset were approved in the 2018-2022 Core Steam System Decision.

In summary, the After-tax Regulatory Pension Asset first approved by the BCUC in the 2015-2017 Core Steam System Decision remained at \$414,012 for each test period from 2015 to 2018 inclusive. The regulatory asset was not updated to reflect the opening balance equal to the previous year's December 31 Pension Asset (after-tax) reported on the audited financial statements and did not incorporate any difference between forecast pension expense and contributions for the respective test periods. Based on the BCUC directive in the 2015-2017 Core Steam System Decision, the opening balance for the After-Tax Regulatory Pension Asset for future test years should be equal to the previous year's December 31 Pension Asset (after tax) reported on the audited financial statements. The Pension Asset (before tax) recorded in Creative Energy's audited financial statements was \$579,600 for 2017 and \$367,400 for 2018.⁴⁹

Creative Energy's Proposal

During the IR process, Creative Energy submitted that the After-Tax Regulatory Pension Asset account should have been increasing each year based on the cash contributions made above and beyond those related to the current service costs. Using the following information, Creative Energy recalculated the additions to the deferral account as follows:

⁴⁶ Order G-98-15, Directive 1(f).⁴⁷ 2015-2017 Core Steam System Decision, p. 55

⁴⁷ 2015-2017 Core Steam System Decision, p. 55

⁴⁸ 2016-2017 Core Steam System and NEFC Decision, p. 56.

⁴⁹ Exhibit A2-1.

Table 4 – 2016-2019 Proposed Additions to After-tax Regulatory Pension Asset Account⁵⁰

	2016	2017	2018	2019	Total
Employer contributions	222,800	414,500	417,400	449,100	1,281,000
Current Service Cost	-222,800	-245,200	-251,000	-257,300	-753,500
Solvency Payments	0	169,300	166,400	191,800	527,500
Tax Rate	26%	26%	27%	27%	
Income Taxes	0	-44,018	-44,928	-51,786	-140,732
Additions to Deferral	0	125,282	121,472	140,014	386,768

Using the 2015 Decision amount of \$414,012 as a starting point, Creative Energy recalculated the rate base by adding the amounts in the table above to the After-tax Regulatory Pension Asset account. The recalculated mid-year balances for the year 2015 through 2019 are as follows:

Table 5 – 2015-2019 Pro	posed After-tax Regulator	y Pension Asset Account Balances ⁵¹

	Opening		Additions/					Ending	N	1id-year	
	E	Balance		(Deductions)		Amortization		Balance		Balance	
2015	\$	414,012	\$	-	\$	-	\$	414,012	\$	414,012	
2016	\$	414,012	\$	-	\$	-	\$	414,012	\$	414,012	
2017	\$	414,012	\$	125,282	\$	-	\$	539,294	\$	476,653	
2018	\$	539,294	\$	121,472	\$	-	\$	660,766	\$	600,030	
2019	\$	660,766	\$	140,014	\$	-	\$	800,780	\$	730,773	

The difference between the new mid-year balance of \$730,773 above and the \$414,012 (which Creative Energy originally used in the Application) results in an incremental increase of \$23,523 to the return on rate base in the 2019 test year.⁵² The impact on the 2020 proposed revenue requirement is addressed in subsection 3.1.

Remote Metering Project

The Remote Metering Project is a multi-year project which will allow Creative Energy to gather real time steam consumption data from its customer buildings. Creative Energy states the availability of this data tracking will improve its billing and engineering processes, as well as allowing customers to gain insights into their energy consumption. The annual cost savings from this project are expected to be \$68,065.⁵³ The total expected capital cost of the project, which began in 2017, is approximately \$750,000.⁵⁴ Initially for 2019, Creative Energy included a \$293,666 capital addition for the Remote Metering Project.⁵⁵

Through subsequent information requests, Creative Energy sought to update its Application and initial responses related to the Remote Metering Project. Creative Energy confirmed that there are currently no remote metering assets in service and that costs reported in Appendix D in 2019 in the amount of \$293,666 should be removed from rate base at this time.⁵⁶

Creative Energy has identified two project execution risks related to the Remote Metering Project: 1) unforeseen delays in getting permission from customer buildings for utilization of their data service to transmit

⁵⁰ Exhibit B-9, BCUC IR 55.1.1.

⁵¹ Exhibit B-9, Schedules for RRA Filing – IR Round 2, Core Schedule 11.

⁵² Exhibit B-9, BCUC IR 55.1.1; As per Core Schedule 13 (filed in Exhibit B-9, Schedules for RRA Filing – IR Round 2), the total return on rate base is \$1,620,000 for 2019, inclusive of the incremental return of \$23,523.

⁵³ Exhibit B-4, BCUC IR 18.9.

⁵⁴ Exhibit B-4, BCUC IR 18.3.1.

⁵⁵ Exhibit B-1, Appendix D.

⁵⁶ Exhibit B-9, BCUC IR 52.2.

metered information; and 2) project cost increases related to remote metering and monitoring cabinet construction and the requirement of some installation locations to require a third-party for installation. Creative Energy states risk is considered low and mitigation measures have been initiated.⁵⁷

Positions of the Parties

The CEC submits it is not appropriate for Creative Energy to present rates on a backwards looking basis and Creative Energy should be discouraged from operating in such a fashion because it makes the potential for rate change more difficult. The CEC submits, "...the Commission should adopt the view that where there is a rate application based on historical data, variances should favour the ratepayer." ⁵⁸

In its review of 2019 costs, the CEC recommends that the BCUC use the same caution it would normally exercise when reviewing test year figures and attribute any cost reductions to the benefit of the ratepayer. On that basis, the CEC notes that the costs incurred in 2019 appear to be "generally acceptable." However, the amount which Creative Energy proposes to add to the TPRCDA could be reasonably denied in whole, or part, because Creative Energy did not make an effort to conduct an RRA in a timely manner and the risk should be to the shareholder. The CEC notes that Creative Energy acknowledged that the proposed addition to the TPRCDA relates to "incorporating" an allowed return on equity (ROE) of 9.5 percent into the 2019 revenue requirement.⁵⁹ The CEC views that the TPRCDA should not have the purpose of being used for balancing the utility's ROE.⁶⁰

The CEC does not comment on the proposed addition for 2018 or change to the TPRCDA amortization period and does not raise any issues regarding the Pension Expense Deferral Account and account balances.⁶¹

The CEC does not comment on the Remote Metering project nor the After-tax Regulatory Pension Asset Account and related returns.

In reply to the CEC, Creative Energy states, "the costs at issue are by their nature necessarily incurred for utility purposes and the large majority of the costs are not within management's ability to control and therefore should not be the shareholder's risk." Creative Energy submits regulatory costs are undoubtedly recoverable in utility rates, and deferral account treatment of variances between forecast and actual costs is appropriate between forecast and actual costs because it is very challenging to accurately forecast these costs, utility management has limited or no ability to control variances, and the variance can be material.⁶²

Panel Determination

Use of 2019 Actual Results for Proposed Revenue Requirement

In general, the Panel does not agree that it is appropriate to set utility rates based on hindsight, using actual results. The Panel agrees with the CEC that it is not appropriate for a utility to present rates on a "backwards looking basis". Rates should be set prospectively based on a reasonable forecast of revenue requirement items, and the utility should be held accountable for its forecast with the exception of those revenue requirement items that the regulator approves as being appropriate for deferral treatment because the items are outside the control of management or are subject to a high degree of forecast uncertainty. If rates are set based on actual results, there is no incentive for a utility to operate efficiently since it is effectively guaranteed its allowed rate of return. This is contrary to the regulatory compact under which a utility is afforded a <u>reasonable opportunity</u> to earn a fair return on its invested capital.

⁵⁷ Exhibit B-9, BCUC IR 52.3.2.

⁵⁸ CEC Final Argument, pp. 7-8.

⁵⁹ CEC Final Argument, p. 8.

⁶⁰ CEC Final Argument, p. 8.

⁶¹ CEC Final Argument, p. 22.

⁶² Creative Energy Reply Argument, p. 2.

Furthermore, while Creative Energy's rates were set on an interim basis, thereby alerting customers that the rates may change, the Panel agrees with the CEC that Creative Energy did not file its RRA in a timely manner. In the Panel's view, an unreasonable amount of time (more than a year) passed between the filing for interim rates (December 14, 2018) and final rates (December 19, 2019).

As stated above, the Panel views that rates should not be set using actual results, so it must consider alternative bases for setting 2019 rates. The Panel notes that the BCUC found the forecast information filed in the 2019 interim rate application was not adequate for setting interim rates, and therefore, the Panel concludes this information should not be used to set permanent rates. Given that Creative Energy did not provide supportable forecasts for 2019, the Panel finds it appropriate to maintain the BCUC approved 2018 Steam System Rates and approved forecast for 2018 for setting 2019 permanent rates, including the continuation of variance treatment for Regulatory Costs and Pension Expense.

In making this finding, the Panel takes into consideration that the evidence indicates that maintaining the 2018 rates and underlying 2018 approved forecast provided Creative Energy with a reasonable opportunity to earn its allowed return on equity. As set out in Table 6 below, the Panel estimates that net earnings achieved based on maintaining the 2018 approved rates, which is equal to the 2019 interim rate, were sufficient to enable Creative Energy to earn its allowed return for 2019. The actual 2019 actual revenues exceed actual 2019 costs, after taking into account the Panel's decision regarding treatment of regulatory cost and pension expense variances as set out below. For 2019, the Panel estimates that Creative Energy's achieved rate of return on equity, as follows:

Table 6 – Estimated 2019 Net Earnings Comp	ared to Allowed Equity ROE ⁶³
--	--

		201		
Revenues collected based on Existing rates			\$	8,614,359
Operations and Maintenance Expenses				
Wages and Benefits				2,715,920
Water-related and Electricity Expenses				900,422
Maintenance (including parts, supplies, safety				
and vehicles)				525,87
Special Services				
Audit	\$	57,234		
Legal		5,586		
Outside Services		112,067		
Regulatory		135,651		310,53
Other General & Administration, Sales Expense				
(e.g. insurance, office expenses)		-		274,29
otal Operation and Maintenance Expenses				4,727,05
Aunicipal Taxes				268,38
Ion-Operation and Other				
Property Taxes				663,82
Income Tax Expense				286,70
Depreciation Expense				975,99
Depreciation Expense Allocated to Non Reg				(70
Amortization of CIAC				(31,41
Amortization of Rate Base Deferred Expenses				-
Amortization of Non-Rate Base Deferred Expenses				-
Third Party Regulatory Costs Deferral Account ^a		103,830		
Pension Expense Deferral Account ^b		(25,735)		78,09
Interest				589,00
otal Expenses		-		7,556,93
stimated Net Earnings		-	\$	1,057,42
llowed Equity Rate of Return per Core Schedule 11				1,031,00
Adjustment for After-Tax Regulatoy Pension Asset ^c				(14,86
djusted Equity Rate of Return		-		1,016,13
stimated Difference		-	\$	41,28
lotes:				
a: \$207,659 x 50% = \$103,830, before interest				
: \$25,735 x 100% = \$25,735, before interest				
· Coo Toble 11 bolew (\$253 570 v 43 59/ v 0 59/) (\$720	772		-	1014.00

c: See Table 11 below. (\$362,670 x 42.5% x 9.5%) - (\$730,773 x 42.5% x 9.5%) = (\$14,862)

For the reasons outlined above, the Panel approves the 2019 interim rates as permanent. The Panel's decision regarding treatment of regulatory cost and pension expense forecast variances in 2019 rates is set out below.

Regulatory Cost Variances

The Panel approves the continuation of the non-rate base Third Party Regulatory Cost Deferral Account for 2019. Variance treatment for third-party regulatory costs is consistent with previous BCUC decisions, and Creative Energy has established that these costs are outside the control of Creative Energy and are difficult to

⁶³ This table was prepared by the Panel using information from: Exhibit B-1-1, Appendix 1, pp. 4–5, Tables 15, 17, 19, 21 and 27 Update; and Exhibit B-9, Schedules for RRA Filing – IR Round 2, Core Schedules 1 and 13.

estimate. Therefore, the Panel directs that the following forecast variance amounts be recognized as additions to the TPRCDA:

- 2018 forecast variance of \$207,659 plus interest; and
- 2019 forecast variance of \$177,340 plus interest.

Regarding Creative Energy's request to amortize the 2018 TPRCDA balance over two years, the Panel acknowledges the variance for 2018 is significant relative to the 2018 approved forecast amount. The Panel also notes Creative Energy's desire to maintain 2019 permanent rates at the same amount as the approved interim rates. While the Panel agrees it is reasonable to amortize this amount over two years, consistent with the Panel's finding above that it should maintain the 2018 approved rates and forecast for 2019, including the continuation of forecast variance treatment for Regulatory Costs and Pension Expense, the Panel sees no rationale for Creative Energy's proposal not to record any of the amortization of the 2018 TPRCDA balance in 2019. Using a two-year amortization period for the 2018 balance sufficiently moderates the impact on the 2019 revenue requirement as reflected in Table 6 above. For 2019, Creative Energy is directed to recognize amortization of 50 percent of the 2018 TPRCDA account balance or \$103,830, plus interest. The amounts to be amortized for 2020 are addressed in Subsection 3.2.3.

Pension Expense Variances

The BCUC has previously approved forecast variance treatment for pension expense with a one-year amortization period and a carrying cost equal to Creative Energy's short-term debt rate. This variance treatment recognizes that pension expense is based on many elements that are outside the control of management and that it is difficult to forecast. Creative Energy explains that pension remeasurement gains and losses cannot be forecast in advance and that the gains and losses experienced in 2018 and 2019 demonstrate the fluctuations can be significant. The Panel agrees that the continuation of deferral treatment for pension expense forecast variances is appropriate.

Accordingly, the Panel approves the continuation of the Pension Expense Deferral Account with a carrying cost equal to Creative Energy's short-term debt rate, until otherwise determined by the BCUC. The Panel directs that the following forecast variance amounts be recognized as additions to the Pension Expense Deferral Account:

- The 2018 forecast variance of \$377,966 plus interest; and
- The 2019 forecast variance, based on the 2018 approved forecast, plus interest, as set out below.

Description	1	Amount		
Current Service Costs	\$	56,232		
Remeasurement Gain/(Loss)	\$	446,200		
Total	\$	502,432		

Table 7 – 2019 Pension Expense Variances⁶⁴

The Panel is not convinced that the previously approved one-year amortization period remains appropriate for pension remeasurement gains and losses. The magnitude and volatility of the gains and losses, and the likelihood that such variances may continue in the future especially given the expected economic impacts of COVID 19, indicate that a one-year amortization period of these amounts may not be appropriate. While Creative Energy's proposal to amortize the pension expense variances over two years has the effect of smoothing the 2018 and 2019 remeasurement impacts over a longer period, the fact that the amounts are largely offsetting will result in significant volatility of rates.

⁶⁴ This table was prepared by the Panel using information from: Exhibit B-9, BCUC IR 54.8 and 54.9.

Creative Energy states it would be amenable to removing the timing difference by offsetting the 2018 remeasurement loss with the 2019 gain. In the Panel's view, this approach is practical given that the amount of 2019 Pension Remeasurement gain is known. Therefore, Creative Energy is directed to record the following adjustments to the Pension Expense Deferral Account, effective 2019:

Description	Amount		
2018 Remeasurement Loss	\$ (403,700)		
2019 Remeasurement Gain	\$ 446,200		
Difference 2018 and 2019	\$ 42,500		

Table 8 – Offset of the 2018 and 2019 Remeasurement amounts⁶⁵

After this adjustment, the Panel estimates that the remaining amounts in the Pension Expense Deferral Account can reasonably be amortized over a one-year period and recognized in 2019 and 2020 rates. Accordingly, Creative Energy is directed to amortize the 2018 current service costs variance of \$25,735 recorded in the Pension Expense Deferral Account, plus interest, in 2019. The amounts to be amortized in 2020 are addressed in subsection 3.2.4.

Given the significant pension remeasurement gains and losses noted above, to assist the BCUC in determining an appropriate amortization period for Pension Expense Deferral Account in the next RRA, **Creative Energy is directed to file information in the next RRA detailing how other BC and Canadian regulated utilities treat pension remeasurement gains and losses for ratemaking purposes.**

After-tax Regulatory Pension Asset Account and Related Return

Based on the BCUC's determinations in the 2015-2017 Core Steam System Decision, the Panel finds Creative Energy has not calculated the mid-year After-tax Regulatory Pension Asset Account and Related Return correctly for 2019 and 2020. Creative Energy uses the previous year's approved mid-year balance as the opening balance instead of the previous year's December 31 Pension Asset (after tax) reported on the audited financial statements as set out in the 2015-2017 Core Steam System Decision.

Updated calculations reflecting the BCUC's determinations in the 2015-2017 Core Steam System Decision are as set out in the following tables:

	For	ecast 2019
December 31 Pension Asset (pre-tax) reported on		
the audited financial statements (Opening Asset)	\$	367,400
Employer Contributions ¹		449,100
Approved Forecast Pension Expense ²		<mark>(190,284</mark>)
Additions to Asset		258,816
Closing Asset		626,216
Average of Closing Actual and Forecast		496,808
Tax Rate ³		<u>27%</u>
After-tax Mid-year Balance	\$	362,670

 Table 9 – 2019 After-tax Mid-Year Regulatory Pension Asset

¹ Taken from Table 4: 2016-2019 Proposed Additions to After-tax Regulatory Pension Asset Account, 2019 Employer Contributions

² 2018 Approved Forecast Pension Expense, Section 2.2

³ Taken from Exhibit B-9, BCUC IR 55.1.1

⁶⁵ This table was prepared by the Panel using information from: Exhibit B-4, BCUC IR 21.4; and Exhibit B-9, BCUC IR 54.8.

	 fore 2019 justment	fter 2019 ljustment	Impact on Allowed Return		
After-tax mid- year balance	\$ 730,773	\$ 362,670			
Equity of 42.5%	 310,579	 154,135			
Allowed Return on Equity of 9.5%	\$ 29,505	\$ 14,643	\$	(14,862)	

Table 10 – Adjustment to 2019 Allowed Return

While Creative Energy only provides actual employer contributions for 2019, use of the actual contribution amount is appropriate in this circumstance because the employee contributions are determined based on funding requirements and are outside the control of management.

Creative Energy is directed to reduce the allowed return for 2019 by \$14,862 based on a mid-year After-tax Regulatory Pension Asset Account balance of \$362,670. The Panel's determination of the impact on the 2020 proposed revenue requirement is addressed in subsection 3.1.

Remote Metering Project

Since the costs associated with the Remote Metering Project are not capital additions in the current test period and Creative Energy is not requesting section 44.2 UCA expenditure schedule acceptance of the project, the Panel makes no determination on the need for or the costs associated with this project. The BCUC will consider need, prudency and project execution issues if Creative Energy files an expenditure schedule for acceptance or applies to have the project costs included in rates. **Accordingly, the Panel makes no determination on the Remote Metering Project or any of the costs associated with this project.** Additionally, the Panel notes that Creative Energy has not requested an Allowance for Funds Used During Construction on this capital project in accordance with usual regulatory practice.

3.0 Setting 2020 Rates for the Core Steam System

To review Creative Energy's proposals for 2020 rates for the Core Steam System, the Panel considers the following:

- The proposed 2020 revenue requirement;
- The deferral treatment of certain O&M expenditures and the proposal to establish a DARR to recover related variances; and
- The proposals regarding the Fuel Switch Study and LTRP Deferral Account.

The Panel provides its overall determination on 2020 rates for the Core Steam System in subsection 3.4.

3.1 2020 Proposed Revenue Requirement

Creative Energy submits, at the time of the Application, that the outlook for 2020 operating conditions is "business as usual." It explains that the increase in its revenue requirement is driven by increased charges and

levies by governmental bodies and general inflationary cost pressures which are offset by modest expected new customer load.⁶⁶

Creative Energy summarizes its proposed 2020 revenue requirement reflecting the proposed rate increase of 4.2 percent in the following table:

Common and	2020
Component	Test Year
Operation & Maintenance Expense	\$ 5,121,761
Municipal Taxes	277,983
Other	-
Total Operating and Maintenance Expense	5,399,744
Property Tax Expense	763,300
Income Tax Expense	291,800
Depreciation Expense	994,600
Depreciation Expense Allocated to Non Reg	(706)
Amortization of CIAC	(31,410)
Actual/Proposed Return on Rate Base	1,669,000
Cost of Services	9,086,328
Total Revenue Requirement	\$ 9,086,328

Table 11 – Proposed 2020 Revenue Requirement⁶⁷

Creative Energy seeks approval to increase Core Steam System rates by 4.2 percent, effective January 1, 2020, and proposes to establish a new DARR, effective January 1, 2020, to recover forecast variances on certain O&M items on which it requests deferral treatment.⁶⁸ The request for a DARR mechanism is considered in subsection 3.2.5 and the Panel's overall determination on 2020 rates for the Core Steam System is in subsection 3.4.

Forecast Steam Load

Creative Energy seeks approval of a 2020 steam load forecast of 1,140,634 thousand pounds of steam⁶⁹ for the purpose of determining the average rate increase in the 2020 RRA test period.⁷⁰ Creative Energy explains that steam loads are the essential billing determinants for Core Steam system rate setting since all costs are recovered through the volume of steam sold to customers. In addition to its use in determining the rates for the 2020 test year, the steam load forecast is used for other ratemaking purposes, for example, forecasting the fuel charge, forecasting recovery of Fuel Cost Stabilization Account (FCSA) balances, forecasting the proposed DARR, and implementing the load variance component of the COVID-19 Deferral Account.⁷¹ It should also be noted that the Core Steam system supplies thermal energy to the NEFC system, and therefore, the steam load consists of the load to serve both Core and NEFC customers.

The 2020 forecast is normalized for weather and the addition of new customer load as set out in the table below:

⁶⁶ Exhibit B-1, pp. 1, 5.

⁶⁷ Exhibit B-9, Schedules for RRA Filing – IR Round 2, Core Schedule 1.

⁶⁸ Creative Energy Final Argument, pp. 2, 6.

⁶⁹ Denoted as "M#".

⁷⁰ Exhibit B-1, p. 21.

⁷¹ Ibid.; Creative Energy Final Argument, p. 5.

Table 12 – 2020 Forecast Steam Load – Existing and New⁷²

M#	Core Steam	NEFC Steam	Total Steam
Existing Load	1,069,572	57,459	1,127,031
New Load	753	12,849	13,602
Total	1,070,325	70,309	1,140,634

The total 2020 steam load forecast generally consists of: (i) a base forecast of the Core component of steam load based on the 2018 approved forecast, adjusted for known customer additions; and (ii) a forecast of the NEFC steam load based on 2018 and 2019 hot water consumption (megawatt hours or MWh), adjusted for customer additions known at the time of the forecast, converted to thousand pounds of steam and adjusted for system efficiencies.⁷³

Creative Energy noted that in the course of preparing the load forecast, it determined that the PARQ Hot Water Plant was underreporting steam flow from around March or April 2018 to March 2020. Creative Energy proposes to address this issue for the 2020 Test Year by recognizing the increased steam supplied to the NEFC system, and therefore, included a higher 2020 steam load forecast based on a calculation of steam usage at PARQ derived using the total forecast MWh of hot water consumption as detailed below.⁷⁴ The impact of the PARQ metering issue for the 2019 Test Year is discussed in subsection 4.2.

A more detailed account of the methodology for calculating the Core component of the steam load is as follows:⁷⁵

- 1. Core customer base load is normalized by adjusting for customer attrition, entire-year consumption, and removing customer load with unpredictable or atypical load shapes using the most recent 5 years of customer data.
- 2. The normalized Core Load data set is used to calculate both a simple average and a predictive result based on a linear regression of monthly Heating Degree Days (HDD) and monthly steam load over the last 5 years, inputting the average of HDD over that period.
- 3. The customer load removed in step 1 is added back to the dataset to derive the overall base load forecast.
- 4. Two additional customers in July 2020 are added.

In addition, Creative Energy calculates the NEFC component of the steam load as follows:⁷⁶

- 1. Hot water consumption from 2018 and 2019 for existing customers is averaged.
- 2. Additional load for one new customer is added scaled on the basis of its floor area compared to similar customers.
- 3. The forecast hot water consumption is then adjusted for the measured efficiencies of the distribution system and hot water plants and converted from MWh to thousand pounds of steam using a conversion factor of 3.14 thousand pounds of steam per MWh.

Finally, the Core and NEFC components are combined for a total forecast steam load of 1,140,634 thousand pounds of steam.

⁷² Exhibit B-1, p. 21.

⁷³ Exhibit B-1, Appendix C, Section 1.3.1

⁷⁴ Creative Energy Final Argument, p. 29.

⁷⁵ Exhibit B-1, Appendix C, Section 1.3.2

⁷⁶ Exhibit B-1, Appendix C, Section 1.3.2

In the previous two revenue requirement applications, Creative Energy was not amenable to establishing a deferral account to capture steam load variances. While the BCUC respected this position, it recommended that the utility consider options to more effectively manage variances within the load forecast.⁷⁷ Since the Application did not consider any options, in its letter dated May 12, 2020, the BCUC invited all parties to discuss in their final submissions whether a deferral account to capture steam load variances for the Core Steam System may be warranted in light of the evidence of recent variances between forecast and actual load and ongoing developments regarding the COVID-19 global pandemic.

Creative Energy's response to this request focused on the impacts of COVID-19 and did not provide much in the way of commentary on the general issue of variances between the forecast and actual steam load.⁷⁸

Creative Energy outlines that the steam load may vary plus or minus 10 percent in any given year due to the effect of temperature. Creative Energy illustrated the impacts on the revenue requirements and the requested rate increase for the 2020 Test Year under the scenarios that the steam load varies by plus or minus 10 percent, as follows:

	Base Case	10% Increase in Load	10% Decrease in Load
2019 Average Rate	7.65	7.65	7.65
2020 Povonuo Poquiroment	0 002 262	0 102 706	8 002 020
2020 Revenue Requirement	9,093,363	9,193,706	8,993,020
2020 Actual Load Forecast	1,140,634	1,254,697	1,026,571
Average Rate	7.97	7.33	8.76
Increase in Average Rate (\$)	0.32	- 0.32	1.11
Increase in Average Rate (%)	4.2%	-4.2%	14.5%

Table 13 – Impact of Variations in Load⁷⁹

Regarding COVID-19, the Application was filed prior to the emergence of the pandemic, and as such, the load forecast reflects a business-as-usual outlook in which system load was expected to remain stable for existing customers while including a modest additional steam load for new customers. However, Creative Energy confirms that its business has been affected in various ways since the onset of COVID-19 and cites unplanned costs and a decline in steam load as examples. These unexpected impacts are not accounted for within the 2020 revenue requirements or load forecast and Creative Energy has not made any amendments to its Application regarding these impacts.

Instead, Creative Energy stated that it will file a separate application requesting approval to establish a new deferral account to record cost of service and revenue variances that directly result from the impact of COVID-19 on Creative Energy and its customers.⁸⁰ By Order G-214-20 issued August 14, 2020, the BCUC approved such request.

Creative Energy submits that the 2020 load forecast should be approved as proposed on its own merits assuming that COVID-19 had not arisen. Creative Energy maintains that the forecast has been developed using existing forecasting methodologies, with minor exceptions, and is transparent. Further, Creative Energy believes that approval of the 2020 load forecast as proposed will provide the appropriate basis for comparison of the revenue loss relief that will be sought through the COVID-19 Deferral Account.⁸¹

⁷⁷ 2018-2020 Core Steam System Decision, pp. 18–19.

⁷⁸ Creative Energy Final Argument, p. 2.

⁷⁹ Exhibit B-4, BCUC IR 6.1

⁸⁰ Exhibit B-9, BCUC IR 41.1 and 41.1.1.

⁸¹ Creative Energy Final Argument, pp. 6-8.

Forecast O&M Expenses

Creative Energy examines its O&M expenses using five categories of cost drivers: 1) wages and benefits, including overtime and pension expense; 2) water and electricity-related expenses; 3) maintenance; 4) special services, including regulatory, audit, legal, and third-party consultant (outside services) costs; and 5) other general and administrative expenses plus sales expenses.⁸²

For the Core Steam System, Creative Energy forecasts O&M expenses of \$5,121,761 for 2020 compared to 2019 actuals of \$4,934,103.⁸³ It explains that the forecast increase in O&M expenses primarily relates to union-approved wage increases of 1.8 percent and increased wages and benefits pertaining to the expected hiring of a Chief Financial Officer in early 2020, which is a role that was vacant during 2019.⁸⁴

The Panel examines below the following issues with respect to the Core Steam System's forecast O&M for 2020:

- Proposed changes to the approved Massachusetts Formula methodology for allocating sales, general and administrative expenses between Creative Energy's Core Steam System and other areas of business;
- Forecast outside services costs; and
- Forecast overtime expense for steam production.

Massachusetts Formula

In the 2018-2022 Core Steam System Decision, the BCUC approved the use of a Massachusetts Formula methodology for allocating sales, general and administrative expenses between the Core Steam System and Creative Energy's other regulated TES projects, excluding the NEFC. The approved formula is based on the arithmetic average of three factors: the average net book value of capital assets or property, plant and equipment; salaries or direct labour expenses; and operating revenues.⁸⁵

Creative Energy requests the following changes to the approved Massachusetts Formula methodology, commencing in 2020:

- 1. Approval to include the NEFC in the cost allocation; and
- 2. Approval to modify the Massachusetts Formula to use either:
 - Two out of the three previously approved factors only: salaries or direct labour expenses; and operating revenues (preferred approach or Option 1); or
 - Gross (undepreciated) book values of capital assets or property, plant and equipment in a continued three-factor formula including salaries or direct labour expenses and operating revenues (Option 2).⁸⁶

Regarding the approval, Creative Energy submits that moving to a formulaic approach to estimating NEFC's overhead costs is consistent with how Creative Energy's other energy systems are treated, practical, and is an

⁸² Creative Energy Final Argument, p. 8; Exhibit B-1-1, Appendix 1, p. 4.

⁸³ Exhibit B-9, Schedules for RRA Filing – IR Round 2, Core Schedules 1 and 15

⁸⁴ Exhibit B-1, pp. 25, 27.

⁸⁵ 2018-2022 Core Steam System Decision, p. 37. The GL account numbers included in the methodology were: 910, 911, 915, 920-926, 930.1, 930.2 and 932.

⁸⁶ Exhibit B-4, BCUC IR 4.3; Creative Energy Final Argument, pp. 17–18.

improvement on how NEFC's overhead costs are currently determined (i.e. direct allocation) for recovery in rates.⁸⁷

Regarding its preferred approach to the modification of the Massachusetts Formula (i.e. Option 1), Creative Energy submits that the current approved three-factor Massachusetts Formula of allocating costs partly based on depreciated capital assets does not yield the most equitable results because the Core Steam System has comparatively lower embedded costs on a unit basis compared to newer systems. Therefore, while the Core Steam System has significantly more customers, the relative allocation of costs is disproportionately weighted toward newer systems with fewer customers.⁸⁸ Creative Energy states that it is not aware of any other jurisdictions using its proposed two-factor approach as proposed, but it submits that the proposal's merits are as described above.⁸⁹ Notwithstanding, Creative Energy states that it would also be amenable to a three-factor methodology that uses gross or undepreciated capital assets as one of the Massachusetts Formula components (Option 2), replacing the net book value of capital assets or property plant and equipment. Creative Energy considers that this would also serve to improve the fairness in the allocation of costs across all of Creative Energy's projects and is consistent with standard industry practice to use gross book values in the Massachusetts Formula.⁹⁰

During the course of the proceeding, the Panel noted that the Creative Energy stated that approximately half of the forecast special services expenses and half of the insurance costs are directly charged to the Core Steam System. The remaining forecast for these costs and other general and administrative costs is calculated based on the proposed two-factor Massachusetts Formula. However, sales expenses are not included.⁹¹

Forecast Outside Services Costs

Outside services costs relate to consulting costs for government advisory services, business development opportunities and the scoping of low-carbon project opportunities and funding sources.⁹² These costs are required when Creative Energy considers that its internal staff do not have the capacity or specific expertise required to assist with certain priorities and the additional cost of hiring a full headcount cannot be justified.⁹³ Forecast outside services costs are included in special services expense,⁹⁴ which are both directly charged and allocated to the Core Steam System based on the Massachusetts Formula as discussed above.

Creative Energy's forecast for outside services costs is \$105,466 for 2020, made up of \$50,028 for government advisory costs for low-carbon energy development and \$55,438 for other advisory and consulting services.⁹⁵ Creative Energy submits that the average actual cost from 2015 through 2018 has been approximately \$100,000⁹⁶ and is the basis for the 2020 test year forecast. Creative Energy submits that 2019 actuals of \$112,067, which became known after the 2020 forecast was developed, confirm that the 2020 estimate is reasonable.⁹⁷

Forecast Steam Production Overtime Expense

Overtime expense is a component of Creative Energy's wages and benefits. Creative Energy states that overtime expense is unpredictable as it is driven primarily by sick days.⁹⁸ For operational and safety reasons, Creative

⁸⁷ Creative Energy Final Argument, p. 17; Exhibit B-9, BCUC IR 60.1.

⁸⁸ Creative Energy Final Argument, p. 17.

⁸⁹ Exhibit B-4, BCUC IR 4.4.

⁹⁰ Creative Energy Final Argument, pp. 17–18.

⁹¹ Exhibit B-4, BCUC IR 4.1.1. This IR response shows the following GL account numbers were included in Creative Energy's proposed two factor Massachusetts Formula: 915, and 920-926. ⁹² Exhibit B-1, p. 32.

⁹² Exhibit B-1, p. 32.

⁹³ Creative Energy Final Argument, p. 11.

⁹⁴ Exhibit B-1, p. 31.

⁹⁵ Exhibit B-9, BCUC IR 48.1.

⁹⁶ Exhibit B-1, p. 32.⁹⁷ Creative Energy Final Argument, p. 11; Exhibit B-1-1, Appendix 1, p. 5, Table 27 Update.

⁹⁷ Creative Energy Final Argument, p. 11; Exhibit B-1-1, Appendix 1, p. 5, Table 27 Update.

⁹⁸ Exhibit B-9, BCUC IR 44.3.

Energy explains that a minimum number of staff are required for each shift at the steam plant. This results in overtime to meet the staffing requirement when any staff member is absent on short notice due to sickness. In addition, overtime is used to cover staff absences to complete scheduled plant and distribution projects that cannot be rescheduled and that have minimum personnel requirements for safety reasons.⁹⁹

Creative Energy seeks approval for its 2020 forecast overtime expenses of \$49,146 for the steam production team based on an average of 2018 and 2019 actual overtime.¹⁰⁰

Forecast Capital Additions

Creative Energy states that forecast capital additions for 2020 are \$1.6 million. Creative Energy further states that year-on-year increases to rate base presented in this Application are primarily driven by these projected capital additions.¹⁰¹

Of this amount, the costs directly associated with the addition of new customers are projected to be \$518,382 in 2020.¹⁰² Through responses to information requests, Creative Energy amended the forecast stating that the customer connection to 410 W Georgia is now expected to commence service in 2021 instead of 2020.¹⁰³ Accordingly, Creative Energy removed \$249,862 from the 2020 forecast capital additions.¹⁰⁴

Forecast After-tax Regulatory Pension Asset Account and Related Returns

As discussed in subsection 2.2, during the course of the proceeding Creative Energy recalculated its After-tax Regulatory Pension Asset Account. It recalculated the mid-year balance for 2020 to be \$800,780, as follows:

	Opening Balance	dditions/ eductions)	Amo	ortization	Ending Balance	/lid-year Balance
2019	\$ 660,766	\$ 140,014	\$	-	\$ 800,780	\$ 730,773
2020	\$ 800,780	\$ -	\$	-	\$ 800,780	\$ 800,780

Table 14 – 2020 Proposed After-Tax Regulatory Pension Asset Account Balances¹⁰⁵

For 2020, the difference between the recalculated 2020 mid-year balance of \$800,780 and \$414,012 (which Creative Energy originally used in the Application) results in an incremental increase of \$28,728 to the return on rate base.¹⁰⁶

Positions of the Parties

Forecast Steam Load

The CEC states that it has reviewed the evidence related to the load forecast and does not object to the forecast as a basis for the revenue requirement,¹⁰⁷agreeing with Creative Energy's proposal to file a separate application

⁹⁹ Creative Energy Final Argument, p. 10; Exhibit B-9, BCUC IR 44.4.

¹⁰⁰ Exhibit B-9, BCUC IR 44.1; Creative Energy Final Argument, p. 10.

¹⁰¹ Exhibit B-1, p. 39.

¹⁰² Exhibit B-1, Appendix D.

¹⁰³ Exhibit B-9, BCUC IR 51.2.

¹⁰⁴ Exhibit B-9, BCUC IR 42.1.

¹⁰⁵ Exhibit B-9, Schedules for RRA Filing – IR Round 2, Core Schedule 11.

¹⁰⁶ Exhibit B-9, BCUC IR 55.1.1; As per Core Schedule 13 (filed in Exhibit B-9, Schedules for RRA Filing – IR Round 2), the total return on rate base is \$1,669,000 for 2020, inclusive of the incremental return of \$28,728.

which will address the impacts of COVID-19.¹⁰⁸ In the CEC's view, impacts from the pandemic are not predictable with any reasonable level of confidence, and as such, addressing the impacts of COVID-19 in this proceeding would increase costs and would not provide definable improvements in forecasts. The CEC submits that a deferral account is or will be necessary to account for COVID-19 impacts and is preferable to attempt to isolate those from business as usual rather than from a mid-pandemic perspective.¹⁰⁹

Forecast O&M Expenses

Subject to the comments noted below, the CEC submits that the 2020 Forecast O&M expenses for the Core Steam System are reasonable. 110

Massachusetts Formula

The CEC submits that both of Creative Energy's proposed changes to the Massachusetts Formula should be rejected because a two-factor method has no known precedent in any other jurisdiction and would be inappropriate to make significant changes to a Massachusetts Formula model in order to achieve a particular outcome. Similarly, the CEC further submits that moving to a revised three-factor Massachusetts Formula is not appropriate, stating that there is a lack of clarity in the evidence in this respect.¹¹¹

In its reply argument, Creative Energy submits that its proposals are non-controversial, unbiased and based on the principle of fairness to the utility customers served. With respect to a three-factor approach, Creative Energy submits that the industry standard is gross book values, as opposed to net book values, of capital assets or property plant and equipment and provides reference to two sources, which were submitted in the proceeding on the 2018-2022 Core Steam System Decision.¹¹²

Forecast Outside Services Costs

The CEC submits that Creative Energy has not adequately justified the forecast outside services costs and submits that the BCUC should consider denying half of the costs, noting that the 2020 forecast is \$54,261 higher than 2018 approved of \$51,205.¹¹³

Creative Energy responds stating that 2018 approved is not representative of ongoing priorities and requirements, whereas the 2015 to 2018 actual outside services costs are representative and have been approximately \$100,000 per year. Accordingly, Creative Energy submits the CEC's suggestion should be disregarded as it is without merit and is unsubstantiated.¹¹⁴

Forecast Steam Production Overtime Expense

The CEC submits that Creative Energy should standardize its forecasting approach and that it is "appropriate practice" to use the average of the most recent three years of actuals, instead of two years, in order to avoid applying too much weight to what appears to be an outlier year. The CEC calculates that forecast steam production overtime expense would be reduced by approximately \$4,500 based on three years of actuals.¹¹⁵

In its reply argument, Creative Energy submits that it is amenable to forecast steam production overtime expense as suggested by the CEC, but it does not view that it should adopt a standardized approach to

¹⁰⁷ CEC Final Argument, p. 10.

¹⁰⁸ CEC Final Argument, p. 3.

¹⁰⁹ CEC Final Argument, p. 3

¹¹⁰ CEC Final Argument, p. 17.

¹¹¹ CEC Final Argument, pp. 25–27, 35.

¹¹² Creative Energy Reply Argument, p. 4.

¹¹³ CEC Final Argument, p. 16.¹¹⁴ Creative Energy Final Argument, pp. 6–7.

¹¹⁴ Creative Energy Final Argument, pp. 6–7.

¹¹⁵ CEC Final Argument, p. 17.¹¹⁶ Creative Energy Final Argument, p. 6.¹¹⁷ Creative Energy Final Argument, p. 6.

forecasting.¹¹⁶ In its view, a standardized approach would "remove the transparent accounting and judgement required and would be expected to produce inaccurate forecasts in respect of any cost category with cost pressure in one direction year-after-year and/or where specific outlier effects need to be considered in view of a longer-term average."¹¹⁷

Forecast Capital Additions

The CEC recommends that the BCUC accept Creative Energy's capital addition plan as filed.¹¹⁸

Forecast After-tax Regulatory Pension Asset Account and Related Returns

The CEC did not comment on the forecast After-tax Regulatory Pension Asset Account.

Panel Determination

The Panel approves the 2020 forecast revenue requirement of \$9,086,328 subject to the adjustments directed below.

Forecast Steam Load

With respect to the steam load forecast, Creative Energy has proposed a forecast load based largely on existing methodologies, apart from some exceptions. Creative Energy proposes a forecast in keeping with previous BCUC directives by incorporating weather normalization, and by obtaining information from significant customers regarding any pertinent changes to their building or operations that could impact demand. The Panel considers this approach to the load forecast methodology for steam customers to be reasonable and appropriate. Regarding the exceptions, Creative Energy has adjusted the NEFC forecast that considers its limited historical data and experience and addresses issues related to the underreporting of steam at the PARQ plant. The Panel considers these changes to be reasonable in light of the lack of available data and alternatives. Therefore, **the Panel approves the total 2020 steam load forecast of 1,140,634 thousand pounds of steam as requested by Creative Energy.**

However, the Panel expects that as more data is available for the NEFC customer load, Creative Energy will use a methodology for the NEFC forecast more similar to that used for the Core customer load forecast. That is, Creative Energy will calculate the forecast based on historical data and information from significant customers regarding changes that may impact the load forecast and will normalize the load in a manner more consistent with methodology used for the Core component of the steam load forecast.

Regarding the issue of whether a deferral account to record variances between forecast and actual load is warranted, the Panel has difficulty understanding why a deferral mechanism for forecast variances continues not to be considered. A deferral account mechanism would allow any variances in load forecasts to be corrected and trued up with neither the ratepayer nor the utility having an advantage.

While the Panel agrees that the impacts of COVID-19 have been dealt with in a separate proceeding, it points out that it is common utility practice to use a load forecast deferral account in order to avoid the impact of such events that are beyond the control of the utility. For this reason, the Panel encourages Creative Energy to further consider a load forecast deferral account in its next revenue requirements application. The Panel

¹¹⁶ Creative Energy Final Argument, p. 6.¹¹⁷ Creative Energy Final Argument, p. 6.

¹¹⁷ Creative Energy Final Argument, p. 6.

¹¹⁸ CEC Final Argument, p. 20.

considers it beneficial if Creative Energy were to include in its next revenue requirement application information comparing the forecast, the approved and actual historical demand to assist in the timely review of that application.

Forecast O&M Expenses

With respect to 2020 Forecast O&M expenses, the Panel acknowledges the issues raised by the CEC and addresses each in turn.

As for the proposed changes to the Massachusetts Formula, **the Panel approves the inclusion of NEFC in the Massachusetts Formula cost allocation.** This approach is the same as Creative Energy's other regulated TES projects, and the Panel is persuaded that this is a practical approach to estimating the NEFC's costs in serving a separate energy system.

As for the second proposed change involving the proposed modification to the three-factor Massachusetts Formula, the Panel agrees with the CEC that the proposed two-factor formula is "a Creative Energy construct" and is not persuaded that it should be adopted. However, the evidence submitted in Creative Energy's reply argument shows that a three-factor formula based on gross (undepreciated) capital assets or property plant and equipment reflects the industry standard. Accordingly, the Panel denies Creative Energy's request for a twofactor Massachusetts Formula and approves the use of a three-factor Massachusetts Formula, beginning in 2020, based on the following factors: the average gross book value of capital assets or property, plant and equipment; salaries or direct labour expenses; and operating revenues.

Creative Energy is directed to recalculate the costs allocated to the Core Steam System and NEFC, respectively, based on the three-factor Massachusetts Formula approved above in a compliance filing due to the BCUC within 30 days of the date of this decision (Compliance Filing), including the rate impact for each service area. The Panel expects that the impacted costs will include sales expenses and general and administrative costs as previously determined by the BCUC in the 2018-2022 Core Steam System Decision.

Regarding the 2020 forecast outside services costs, the Panel rejects the CEC's proposal to deny half of the costs as the Panel finds this proposal has no evidentiary basis. The 2020 forecast is comparable to that of previous years and the CEC's recommendation does not have a material impact on the 2020 forecast revenue requirement overall. Therefore, the Panel approves the 2020 forecast for outside services costs of \$105,466 as requested by Creative Energy.

The Panel approves the 2020 forecast steam production overtime expense. The Panel finds that basing the forecast on the average of the 2018 and 2019 actuals is a reasonable approach. The Panel notes the difference in amount based on the CEC's recommendation is minimal and that there is no real basis for the three-year forecast suggested by the CEC. The Panel finds no issue with Creative Energy's response regarding a standardized forecast approach and considers it reasonable to have flexibility when preparing forecasts.

Forecast Capital Additions

Finally, the Panel notes that throughout the proceeding the intervener has not raised any specific objections concerning Creative Energy's projected capital additions. The Panel is satisfied that the capital additions as filed and amended during the proceeding are reasonable. However, the Panel has included specific comments earlier regarding the Remote Metering Project in setting 2019 rates.

Forecast After-tax Regulatory Pension Asset Account and Related Returns

As discussed in section 2.0, based on the BCUC's determinations in the 2015-2017 Core Steam System Decision, the Panel finds Creative Energy has not calculated the mid-year After-tax Regulatory Pension Asset Account and related return correctly for 2019 and 2020. Creative Energy uses the previous year's approved mid-year balance as the opening balance instead of the previous year's December 31 Pension Asset (after tax) reported on the audited financial statements as set out in the 2015-2017 Core Steam System Decision.

The updated calculations for 2020, reflecting the BCUC's determinations in the 2015-2017 Core Steam System Decision, are as set out in the following tables:

	Forecast 202			
December 31 Pension Asset (pre-tax) reported on				
the audited financial statements (Opening Asset)	\$	367,400		
Employer Contributions		242,519		
Approved Forecast Pension Expense ¹		(242,519)		
Additions to Asset		-		
Closing Asset		367,400		
Average of Closing Actual and Forecast		367,400		
Tax Rate ²		<u>27%</u>		
After-tax Mid-year Balance	\$	268,202		

Table 15 – 2020 After-tax Mid-year Regulatory Pension Asset

¹ 2020 Approved Forecast Pension Expense subject to the recalculation of pension costs allocated to the Core Steam System and NEFC, respectively, based on the (newly) approved three factor Massachusetts Formula, Section 3.1

² Taken from Exhibit B-9, BCUC IR 55.1.1

	 fore 2020 justment	 fter 2020 ljustment	Impact on Allowed Return		
After-tax mid- year balance	\$ 800,780	\$ 268,202			
Equity of 42.5%	 340,332	 113,986			
Allowed Return on Equity of 9.5%	\$ 32,331	\$ 10,829	\$	(21,503)	

Table 16 – Adjustment to 2019 Allowed Return

The Panel uses the 2018 December 31 Pension Asset as the opening balance since the 2019 audited Financial Statements are not on the evidentiary record. Further, since Creative Energy did not provide forecast employer contributions for 2020 or otherwise request adjustments to the After-tax Regulatory Pension Asset for 2020, the Panel sets the 2018 forecast contributions equal to the approved forecast pension expense.

Creative Energy is directed to reduce the allowed return for 2020 by \$21,503 based on a mid-year After-tax Regulatory Pension Asset Account balance of \$268,202.

3.2 Deferral Accounts

For the Core Steam System, Creative Energy proposes to recognize certain O&M items on a forecast basis and requests deferral treatment for any forecast variances. This requested treatment means that the variance between forecast and actual cost would be captured in an approved regulatory account and carried forward to be disposed of in future years using an approved amortization period and recovery mechanism.

In the decisions establishing Core System Steam rates for 2016-2017 and 2018¹¹⁹, the BCUC approved deferral treatment of forecast variances for regulatory costs and pension expense with resulting variances to be amortized over one year. For the 2019-2020 test period, Creative Energy seeks continuation of deferral treatment for these items and requests other changes as summarized in the following table.

Type of Change	Account	Return Request	Amortization Period Request	Recovery Mechanism Request
New	Water Cost Deferral Account	Short-term debt requested	1 year commencing January 1, 2021 requested	DARR
Account	Property Tax Deferral Account	Short-term debt requested	1 year commencing January 1, 2021 requested	DARR
Other	Third Party Regulatory Cost Deferral Account	N/A – no change requested	2 years commencing January 1, 20202	DARR
other	Pension Expense Deferral Account	N/A – no change requested	2 years commencing January 1, 20202	DARR

 Table 17 – Summary of 2020 Deferral Treatment Requests¹²⁰

The Panel notes that deferral treatment provides certainty of cost recovery for the utility. Accordingly, to consider Creative Energy's requests, the key questions for the Panel are:

- 1. Is the approval of forecast variance treatment for the cost item warranted, and if so, what is the appropriate amortization or recovery period for the variance; and
- 2. Should the recovery mechanism of the cost variance be through the use of a DARR mechanism as proposed by Creative Energy, or alternatively, should the amount be included in subsequent revenue requirement test periods.

In the subsections below, the Panel applies the following criteria to address the first question regarding whether the requested deferral treatment is warranted:

- The extent to which the cost is within the control of management; and
- The degree of forecast uncertainty associated with the cost.

In the Panel's view, if the cost item is reasonably controllable and capable of forecast then it should form part of the forecast revenue requirement and the utility should bear the risk of variance. In contrast, if a utility has limited control over the item or there is a high degree of forecast uncertainty, it may not be appropriate for the utility to bear the risk of forecast variances. In such a case, the establishment of some form of deferral mechanism may be appropriate. The Panel's approach is consistent with the BCUC's Regulatory Accounting Filing Checklist (Checklist), which among other things asks regulated entities applying for a regulatory account to

¹¹⁹ 2016-2017 Core Steam System and NEFC Decision and 2018-2022 Core Steam System Decision.

¹²⁰ Creative Energy Final Argument, pp. 18, 20, 22, 25.

comment on "whether, or to what extent, the item is outside of management's control" as well as the degree of forecast uncertainty associated with the item.¹²¹

Regarding the second key question noted above, the Panel considers Creative Energy's request for permanent approval to use its proposed DARR mechanism in subsection 3.4.5.

3.2.1 Water Cost Deferral Account

Creative Energy requests approval of a deferral account to track the difference between actual total annual water costs versus forecast total annual water costs (the Water Cost Deferral Account or WCDA). The balance in the proposed WCDA would be amortized over a one-year period on an ongoing basis commencing January 1, 2021, at a carrying cost equal to Creative Energy's short-term debt rate.

The cost of water is one of the major expenses for Creative Energy and water usage primarily consists of:

- Feed water for steam production; and
- Water cooling for the safe discharge of distribution system condensate.¹²²

Water is obtained from the City of Vancouver and is subject to water rates and rate design as set by the City of Vancouver. Water rates have increased more than inflation since 2016. The history of water rate increases since 2016 is as follows:¹²³

- 4 percent between 2016 and 2017;
- 6 percent between 2017 and 2018; and
- 10 percent between 2018 and 2019.

The internal budget for water costs that Creative Energy used to support the prior RRAs and actual costs is provided below:¹²⁴

Table 23 Updated and including 2015-2017 budgeted										
502 Steam Expenses - Partial	2015 Actual	2015 Budgeted	2016 Actual	2016 Budgeted	2017 Actual	2017 Budgeted	2018 Actual	2018 Approved	2019 Projected	2020 Test Year
Water	500,997	525,267	580,806	469,090	617,733	492,353	<mark>661,905</mark>	502,200	696,145	743,622

Actual costs were higher than budget in 2016, 2017 and 2018 by an average amount of approximately \$130,000, or 27 percent as detailed in the table above.¹²⁵

Creative Energy submits that water expenses are challenging to budget and forecast due to the underlying lag and uncertainty in the City of Vancouver's water rates and the lack of historical data which creates limitations when building a forecast of water volumes.¹²⁶

Creative Energy states that the issue is a persistent, material, and asymmetrical variance from budgeted amounts of the water expense due to factors outside of its control.¹²⁷

¹²¹ Retrieved from: https://www.bcuc.com/Documents/Guidelines/2017/05-03-2017_RegulatoryAccountFilingChecklist.pdf.

¹²² Exhibit B-1, p. 28

¹²³ Exhibit B-1, p. 29

¹²⁴ Exhibit B-9, BCUC IR 45.1.

¹²⁵ Creative Energy Final Argument, p. 22

¹²⁶ Exhibit B-4, BCUC IR 10.1.1.

¹²⁷ Exhibit B-9, BCUC IR 45.4

Additionally, it argues that the proposed WCDA will not impact the utility's incentive to improve steam plant efficiency or contain costs and cites previous projects which improved efficiency and safety as evidence of their independence.¹²⁸ Instead, the proposed WCDA will reduce its exposure to a material risk which is beyond its control and will not impact its approved ROE.¹²⁹

No alternatives to the proposed deferral treatment for any water cost variances were submitted during the proceeding, but Creative Energy notes that it has implemented a process to track its water volumes more closely with a view to evaluating and improving the rigor of its water expense forecast in future RRAs.¹³⁰

Position of the Parties

The CEC does not object to the establishment of the WCDA and accepts that there is no requirement to adjust the ROE as a result.¹³¹

Panel Determination

The Panel approves the establishment of a non-rate base Water Cost Deferral Account to capture water cost forecast variances for the 2020 test year only. Since a process to better track water volumes has been implemented, Creative Energy is directed to address the continued need for the Water Cost Deferral Account and to re-evaluate its water cost forecast methodology in its next revenue requirement application. Creative Energy must provide in that application information comparing the forecast, approved and actual historical water expenses and volumes to assist in the timely review of the application.

The Panel also approves Creative Energy's proposal for a one-year amortization period on the Water Cost Deferral Account commencing January 1, 2021, and a carrying cost equal to Creative Energy's short-term debt rate, unless otherwise determined by the BCUC. Creative Energy's evidence demonstrates that water costs are primarily outside of its control and have been difficult to forecast and the Panel agrees that the requested deferral treatment is reasonable. The Panel notes that the CEC does not object to this proposed treatment.

3.2.2 Property Tax Deferral Account

Creative Energy seeks approval to establish a Property Tax Deferral Account to record variances between total annual actual property taxes and forecast property taxes for the Core Steam System.¹³² Property taxes are paid to the City of Vancouver for the Core Steam System properties located at 720 Beatty Street and 701 Expo Boulevard.

Creative Energy explains that the annual property tax expense is a function of the value of the properties (as assessed annually by BC Assessment) multiplied by various levy rates, which are externally determined and typically changed annually.¹³³ It submits that property taxes are challenging to forecast due to lag and uncertainty in the levy rates, which are not confirmed until May/June of the applicable year (e.g. mill rates for property taxes for 2020 are not known until May/June 2020).¹³⁴

¹²⁸ Exhibit B-9, BCUC IR 45.3

¹²⁹ Exhibit B-9, BCUC IR 45.4

¹³⁰ Creative Energy Final Argument, pp. 22–23

¹³¹ CEC Final Argument, p. 22

¹³² Creative Energy Final Argument, p. 5.

¹³³ Creative Energy Final Argument, p. 24; Exhibit B-1, p. 34.

¹³⁴ Creative Energy Final Argument, p. 25; Exhibit B-4, BCUC IR 17.2.

Creative Energy proposes to amortize the Property Tax Deferral Account over one year and accrue a carrying cost based on its short-term interest rate, consistent with other deferral accounts, on the mid-year value.¹³⁵

Position of the Parties

The CEC has no objection to Creative Energy's proposal for the property tax variance.¹³⁶

Panel Determination

The Panel approves the establishment of a non-rate base Property Tax Deferral Account to capture property tax forecast variances. The Panel also approves Creative Energy's proposal for a one-year amortization period on the Property Tax Deferral Account commencing January 1, 2021 and a carrying cost equal to Creative Energy's short-term debt rate on the mid-year value, unless otherwise determined by the BCUC. Creative Energy's evidence demonstrates that property tax expense is primarily outside of its control and is difficult to forecast. The Panel agrees that the requested deferral treatment is reasonable. The Panel notes that the CEC does not object to this proposed treatment.

3.2.3 Third Party Regulatory Costs Deferral Account

As discussed in section 2.0, the Panel approves the continuation of the TPRCDA for 2019, sets a two-year amortization period for 2018 forecast variances, and allows for the addition of the 2019 forecast variance of \$177,340 to the TPRCDA. As a result of these determinations, the following variances remain in the TPRCDA for 2020.

Description	Amount			
Unamortized 2018 forecast variance	\$	103,830		
2019 forecast variance		177,340		
Total	\$	281,170		

Table 18 – 2020 Opening TPRCDA Balance

Below the Panel determines whether deferral treatment for 2020 regulatory costs forecast variances remains appropriate and sets the amortization period for the unamortized 2018 and 2019 forecast variances (plus interest) recorded in the TPRCDA. The Panel addresses Creative Energy's proposals related to the DARR mechanism in subsection 3.2.5.

Position of the Parties

The CEC does not comment on the continuation of variance treatment or amortization period for regulatory costs for 2020.

In reply to the CEC's comments on the TPRCDA, Creative Energy states, "the costs at issue are by their nature necessarily incurred for utility purposes and the large majority of the costs are not within management's ability to control and therefore should not be the shareholder's risk." As noted in subsection 2.2, Creative Energy submits regulatory costs are undoubtedly recoverable in utility rates, and deferral account treatment of variances between forecast and actual costs are appropriate between forecast and actual costs because it is very challenging to accurately forecast these costs, utility management has limited or no ability to control variances, and the variance can be material.¹³⁷

¹³⁵ Creative Energy Final Argument, p. 25.

¹³⁶ CEC Final Argument, p. 22.

¹³⁷ Creative Energy Reply Argument, p. 2.

Panel Determination

The Panel approves the continuation of the non-rate base Third Party Regulatory Cost Deferral Account for 2020 to capture third-party regulatory costs forecast variances. As noted in its approvals for 2019 rates, variance treatment for third-party regulatory costs is consistent with previous BCUC decisions and Creative Energy has established that these costs are outside the control of Creative Energy and are difficult to estimate.

As discussed in section 2.0, Creative Energy proposes to amortize the TPRCDA balance over two years and the Panel determines this is appropriate for the 2018 variance. While a two-year period was appropriate for 2019 rates for the reasons stated, the Panel considers that it is generally more appropriate to recognize regulatory costs variances over a one-year period, consistent with Creative Energy's previous treatment. Given the mitigation of the rate impact resulting from the offset of the 2018 and 2019 pension remeasurement amounts, the rate impacts of recognizing the remaining 2018 and 2019 regulatory costs over one year are acceptable. The Panel notes Creative Energy's estimate that a one-year amortization period would result in an overall 2020 rate impact that would exceed 10 percent¹³⁸ is calculated before the pension remeasurement offset. The Panel estimates the impact of using a one-year period is between 4 and 5 percent.¹³⁹ Accordingly, for 2020, the Panel approves a one-year amortization period for the TPRCDA, and Creative Energy is directed to recognize amortization of \$281,170 plus interest for 2020.

3.2.4 Pension Expense Deferral Account

Plus: 2019 Remeasurement Gain

Plus: 2019 Current Service Cost Variance

2020 Opening Pension Expense Deferral Account Balance

Following the Panel's determinations in section 2.0 related to pension expense variances, the following balances remain in the Pension Expense Deferral Account as at the beginning of the year, including 2019 current service cost variances:¹⁴⁰

Description	Amount
2018 Remeasurement Loss	\$(403,700)
2018 Remeasurement Loss	\$(403,700

\$446,200

\$56.232

\$98,732

Table 19 – 2020 Opening Pension Expense Deferral Account Balance

Below the Panel determines whether deferral treatment for 2020 pension expense variances remains appropriate and sets the amortization period for the 2020 opening Pension Expense Deferral Account balance. The Panel addresses Creative Energy's proposals related to the DARR mechanism in subsection 3.4.5.

Position of the Parties

The CEC does not raise any issues regarding the Pension Expense Deferral Account.¹⁴¹

Panel Determination

The Panel approves the continuation of the non-rate base Pension Expense Deferral Account for 2020 to capture pension expense forecast variances. As noted in its approvals for 2019 rates, variance treatment for pension expense is consistent with previous BCUC decisions, and Creative Energy has established that these costs are outside the control of Creative Energy and are difficult to estimate.

¹³⁸ Creative Energy Final Argument, p. 20.

¹³⁹ Calculated as: \$281,170 related to the 2020 opening TPRCDA balance plus \$98,732 related to the 2020 Pension Expense Deferral Account divided by the 2020 approved steam load forecast of 1,140,634 pounds of steam, compared to 2019 approved permanent rates.

¹⁴⁰ Exhibit B-9, BCUC IR 54.9.

¹⁴¹ CEC Final Argument, p. 22.

In the Panel's view, given the mitigation of the rate impact resulting from the offset of the 2018 and 2019 pension remeasurement amounts as well as the favourable current service cost variance, amortization of the account over one year, including interest, is appropriate. Recognition of this amount offsets the rate impact of regulatory cost variances as discussed above. Accordingly, for 2020, the Panel approves a one-year amortization period for the Pension Expense Deferral Account, and Creative Energy is directed to recognize amortization of \$98,732 plus interest.

3.2.5 DARR Mechanism

In this subsection, the Panel considers whether the establishment of a DARR mechanism is warranted, and if so, what the DARR rate should be in this test period. The DARR rate proposed by Creative Energy is \$0.29 per thousand pounds of steam.

Creative Energy requests permanent approval to use a DARR mechanism. Creative Energy describes the DARR mechanism as the implementation of a single DARR, which will span the recovery of costs across multiple deferral accounts using the amortization period determined in respect of each account.¹⁴²

The deferral accounts which Creative Energy proposes to include in the DARR mechanism are:¹⁴³

- The existing Pension Expense Deferral Account, with refunds/recoveries from this account to be applied through the DARR rate beginning in 2020;¹⁴⁴
- The existing TPRCDA, with refunds/recoveries from this account to be applied through the DARR rate beginning in 2020;¹⁴⁵
- The proposed Water Cost Deferral Account, with refunds/recoveries from this account to be applied through the DARR rate beginning in 2021;¹⁴⁶ and
- The proposed Property Tax Deferral Account, with refunds/recoveries from this account to be applied through the DARR rate beginning in 2021.¹⁴⁷

Review of the specifics of these accounts and consideration of whether forecast variance treatment is appropriate and determination of the amortization or recovery period for the variance are set out in subsections 3.2.1 to 3.2.4.

Creative Energy submits that the proposed DARR mechanism is a straightforward and accepted means to recover discrete costs approved for deferral account treatment separate from base rates. In its view, the DARR mechanism:

- Supports transparency and rate and revenue stability;
- Provides the ability to readily adjust the level of the rate rider in response to emergent changes in costs or when subject account balances have been cleared;¹⁴⁸ and
- Promotes equitable allocation of cost recovery from customers.

¹⁴² Creative Energy Final Argument, pp. 6, 18.

¹⁴³ Creative Energy is not in favour of this approach, but states in Option 2 of the PARQ Hot Water Plant steam meter issue, that the DARR could also be used to incorporate the impacts (refund) of an increase to 2019 Core Steam system revenues in 2021 (Exhibit B-9, BCUC IR 59.2; Creative Energy Final Argument, p. 32).

¹⁴⁴ Exhibit B-9, BCUC IR 54.4.

¹⁴⁵ Exhibit B-4, BCUC IR 23.11 to 23.11.1.

¹⁴⁶ Exhibit B-4, BCUC IR 10.3; Creative Energy Final Argument, pp. 18, 22.

¹⁴⁷ Exhibit B-9, BCUC IR 49.1.1; Creative Energy Final Argument, pp. 18, 25.

¹⁴⁸ Creative Energy Final Argument, p. 19.

Creative Energy explains the above are overall benefits relative to recovering the deferral account balances through an adjustment to the Core Steam System's declining block rate design. In the alternative, it submits that larger customers will receive a relative discount on their contribution to the recovery of the costs recorded in the subject deferral accounts.¹⁴⁹ Creative Energy states that it also considered whether a basic recovery charge which would be the same for each Core Steam System customer would be appropriate (e.g. \$/day), but regarded that to be unfair to small customers.¹⁵⁰

Position of the Parties

The CEC does not object to the approach of recovering costs on a per unit basis (\$/thousand pounds of steam) as proposed, agreeing that it is more equitable for customers because it is based on steam usage. The CEC also submits a two-year time frame is appropriate for recovery and recommends the BCUC approve the DARR as filed.¹⁵¹

Panel Determination

In considering whether the establishment of a rate rider mechanism to recover forecast variances in O&M is warranted, the Panel disagrees with Creative Energy's submissions that the proposed DARR mechanism is a "straightforward and accepted means"¹⁵² to recover the forecast variances related to pension, regulatory and other recurring non-controllable O&M costs. Typically, the type of O&M forecast variances captured in deferral accounts is recovered in subsequent revenue requirements within the existing approved rate structure.

In support of its view, Creative Energy references the BCUC directive for Creative Energy to establish a FCSA and rate rider mechanism¹⁵³ to record variances between forecast and actual fuel costs and a fuel cost rate rider mechanism.¹⁵⁴ However, it has not provided any evidence that its proposed rate rider mechanism is an accepted means of recovering pension, regulatory and other recurring non-controllable O&M costs variances. While the Panel agrees that the FCSA and related rate rider mechanism is a typical and an understood approach used to capture and recover fuel cost variances, there is no evidence to support this approach for pension, regulatory and other recurring non-controllable O&M costs.

Creative Energy also submits that a rate rider mechanism provides the ability to quickly adjust the level of the rate rider and, if applicable, cancel the rate rider when the subject costs have been recovered.¹⁵⁵ Given that a forecast of these costs will form part of a revenue requirement application, the Panel considers that recovery of related variances can be readily accommodated within the ongoing RRA process. Any issues related to rate and revenue stability can also be dealt with within the context of an RRA proceeding and approved rate structure.

The Panel disagrees with Creative Energy's submission that its proposed rate rider mechanism supports transparency.¹⁵⁶ Recovering multiple items through one rate rider is not transparent, lacks clarity and adds complexity. Rate rider mechanisms can be useful and transparent when designed to recover a specific item or type of cost, but combining various accounts diminishes the comprehension of a rate mechanism.

The Panel is concerned with the submissions that the rate rider mechanism results in a more equitable rate design for recovery of certain O&M forecast variances. Creative Energy and the CEC do not address why it is appropriate (i.e. not unjust, unreasonable, unduly discriminatory or unduly preferential) for forecast amounts

¹⁴⁹ Exhibit B-4, BCUC IR 23.5; Creative Energy Final Argument, pp. 4, 19-20.

¹⁵⁰ Exhibit B-4, BCUC IR 23.2.

¹⁵¹ CEC Final Argument, p. 21.

¹⁵² Creative Energy Final Argument, p. 18; Exhibit B-4, BCUC IR 23.1.

¹⁵³ 2016-2017 Core Steam System and NEFC Decision, p. 30.

¹⁵⁴ Exhibit B-4, BCUC IR 23.1.

¹⁵⁵ Exhibit B-4, BCUC IR 23.1.

¹⁵⁶ Exhibit B-4, BCUC IR 23.1.

related to pension, regulatory, water costs and property taxes to be recovered through the existing declining block rate design, but on the other hand, the related variances from forecast would be more equitably recovered through the proposed DARR mechanism. If Creative Energy has identified potential inequities among customers within the existing rate design, it should undertake a cost of service study and assess if its existing rate design reflects the costs of providing steam service to those customers.

For the reasons outlined above, the Panel denies Creative Energy's request for permanent approval of its proposed DARR mechanism. Creative Energy is directed to provide, in its Compliance Filing the calculation of and proposed timing by which customers will be refunded the amounts collected in 2020 on an interim basis, plus interest, by way of bill credits. Creative Energy is also directed to record the amortization of the Pension Expense Deferral Account and Third Party Regulatory Costs Deferral Accounts in the revenue requirement for 2020 Steam Rates.

3.3 Proposals regarding the Fuel Switch Study and LTRP Deferral Account

3.3.1 Background

As part of Order G-7-20A, the BCUC directed Creative Energy to file a proposal for the potential recovery of the balance in its Fuel Switch Study and LTRP Deferral Account for review as part of this Application.¹⁵⁷ The BCUC initially established the Fuel Switch Study and LTRP Deferral Account in the 2018-2022 Core Steam System Decision issued on October 25, 2018, following review of that proceeding.

As part of its 2018-2022 RRA, Creative Energy sought approval to include \$720,232 of costs incurred on account of its 2017 LTRP Application (LTRP Application) into the TPRCDA, which the BCUC had approved as part of Creative Energy's 2016-2017 RRA. Those costs included amounts expended on the Fuel Switch Study, which was filed as part of Creative Energy's LTRP Application. However, the Fuel Switch Study predated the BCUC directive made pursuant to Order G-98-15 on June 9, 2015 directing Creative Energy to file the LTRP and to include the information from the Fuel Switch Study as part of that proceeding.

Following a review of the LTRP Application, the BCUC adjourned the proceeding until the filing of Creative Energy's 2020-2021 RRA. The BCUC directed Creative Energy "to file a complete and updated LTRP that satisfies all requirements under section 44.1 of the UCA" and adjourned the review of that application pending such filing.¹⁵⁸ In so doing, the BCUC made the following findings:¹⁵⁹

The Panel finds that it is not reasonable to review Creative Energy's LTRP at this stage because the Fuel Switch Program, which is central to its LTRP, has not sufficiently progressed and is still in the preplanning stage. Creative Energy is still in the midst of pursuing enabling mechanisms that would advance the Fuel Switch Program and it will continue to pursue the enabling mechanisms for the next two years. However, based on its Application, Creative Energy is currently not in the position to commit to the Fuel Switch Study and thus lacks important information regarding details, schedules and impacts that would normally be reviewed within the LTRP. Absent of the commitments that Creative Energy needs in the next two years, the Fuel Switch Program may not continue. As such, the Panel does not believe Creative Energy is able to provide the full information required to facilitate a complete review of its LTRP.

During the BCUC's review of Creative Energy's 2018-2022 RRA, the BCUC determined in light of the adjournment of the LTRP Application, there was limited evidence which would substantiate whether the Fuel Switch Study was or would be useful in preparing Creative Energy's LTRP and whether the associated expenditures incurred between 2013 and 2017 were prudently and reasonably incurred for the benefit of Core steam customers such

¹⁵⁷ Order G-7-20A, Directive #7.

¹⁵⁸ Creative Energy 2017 LTRP, Order G-147-17 with Reasons for Decision dated September 25, 2017, Directive #1.

¹⁵⁹ Ibid, Appendix A, p. 3.

that they ought to be approved for recovery. Accordingly, the BCUC declined to make any determination on the recoverability of the costs proposed for deferral treatment, but noted in doing so the following evidence:

- In its 2015-2017 RRA, Creative Energy forecast \$0 for LTRP costs for 2015;
- In its 2016-2017 RRA, it forecast \$0 for LTRP costs for 2016 and \$48,000 for 2017, on account of "Consultants – LTRP", which amount was approved for inclusion in the approved 2017 forecast for Special Services expenses, with any variances between forecast and actual costs approved to be added to the TPRCDA;
- By the time of the 2016-2017 RRA, Creative Energy had purportedly spent more than \$472,000 on the Fuel Switch Study and LTRP, but this was not disclosed to the BCUC except for the \$48,000 forecast expenditure for 2017 for "Consultants LTRP" referred to above;
- By the time of the 2018-2022 RRA, the amount claimed for deferral related to the Fuel Switch Study and LTRP costs was \$720,232, net of offsetting government grants; and
- The majority of the costs claimed comprised consulting fees paid to 15 different firms between 2013 and 2017 entailing "hundreds of associated invoices" for which no supporting detail was provided to the BCUC despite its request to Creative Energy to provide same.¹⁶⁰

In light of this evidence, rather than acceding to Creative Energy's proposal during the RRA to include all of the associated costs into the TPRCDA, the BCUC directed that Creative Energy transfer all of the costs claimed with respect to the Fuel Switch Study and LTRP, net of all applicable offsetting grants, to a new Fuel Switch Study and LTRP Deferral Account pending the filing of an updated LTRP. The BCUC found that the TPRCDA "is only intended to capture ongoing Third Party Regulatory Costs incurred by Creative Energy in the ordinary course for its regulatory filings, as evidenced by the one-year amortization period approved in the 2016-2017 RRA" and is not meant to capture extraordinary costs of the nature and magnitude claimed on account of the Fuel Switch Study and LTRP, which span four years from 2013 to 2017.¹⁶¹

In making this finding, the BCUC also noted that "separation of the Fuel Switch Study costs from the TPRCDA will allow for greater transparency when these costs are being reviewed" by a subsequent panel.¹⁶² The BCUC observed that "it would be appropriate for the subsequent panel assigned to review the updated LTRP to assess the reasonableness of the Fuel Switch Study and LTRP costs, as that panel will have a better understanding of the usefulness and applicability of the Fuel Switch Study to the overall LTRP filing."¹⁶³ This would enable a subsequent RRA panel to assess the recovery mechanism for these costs based on the recommendations and determinations of the panel reviewing the updated LTRP.¹⁶⁴ At the same time, the BCUC approved carrying costs on the balance in the new deferral account at a weighted average cost of debt. The BCUC further directed Creative Energy to submit a proposal for the disposition of that account at the earlier of: (i) the filing of an updated LTRP by December 31, 2019, pursuant to Creative Energy's stated intentions; and (ii) January 31, 2020 (Directive #5).¹⁶⁵

In so ordering, the BCUC made the following observations:

The Panel observes that Creative Energy ought to have been more forthcoming and transparent about the extent of the Fuel Switch Study and LTRP expenditures when it applied for and obtained approval for the TPRCDA in the 2016-2017 RRA Decision. In the Panel's view, it is unreasonable for Creative Energy

¹⁶⁰ 2018-2022 Core Steam System Decision, pp. 28-29.

¹⁶¹ 2018-2022 Core Steam System Decision, p. 32.

¹⁶² Ibid.

¹⁶³ Ibid.

¹⁶⁴ Ibid.

¹⁶⁵ Order G-205-18, Directive #5.

to withhold relevant information about significant expenditures which it expects ratepayers to bear in subsequent RRAs.¹⁶⁶

The BCUC directed Creative Energy to submit a compliance filing showing:

- The updated balance in the TPRCDA to reflect the removal of the "Consultants LTRP" costs which the BCUC had already approved for recovery as part of Creative Energy's forecast for Special Services in 2017; and
- The newly established Fuel Switch Study and LTRP Deferral Account and associated cost additions.

On December 11, 2018, Creative Energy submitted a compliance filing showing the updated balance in the latter account totalled \$714,880 as of that date.

As part of this Application, Creative Energy initially applied for formal relief from Directive #5 of Order G-205-18 along with a request for extension of that Directive to December 31, 2020 for filing of its updated LTRP, along with a proposal for disposition of the new deferral account. It attributes this delay to resource constraints and the ongoing uncertainty with respect to fuel switch planning sources, which prevent it from having an updated LTRP. Instead, Creative Energy proposed to address the recovery of the balance in the Fuel Switch Study and LTRP Deferral Account at the time of the filing of an updated LTRP as part of its 2021 RRA.

Following a review of that request, the Panel ordered Creative Energy to file a proposal for the disposition of the balance in the Fuel Switch Study and LTRP Deferral Account for review as part of this Application. In doing so, the Panel found that given the clear wording in Directive #5, Creative Energy's reasons for the Extension Request did not warrant the length of the requested extension, noting that as of the time of this request, Creative Energy will have had more than 15 months from the issuance of that Directive to either file an updated LTRP or to separately file a proposal to address the potential recovery of the balance in the deferral account. Nonetheless, the Panel exercised its discretion to extend the filing deadline for the proposal from January 31, 2020 to February 21, 2020 in recognition of Creative Energy's stated resource constraints.¹⁶⁷

3.3.2 Creative Energy's Proposal

Pursuant to Order G-7-20A, Creative Energy filed the following proposal on February 27, 2020 for BCUC approval of the disposition of the current balance in the deferral account of \$735,223 (before carrying costs) for three categories of costs:¹⁶⁸

- That the BCUC approves \$214,185 for recovery by Creative Energy from its Core steam customers as
 part of this Application for costs incurred in 2016 on account of consultant costs to pursue low-carbon
 development, with approval of the proposed rate mechanism (including billing determinants and
 recovery period) to be brought forward as part of Creative Energy's 2021 RRA, taking into account
 considerations such as customer bill impacts and the context and drivers of the 2021 RRA which are yet
 to be determined;¹⁶⁹
- That the total costs of the internal project management time and related support currently recorded in the deferral account spanning 2015 to 2016, totalling \$64,222 and \$39,314 respectively, and \$103,536 collectively, be excluded from recovery from ratepayers. Creative Energy makes this concession, stating that the 2015 costs have already been included in the project related costs of \$417,502 associated with the Fuel Switch Project (namely, the assessment, definition and preliminary design of the "Green House Project" described in the Fuel Switch Study) which Creative Energy does not intend to proceed with at this time. With respect to the 2016 costs, it is unable to definitively confirm they were on account of the

¹⁶⁶ 2018-2022 Core Steam System Decision, p. 33

¹⁶⁷ Order G-7-20A, Appendix B, pp. 2–3.

¹⁶⁸ See Exhibit B-3, p.3.

¹⁶⁹ Exhibit B-3, p. 2.

Fuel Switch Project versus other efforts, and whether these costs were necessarily or properly excluded from its revenue requirements and rates at that time;¹⁷⁰ and

• That the BCUC defer consideration of the disposition of the remaining balance of the deferral account, \$417,502, until such time as and when Creative Energy advances "a proposal to capitalize any related development costs into the overall [Green House] project as applicable".¹⁷¹

As part of its proposal, Creative Energy provided the following table summarizing the current Fuel Switch Study and LTRP Deferral Account balance net of project-related grants.

	Fu	el Switch Proje	Enabling Low Carbon Development	
	2013	2014	2015	2016
Lead Consultant	18,060	54,568	191,228	214,185
Other Consultant Support	11,039	70,968	288,639	n/a
Creative Energy Internal Project Management, Executive & Legal Support	-	-	64,222	39,314
Subtotal	29,099	125,536	544,089	253,499
Total			698,724	n/a
Project-related Grants			-217,000	n/a
Net Total			481,724	n/a
Fuel Switch Project & LTRP Deferral Account Net of Project-related Grants				735,223 ²
Creative Energy Proposed Recovery			n/a	214,185

As the table above shows, Creative Energy now distinguishes between the costs incurred from 2013 to 2015 which it attributes to costs incurred on account of the pursuit of a Fuel Switch Project and those incurred in 2016 in pursuit of enabling low-carbon development.

Creative Energy's proposal raises the following issues for the Panel:

- With respect to the first category of costs, totalling \$214,185, which Creative Energy now ascribes to the pursuit of low-carbon development in 2016 (see the last column in Table 20 above), do those costs form part of general O&M costs that ought to have been forecast and recovered as part of rates in 2016 such that Creative Energy is now precluded from recovery of those costs from current or future ratepayers?
- With respect to the second category of costs totalling \$103,536, should these costs be written off as Creative Energy proposes?
- With respect to the third category of costs totalling \$417,502,¹⁷² do these "development" costs warrant deferral treatment until such time if and when Creative Energy brings forward an application to capitalize those costs in connection with an anticipated overall capital project?

¹⁷⁰ Ibid.; see also Creative Energy Reply Argument, pp. 2, 9.

¹⁷¹ Ibid.

¹⁷² Calculated as \$481,724 less \$64,222 in the middle column(s) of Table 20.

3.3.3 Discussion of the Issues

The Panel summarizes below Creative Energy's and the CEC's submissions with respect to the above-noted issues.

Creative Energy describes the first category of costs as being:

...entirely Creative Energy's cost for the efforts of its consultants – during the year 2016 only – that were directed toward supporting Creative Energy's ongoing resource planning efforts in view of it becoming increasingly imperative to decarbonize the energy supply for the Core steam system to meet current load and future customer growth.¹⁷³

At the same time,

Creative Energy acknowledges that these costs were at one point in time expected to be capitalized as part of the overall development costs of the Fuel Switch Project. Creative Energy does not intend to proceed with the Fuel Switch Project at this time; however, we continue to study the technical and financial viability of alternative projects for displacing natural gas in the steam plant. The efforts in 2016 support the ongoing good utility planning and practice and are applicable expenses for recovery from customers.¹⁷⁴

Creative Energy goes on to explain that the work completed in 2016 focused on two areas of long-term resource planning: consideration of government policy and stakeholder engagement. It distinguishes the nature of these costs in 2016 from the third category of costs (the Fuel Switch Project costs) incurred between 2013 and 2015 in these terms:

The focus in 2016 was on the policy environment, enabling tools, and the opportunities and challenges for decarbonization generally, as distinct from the prior detailed feasibility study of the Fuel Switch Project.¹⁷⁵

It further points out that one outcome of the work in 2016 is that the "results of stakeholder engagement efforts were incorporated into section 13 of the Fuel Switch Study final report."¹⁷⁶

With respect to the first category of costs (the 2016 costs), the CEC notes its concern that the consultant Reshape Infrastructure Strategies (Reshape) was not selected through a Request for Proposal (RFP) process and there is a non-arms-length relationship between the parties in that the principal of Reshape (T. Berry) was a past director and Chair of Creative Energy. As a result, the CEC recommends the write-down of a "good portion" of the costs. At the same time, though, the CEC also states that it does not object to Creative Energy's proposal and that it finds the Reshape invoices "acceptable". The CEC, however, goes on to suggest that the BCUC direct Creative Energy to conduct appropriate RFP processes for major expenditures for outside consulting services in the future.¹⁷⁷

In response, Creative Energy submits that the recommendation has no foundation and is unnecessary.¹⁷⁸

As for the second category of costs (2015 and 2016 internal project management and similar costs), the CEC agrees that these costs should be written off as Creative Energy proposes.¹⁷⁹

¹⁷³ Exhibit B-3, p. 3.

¹⁷⁴ Ibid.

¹⁷⁵ Ibid., p. 4.

¹⁷⁶ Ibid., p. 5.

¹⁷⁷ CEC Final Argument, p. 32.

¹⁷⁸ Creative Energy Reply Argument, pp. 2,9.

The CEC does not specifically address Creative Energy's proposed disposition of the third category of costs constituting the remaining balance in the Fuel Switch Study and LTRP Deferral Account.

Panel Determinations

2016 Enabling Low-Carbon Development Costs

In the 2018-2022 RRA, the BCUC directed the establishment of a new deferral account, the Fuel Switch Study and LTRP Deferral Account, pending review of the disposition of the costs initially proposed by Creative Energy to be recorded in the TPRCDA. The Panel notes that contrary to Creative Energy's evidence in the previous RRA that all of the costs proposed for deferral into the TPRCDA were on account of the Fuel Switch Study and LTRP, Creative Energy now submits that \$214,185 of those amount expended in 2016 relate to consulting costs to enable low-carbon development. Creative Energy describes those 2016 costs as supporting good utility practice and planning with a focus on government policy and stakeholder engagement.

Accepting this revised explanation for this first category of costs, the Panel finds that such costs are more properly characterized as period O&M expenses and that they have no direct relationship to the development of the LTRP Application. Accordingly, Creative Energy should have recorded the \$214,185 as part of its O&M expenditures in the 2016 RRA period. The Panel notes these costs were not disclosed to the BCUC or included in forecast O&M in the 2016-2017 RRA and appear to have been tracked separately until Creative Energy proposed to include these costs in the TPRCDA as part of the Fuel Switch Study and LTRP costs in Creative Energy's 2018-2022 RRA. As already noted, the BCUC denied the approval of Creative Energy's request, directed that the balance be put into the Fuel Switch Study and LTRP Deferral Account pending the filing of an updated LTRP or RRA.

Creative Energy failed to forecast these 2016 Enabling Low Carbon Development Costs for recovery in 2016 rates. Similarly, it did not apply for BCUC approval of deferral treatment for those expenses in 2016 when it filed its 2016-2017 RRA. For the BCUC to now (four years after the expenditures) to allow Creative Energy to recover them in rates is to engage in retroactive ratemaking, contrary to the well-established regulatory principle against such. As the Supreme Court of Canada stated in *ATCO Gas and Pipelines v. Alberta (Utilities Commission) (ATCO)*: 180

...The Board was seeking to rectify what it perceived as a historic overcompensation to the utility by ratepayers. There is no power granted in the various statutes for the Board to execute such a refund in respect of an erroneous perception of past over-compensation. It is well established throughout the various provinces that utilities boards do not have the authority to retroactively change rates...

Consistent with this reasoning in the *ATCO* decision and as already noted earlier, the BCUC explained in a previous Creative Energy decision that generally, the BCUC sets rates on a prospective basis only and has no authority to allow recovery on a retroactive basis.¹⁸¹ The Panel notes that well-established exceptions to retroactive ratemaking include, in part:

- Setting of interim rates which are subject to later adjustment; and
- Recognition of amounts in deferral accounts to be carried forward to be disposed of in future years.¹⁸²

¹⁷⁹ CEC Final Argument, p. 31.

^{180 2006} SCC 4, at para. 70 retrieved from: https://scc-csc.lexum.com/scc-csc/scc-csc/en/17/1/document.do,

¹⁸¹ 2016-2017 Core Steam System and NEFC Decision, p. 13.

¹⁸² Union Gas Limited v. Ontario Energy Board, 2015 ONCA 453.

Given this, Creative Energy has not satisfied the Panel that the 2016 consulting expenditures made by Creative Energy in pursuit of enabling low-carbon development ought to be recoverable from current or future ratepayers. The Panel acknowledges that the BCUC, as part of Creative Energy's 2018-2022 Core Steam System Decision, directed that the costs claimed to be put into the Fuel Switch Study and LTRP Deferral Account, but it did so only in contemplation that the BCUC would subsequently determine the recovery of those costs in connection with Creative Energy's filing of an RRA or updated LTRP that would determine the disposition of that temporary regulatory account. Accordingly, **the Panel denies Creative Energy's request for approval to recover from customers of the Core Steam System the \$214,185 spent in 2016, plus interest , along with Creative Energy's request to defer the mechanism and timing for such recovery for determination in the 2021 RRA. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account.**

2015 – 2016 Internal Project Management, Executive & Legal Support Costs

The Panel agrees with Creative Energy's proposal to write off all of the costs attributed to "internal project management time, executive and legal support" in 2015 and 2016, and **directs Creative Energy to write off the amount of \$103,536 plus interest from the balance in the Fuel Switch Study and LTRP Deferral Account as Creative Energy proposes.**

2013 – 2015 Fuel Switch Project Costs

According to the evidence adduced in this proceeding, the third category of costs pertains to costs incurred on account of the Fuel Switch Project between 2013 and 2015. These costs total \$417,502. As already noted, Creative Energy acknowledges in this proceeding that these costs were at one point in time expected to be capitalized as part of the overall development costs of the Fuel Switch Project:¹⁸³

...specifically for the assessment, definition and preliminary design of the 'Green House Project' described in the Fuel Switch Study included with the 2017 LTRP...

The Fuel Switch Project-related efforts were directed by Creative Energy's parent company, Creative Energy Canada, upon its acquisition of Creative Energy (then called Central Heat Distribution Ltd.) and were responsive in general to an increasing public policy imperative to consider low-carbon options to serve current and future heating demand in the City of Vancouver.

Creative Energy further acknowledges that it does not intend at this time to proceed with the Green House project and Creative Energy, therefore, proposes not to recover from existing customers any of the amount in the Fuel Switch Study and LTRP Deferral Account related to this project at this time. However, if this or a similar project is pursued in the future, Creative Energy will at that time advance a proposal to capitalize any related development costs into the overall project as applicable.

Creative Energy's evidence in this proceeding is that these costs were incurred between 2013 and 2015 and relate to research into the feasibility of a Fuel Switch Project. Since these costs were research related and not development or capital expenses in nature, the Panel finds that these costs are 2013 to 2015 period costs. Creative Energy did not request recovery of these costs in its RRAs for 2013 through 2015 or otherwise request deferral treatment of anticipated feasibility costs related to the Fuel Switch Project in the years in question, and the revenue requirements for those years have been set and long since expired. Therefore, the Panel finds that it would be inappropriate to allow Creative Energy to recover those costs now or continue to hold them in the deferral account for future recovery from ratepayers as that would amount to prohibited retroactive ratemaking. Accordingly, the Panel denies Creative Energy's request to maintain the balance of \$417,502 plus

¹⁸³ Exhibit B-3, p. 2.

interest in the Fuel Switch Study and LTRP Deferral Account and to defer for consideration the disposition of such costs until such time as a CPCN for a low-carbon energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Account to the account of the shareholder and further directs Creative Energy to close the deferral account thereafter.

3.4 Setting 2020 Steam Rates

Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel approves a rate increase inclusive of the impact of its decision above, including:

- Directive to recalculate the costs allocated to the Core Steam System and NEFC, respectively, based on the three-factor Massachusetts Formula approved above in its Compliance Filing and including the rate impact for each service area;
- Directive to reduce the allowed return for 2020 by \$21,503, based on a mid-year balance of \$268,202 in the After-tax Regulatory Pension Asset Account.
- Denial of the request for permanent approval of its proposed DARR mechanism; and
- Directive to record the amortization of the Pension Expense Deferral Account and Third Party Regulatory Costs Deferral Accounts in the revenue requirement for 2020 Steam Rates.

Based on these adjustments, Creative Energy is directed to recalculate the rate increase and include the updated financial schedules in its Compliance Filing.

4.0 Setting 2019 Rates for NEFC

4.1 NEFC Hot Water Rates

Background

The BCUC approved the revenue requirements, rate design, and Hot Water rates for the NEFC service area in the 2016-2017 Core Steam System and NEFC Decision. The NEFC rate design was set on a levelized rate design basis for 15 years ending in 2030. The decision also established the Revenue Deficiency Deferral Account (RDDA) to record the impact of timing differences between costs incurred to install the required infrastructure to serve NEFC customer load and the revenues from the buildout of that customer load over time. The RDDA allows for a levelized rate structure to smooth rate increases over time, recognizing that the initial rates will not be sufficient to recover forecast revenue requirements. Shortfalls were approved to be added to the balance of the RDDA to be ultimately recovered through load growth and levelized rate increases over a time horizon of 10 years.¹⁸⁴

In the 2016-2017 Core Steam System and NEFC Decision, the BCUC approved a recovery period for the RDDA, commencing in 2020 and ending in 2030, given "the relatively small peak balance expected and [the] revenue streams put forward."¹⁸⁵

The RDDA is intended to record variances between approved revenue requirements and forecast revenues calculated based on approved rates and load forecast. The RDDA is not approved to capture revenue shortfalls based on differences between actual revenues received and approved forecasts. These variances are addressed separately through the Variance Deferral Account, which was also established in the 2016-2017 Core Steam

¹⁸⁴ Creative Energy Final Argument, p. 25.

¹⁸⁵ 2016-2017 Core Steam System and NEFC Decision, p. 69.

System and NEFC Decision.¹⁸⁶ The Variance Deferral Account also captures forecast variances related to the following items outside of management's control:

- Variances between forecast versus actual Steam Service Rates and Fuel Cost charged to NEFC;
- Variances between forecast versus actual Distribution expenses¹⁸⁷; and
- Variances between forecast versus actual Income Tax expense.

The Variance Deferral Account was directed to be amortized over a one-year period and recovered through the hot water rates and was approved for a five-year period ending December 31, 2020, at which time Creative Energy must apply for renewal for the period beginning January 1, 2021.¹⁸⁸

The BCUC approved the following NEFC Hot Water Rates and the opening balance in the RDDA for 2017 in the 2016-2017 Core Steam System and NEFC Decision:

- A fixed Rate of \$0.27 per square meter per month;
- A variable Rate of \$52.10 per megawatt hour (MWh); and
- An addition to the RDDA in the amount of \$373,900.

These rates were maintained in 2018, due to Creative Energy not applying for any changes to NEFC rates for 2018. A corresponding increase of \$373,900 to the RDDA balance in 2018 was recorded.¹⁸⁹ At the end of 2018, the balance in the Variance Deferral Account was a surplus of \$655,250, which Creative Energy transferred to the RDDA in 2019.¹⁹⁰ The 2018 deferral account additions and transfers have not yet been approved by the BCUC and are addressed in the Panel determination below.

In its approval of 2019 interim rates,¹⁹¹ the BCUC denied Creative Energy's application to increase the variable component of NEFC Hot Water Rates by 1.84 percent. Creative Energy was directed to maintain NEFC's fixed and variable rates at existing rates on an interim basis, effective January 1, 2019. In the decision on interim rates, the BCUC found that Creative Energy had not provided adequate evidence to support the change and the most reasonable approach was to maintain NEFC rates at the existing approved rates on an interim basis.¹⁹²

Creative Energy's Proposal

Creative Energy proposes to maintain 2019 NEFC interim Hot Water rates on a permanent basis, consistent with the rates set in the 2016-2017 Core Steam System and NEFC Decision.¹⁹³ It proposes that the difference between the 2019 actual revenue requirement and the revenues based on existing rates be added to its existing RDDA for future recovery, submitting that this approach complies with the BCUC direction not to begin recovery of the RDDA balance until 2020.¹⁹⁴

The proposed revenue deficiency related to 2019 rates is shown in the table below (2018 approved forecast is based on 2017 approved forecast from 2016-2017 Core Steam System and NEFC Decision, since Creative Energy did not apply for changes to 2018 rates):

¹⁸⁶ 2016-2017 Core Steam System and NEFC Decision, pp. 70–71.

¹⁸⁷ Distribution expenses include GL Accounts 870, 874 and 880.

¹⁸⁸ 2016-2017 Core Steam System and NEFC Decision, pp. 70–71.

¹⁸⁹ Exhibit B-1, pp. 44–45.

¹⁹⁰ Exhibit B-9, Schedules for RRA Filing – IR Round 2, NEFC Schedule 12.

¹⁹¹ Order G-248-18.

¹⁹² Order G-248-18, Appendix A, p. 2.

¹⁹³ Creative Energy Final Argument, pp. 4-5.

¹⁹⁴ Creative Energy Final Argument, p. 26.

Table 21 – NEFC Revenue Requirements¹⁹⁵

	2017	2017	2018	2018	2019
Component	Approved	Actual	Approved	Actual	RRA
	Forecast	Actual	Forecast	Actual	Update
Fuel (excluding FCAC Rate Rider)	313,800	282,725	313,800	354,737	550,767
Steam Tariff	222,900	187,360	222,900	223,974	257,991
Subtotal Cost of Sales	536,700	470,085	536,700	578,711	808,758
O&M Expense	121,600	115,742	121,600	243,981	117,406
Municipal Taxes	10,100	10,601	10,100	13,077	18,097
Income Tax Expense	3,700	16,324	3,700	(28,741)	6,148
Depreciation Expense	180,300	-	180,300	96,582	191,400
Return on Rate Base	321,800	134,205	321,800	284,348	315,000
Total Revenue Requirement	1,174,200	746,957	1,174,200	1,187,958	1,456,809
Hot Water Sales Revenues	800,300	848,053	800,300	1,046,191	1,447,745
Revenue (Surplus)/Deficiency	373,900	n/a	373,900	n/a	9,064
FCAC Rate Rider Recovery					

Since Creative Energy proposes to capture the difference between the 2019 actual revenue requirement and the revenues based on existing rates in the existing RDDA for future recovery, this implies that the Variance Deferral Account will not be used in 2019 to capture any forecast variances, as originally directed by the BCUC.

Based on Creative Energy's proposals, the balance in the RDDA is set out in the following table.

	Opening		Opening Additions/ Interest/				Ending		
	Balance		(De	(Deductions)) AFUDC Amortization		Balance	
2017	\$	-	\$	373,900	\$	11,740	\$	-	\$ 385,640
2018	\$	385,640	\$	373,900	\$	35,959	\$	-	\$ 795,499
2019	\$	795,499	\$	9,064	\$	50,242	\$	(661,861)	\$ 192,943

Table 22 – 2017-2018 Actual and 2019 Proposed Revenue Deficiency Deferral Account Balances¹⁹⁶

At the end of 2018, the balance in the Variance Deferral Account was a surplus of \$655,250, which Creative Energy transferred to the RDDA in 2019.¹⁹⁷ Based on Creative Energy's proposals, the balance in the Variance Deferral Account is set out in the following table.

	Opening Additions/			Interest/			Ending		
	Balance		(Deductions)		AFUDC	Am	ortization		Balance
2017	\$ -	\$	(429,063)	\$	(13,472)	\$	-	\$	(442,535)
2018	\$ (442,535)	\$	(179,294)	\$	(33,421)	\$	-	\$	(655,250)
2019	\$ (655,250)	\$	-	\$	(41,150)	\$	661,861	\$	(34,539)

Position of the Parties

The CEC did not comment on the proposed transfer of the Variance Deferral Account, the additions to the RDDA or NEFC Hot Water Rates.

¹⁹⁵ Exhibit B-1-1, page 4.

¹⁹⁶ Exhibit B-9, Schedules for RRA Filing – IR Round 2, NEFC Schedule 12.

¹⁹⁷ Exhibit B-9, Schedules for RRA Filing – IR Round 2, NEFC Schedule 12.

¹⁹⁸ Exhibit B-9, Schedules for RRA Filing – IR Round 2, NEFC Schedule 12.

Panel Determination

The Panel approves Creative Energy's request for permanent approval of NEFC's 2019 interim hot water rates. These rates align with BCUC's previous decisions. Levelized rates were set in the 2016-2017 Core Steam System and NEFC Decision, and the Panel understands that these rates are to remain in place until commencing recovery of the RDDA in 2020, unless determined otherwise by the BCUC.

The Panel acknowledges that the 2016-2017 Core Steam System and NEFC Decision also determined that the Variance Deferral Account balance was to be amortized over a one-year period and recovered or refunded through the hot water rates. However, the decision covering 2017 rates did not specify how this would happen in subsequent test periods. Since Creative Energy did not file an RRA for 2018 and no amortization of the Variance Deferral account was approved to be recovered as part of the 2018 hot water rates, the 2017 approved rates remained in effect for 2018.

Since Creative Energy did not file an RRA for 2018, the Panel agrees it is appropriate to use the 2017 approved forecast to determine the amount to be added to the RDDA and Variance Deferral Account in 2018. Accordingly, the Panel approves the addition of a \$373,900 revenue deficiency plus interest to the RDDA for 2018, consistent with the amount determined by the BCUC in 2017. For 2018, the Panel directs Creative Energy to file as part of its Compliance Filing the amount to be added to the Variance Deferral Account.

As for Creative Energy's request to transfer or amortize an amount from the Variance Deferral Account to the RDDA, the Panel considers this to be inconsistent with the reasons establishing the Variance Deferral Account. The Variance Deferral Account was to capture forecast variances related to non-controllable operating costs that should be recovered over a shorter period and the BCUC determined that period should be one-year. On the other hand, the RDDA was expected to be recovered over ten years. Accordingly, the Panel denies Creative Energy's request to transfer \$661,861 from the Variance Deferral Account to the RDDA in 2019.

The Panel also takes issue with Creative Energy's proposal to record an addition of \$9,064 as shown in Table 21 above to the RDDA for 2019 being the difference between the 2019 actual revenue requirement and the revenues based on existing rates. First, consistent with reasons for the Panel's determinations in section 2.0 on the use of actual results to set rates, the Panel disagrees with the use of actuals to determine the amount of variances to be recognized in deferral accounts. Rates should be set prospectively based on a reasonable forecast of revenue requirement items with deferral treatment for the items are outside the control of management or are subject to a high degree of forecast uncertainty. Second, as noted above, the Panel does not agree with Creative Energy's proposal to combine the RDDA and Variance Deferral Account. **Therefore, the Panel denies Creative Energy's proposal to add the difference between the 2019 actual revenue requirement and the revenues based on existing rates to the existing RDDA.**

The Panel notes that the BCUC found the forecast information filed in the interim rate application was not adequate for setting interim rates, and therefore, the Panel concludes this information should not be used to set permanent rates. Since Creative Energy did not provide a supportable forecast for 2019, **Creative Energy is directed to recalculate the allocation of variances between the Variance Deferral Account and the RDDA based on the 2017 approved forecast. As part of its Compliance Filing, Creative Energy is also directed to include an updated schedule of the revised amounts in the Variance and Revenue Deficiency Deferral Accounts, reflecting the above and other determinations in this Decision.**

4.2 Underreporting of PARQ Hot Water Plant Steam Meter

Creative Energy indicated that the PARQ Hot Water Plant meter was underreporting steam flow from the Core Steam system for the period around March or April 2018 to March 2020. The result of the discrepancy was that the total metered energy consumption of the building connected to the PARQ plant was greater than the energy the PARQ Plant meter indicated was supplied to the buildings for that time period. For added clarity, the revenue meters that measure the energy consumed by the buildings were operating correctly and the issue was limited to the metering of energy delivered from the Core system to the PARQ Hot Water Plant for the NEFC system.¹⁹⁹

The following section discusses the implications of the metering issue as it pertains to the 2019 Test Year. For further discussion on the impact of this issue for the 2020 Test Year please see subsection 3.1 of this decision.

Based on the plant and distribution efficiencies of the NEFC system, it was estimated that the meter was recording about 50 percent of the steam that would be expected.²⁰⁰ The estimated difference between actual metered PARQ plant steam load and the expected amount based on observed system efficiencies is 30,678 thousand pounds of steam.²⁰¹

Creative Energy submits that the matter at issue is when and how to recognise the additional steam supplied by the Core Steam system to the NEFC system for the purposes of the revenue requirements.²⁰²

Creative Energy proposes to address the matter prospectively in 2020 by recognizing additional steam in the 2020 steam load forecast and proposes to make no adjustments for 2019.

Creative Energy submits that if the underreported steam is incorporated in its 2019 NEFC steam load forecast, then the following impacts would occur:

- The steam load billed from the Core Steam System to NEFC would increase by 30,678 thousand pounds of steam;
- NEFC's steam costs would increase by \$187,807 and fuel costs would increase by \$497,728. The impact on NEFC revenue requirements would be either an increase in the RDDA balance from \$7,230 to \$692,764, or an average increase of 74% to the variable rate for 2019 on account of the increased cost in 2019; and
- Core Steam System revenues would increase as a result of the additional revenues transferred from the NEFC which would either result in a reduction in the 2019 addition to the proposed DARR if current rates are maintained or, alternatively, Steam Rates would be reduced for 2019.²⁰³

The options provided by Creative Energy to incorporate the underreported steam are either to collect the undercharged amount through a one-time surcharge or by allocating the additional NEFC cost of service to the RDDA for future recovery and correspondingly updating the proposed DARR to incorporate the negative deferral balance addition for the Core over an appropriate amortization period.²⁰⁴

Creative Energy justifies its proposed option to address the issue beginning in 2020 on the basis that correction to the steam load forecast on a go-forward basis is reasonable and appropriately aligned with a future planned rate design application. Creative Energy further submits that the option to flow the amount into the RDDA (and DARR) is also practical but notes that this approach does not align with accepted ratemaking principles that support the fair allocation of costs that match cost recovery with cost causation. Further, it was noted that the mechanism to recover the RDDA balance and the appropriate amortization of the DARR deferral balance have not been proposed nor determined at this point. Finally, Creative Energy does not support a one-time surcharge arguing that it would be impractical and problematic.²⁰⁵

¹⁹⁹ Creative Energy Final Argument, p. 29.

²⁰⁰ Exhibit B-1, Appendix C, Section 1.1

²⁰¹ Exhibit B-4, BCUC IR 27.3.1. The total associated steam load is estimated to have been 64,625 thousand pounds of steam compared with the metered steam load of 33,947 thousand pounds of steam.

²⁰² Creative Energy Final Argument, p. 30

²⁰³ Exhibit B-9, BCUC IR 59.2; Creative Energy Final Argument, p. 31.

²⁰⁴ Exhibit B-9, BCUC IR 59.2

²⁰⁵ Exhibit B-9, BCUC IR 59.2.

Position of the Parties

The CEC states that it is satisfied that the PARQ meter issues have been resolved and accepts Creative Energy's proposal to address the matter prospectively in 2020.²⁰⁶

Panel Determination

The question before the Panel is whether the additional steam supplied by the Core Steam system to the NEFC should be recognised, and, if so, how and when should it be recognised.

In subsection 4.1, the Panel approves Creative Energy's request for permanent approval of NEFC's 2019 interim Hot Water Rates.

The Panel previously determined in subsection 4.1 that the Variance Deferral Account continues to be approved for NEFC, and variances in the Steam Service Rates and Fuel Cost charges and other forecast items are captured in this deferral account for 2019.²⁰⁷ Given this framework, the variances between forecast versus actual Steam Service Rates and Fuel Cost charged to NEFC, including the underreported volumes of steam delivered to the PARQ hot water plant would flow into the Variance Deferral Account for 2019.

Regarding the underreporting of the PARQ Hot Water Plant steam meter, the Panel notes it is Creative Energy's preference that NEFC's customers not be charged for the amount related to underreported steam services of approximately \$187,807 provided by the Core Steam system due to the 'controversial impact' of a large one-time surcharge on customers. The Panel acknowledges these concerns and finds this approach is reasonable for the underreported steam service.

However, in stating its preference that the correction be made on a go-forward basis, Creative Energy does not address the fair allocation and recovery of fuel costs among other Core Steam System and NEFC customers. With respect to fuel costs, the Panel notes that revenues and expenses related to fuel costs for the Core Steam System are not included in its revenue requirements and Steam System Rates. For Core Steam customers, fuel costs are recovered through the FCAC and variances related to the difference between the forecast and actual fuel costs are allocated to the FCSA and recovered from customers through the FCAC Rate Rider. Since NEFC is also a customer of the Core Steam system, this mechanism also applies.

Therefore, it follows that, if the underreported PARQ volumes were adjusted, the additional fuel cost charges payable by NEFC would flow into the FCSA. Based on the proportion of steam load attributed to NEFC and Core customers, the Panel observes that if the fuel cost portion is not collected from NEFC customers and the FCSA balance is left unchanged, Core customers will have overpaid for their proportionate share of the FCSA balance. Creative Energy estimates that this underpayment totals \$497,728. The Panel does not agree this result is appropriate and considers that recovery of these costs from NEFC through the FCAC mechanism is necessary.

For the above-stated reasons, Creative Energy is directed to account for the NEFC underreported fuel cost charges of \$497,728 due to the PARQ metering issue in the Core Steam System's FCSA. Recording the amount in the FCSA is appropriate since this account records the differences between the forecast and actual fuel costs and the adjustment resulting in Core customers ultimately paying their proportionate share of fuel costs.

Creative Energy is also directed to record the additional fuel costs in the NEFC Variance Deferral Account, consistent with other fuel cost variances.

²⁰⁶ CEC Final Argument, p. 37

²⁰⁷ Creative Energy 2016-2017 RRA Decision, p. 70.

As discussed in subsection 4.1, **Creative Energy is directed to include in its Compliance Filing the detailed** calculations supporting the 2019 Variance Deferral Account balance, including the details of the amount attributed to the PARQ metering issue.

5.0 Setting 2020 Rates for NEFC

5.1 2020 Hot Water Rates

Proposed Hot Water Rate Increase of 3.7 Percent

As outlined in section 4.0 above, the 2016-2017 Core Steam System and NEFC Decision approved a recovery period for the RDDA, commencing in 2020 and ending in 2030.²⁰⁸ Further, the hot water load forecast that formed the basis of Creative Energy's NEFC CPCN Application in 2015 and its RRA in 2016 assumed full buildout in the NEFC neighbourhood by 2025, with a total connected floor area of 506,300 square meters and total hot water demand of 48,100 MWh.

Creative Energy explains that it has currently a total connected floor area of 162,481 square meters and hot water demand of 19,162 MWh forecast for 2020.²⁰⁹ Creative Energy also states that the City of Vancouver (the City) has now extended its connection bylaw to include the future development in the NEFC neighbourhood, meaning that the City will provide service to future developments in the NEFC, rather than Creative Energy. Creative Energy states that while it intends to supply the hot water to serve the City's loads, the necessary arrangements with the City have not been made yet.

As a result of these developments, Creative Energy states that it does not have a consolidated forecast of load growth in the NEFC neighborhood and that the timing of required incremental capacity investments to support that load growth is uncertain. Therefore, Creative Energy submits that there is not a sufficient basis to determine longer-term levelized increases to NEFC rates that recover the RDDA over a reasonable time frame.²¹⁰

In the absence of the above information, Creative Energy is requesting approval to increase the 2020 NEFC rates over the rates currently in effect by 3.7 percent. It states that this is equivalent to a one cent increase in the fixed charge per square meter per month, an approximate \$2/MWh increase in the variable charge, and would result in the following rates:²¹¹

- Fixed rate of \$0.28 per square meter per month; and
- Variable Rate of \$54.03/MWh.

Below is a summary of Creative Energy's proposed 2020 revenue requirement reflecting the proposed rate increase of 3.7 percent:

²⁰⁸ 2016-2017 Core Steam System and NEFC Decision, p. 69.

²⁰⁹ Exhibit B-1, p. 9.

²¹⁰ Ibid.

²¹¹ Exhibit B-1, p. 11.

C	2019	2020		
Component	Actuals	Test Year		
Fuel	550,767	1,061,666		
Steam Tariff	257,991	536,051		
Cost of Sales	808,758	1,597,717		
Operation & Maintenance Expense	117,406	161,000		
Municipal Taxes	18,097	19,765		
Other				
Total Operating and Maintenance Expense	944,260	1,778,481		
Income Tax Expense	6,148	19,445		
Depreciation Expense	191,400	216,000		
Actual/Proposed Return on Rate Base	315,000	317,000		
Cost of Services	1,456,808	2,330,926		
Total Revenue Requirement	1,456,808	2,330,926		
Hot Water Sales Revenues	1,447,745	1,581,195		
Total Revenues	1,447,745	1,581,195		
Revenue (Surplus)/Deficiency	9,064	749,731		
FCAC Rate Rider Recovery \$16.15 per MWh		-309,360		
Addition (subtraction) to (from) RDDA	9,064	440,371		

Table 24 – Proposed 2020 Revenue Requirement²¹²

In addition to the 3.7 percent rate increase outlined above, Creative Energy is also proposing a FCAC Rate Rider related to a specific portion of fuel cost included in the table above charged directly to NEFC customers, effective January 1, 2020, and discussed in subsection 5.2 below. Creative Energy estimates the average expected bill impact of the NEFC FCAC Rate Rider in 2020 is 20 percent.²¹³ When taken in addition to the 3.7 percent rate increase outlined above, the total resulting impact is an expected bill increase of 23.7 percent.²¹⁴

Creative Energy proposes an addition of the \$440,371 revenue deficiency net of the proposed FCAC Rate Rider recovery to the RDDA balance, plus other adjustments of the addition of interest of \$25,944, and transfer of amortization from the Variance Deferral Account of \$34,539.²¹⁵

Consistent with the proposed approach for 2019, Creative Energy's proposal does not contemplate any addition to the Variance Deferral Account, but instead proposes the addition of interest of \$2,169 and the transfer of amortization of \$34,539 to the RDDA.²¹⁶

Creative Energy explains that the increase of 3.7 percent is estimated as a reasonable minimum required based on an indicative estimate of inflation of 2 percent per year in steam, fuel and operating costs and continued recovery of and on the infrastructure capital invested that would target recovery of the RDDA in 15 to 20 years.²¹⁷ Creative Energy also states that the increase is not based directly on the percentage changes in yearover-year O&M costs, but rather it appears reasonable overall in the context of an otherwise increasing balance in the RDDA, and other factors.²¹⁸ Creative Energy also submits that the request for the 3.7 percent increase is aligned with the BCUC's direction to commence RDDA recovery in 2020, and lowers the rate of addition to the RDDA that would otherwise occur absent a rate increase in 2020.²¹⁹

²¹² Exhibit B-9, Schedules for RRA Filing – IR Round 2, NEFC Schedule 1.

²¹³ Exhibit B-4, BCUC IR 58.1

²¹⁴ 20 percent plus 3.7 percent equals 23.7 percent.

²¹⁵ Exhibit B-9, Schedules for RRA Filing – IR Round 2, NEFC Schedule 12.

²¹⁶ Exhibit B-9, Schedules for RRA Filing – IR Round 2, NEFC Schedule 12.

²¹⁷ Exhibit B-1, p. 11.

²¹⁸ Exhibit B-9, BCUC IR 58.3.

²¹⁹ Creative Energy Final Argument, p. 26.

To clear the RDDA balance by 2030, as contemplated under the original model, Creative Energy states that a rate increase of 5 percent per year over that time frame would be needed (assuming no load growth from additional development due to the uncertainty surrounding development).²²⁰ Creative Energy also submits that due to the discussions with the City of Vancouver, it is premature at this time to advance a comprehensive proposal for the full recovery of the RDDA by 2030. It confirms that it is currently reviewing the NEFC rate design, and plans to bring forward a comprehensive proposal to support its next RRA for 2021, at which time the allocation and recovery of costs to serve the NEFC and the amounts that get added to the RDDA will be revisited.²²¹

The average expected customer bill impact in dollar terms for the proposed rate 3.7 percent increase is shown in Table 25 below.²²²

Incremental			Rate Increase
Annual Bill Impact	Variable Rate	Fixed Rate	Impact
(\$)	component	component	(3.7%)
Customer 1	2,708	1,657	4,365
Customer 2	16,960	8,622	25,582
Customer 3	8,193	4,364	12,556
Customer 4	9,078	4,835	13,913
Sum	36,939	19,478	56,417
Average	9,235	4,870	14,104

Table 25 – Average Expected Bill Impact for Proposed NEFC Rate Increase

Forecast Hot Water Load

For 2020, Creative Energy forecast a hot water load of 19,162 MWh.²²³ The hot water load is an essential billing determinant for NEFC rate setting since all costs are recovered through the volume of hot water sold to customers. In addition, the load forecast is used for other ratemaking purposes, for example, forecasting the fuel costs, forecasting steam tariff costs, and implementing the load variance component of the Variance Deferral Account.

Creative Energy calculates the 2020 hot water load forecast by taking the average of 2018 and 2019 consumption in MWh for existing customers and adding load for the addition of the ARC building which came on in late 2019. The additional load for the ARC building is calculated based on the ratio of the floor area of the ARC building to an existing, comparable customer.²²⁴

Forecast O&M Expenses

For the NEFC system, Creative Energy forecasts O&M Expenses of \$161,000 for 2020 compared to 2019 actuals of \$117,406.²²⁵ Creative Energy explains that the largest increase in O&M costs relates to the elimination of a 2019 net insurance recovery of \$46,894 for expenses incurred in 2018.

Other than a portion of Insurance costs which are directly charged to NEFC as they relate specifically to the NEFC equipment, \$106,309 of the 2020 Forecast O&M costs for the NEFC system are allocated to the NEFC using the Massachusetts formula. For 2020, the allocation is based on the proposed two-factor Massachusetts formula

²²⁰ Exhibit B-1, p. 11.

²²¹ Creative Energy Final Argument, p. 26.

²²² Exhibit B-9, BCUC IR 58.1

²²³ Exhibit B-1, p. 21

²²⁴ Exhibit B-1, p. 43

²²⁵ Exhibit B-4, BCUC IR 1.1; Exhibit B-9, Schedules for RRA Filing – IR Round 2, NEFC Schedule 14.

methodology.²²⁶ Creative Energy's proposals regarding the Massachusetts formula are addressed by the Panel earlier in subsection 3.1 of this decision.

Positions of the Parties

In general, the CEC submits that a 3.7 percent rate increase is acceptable in the context of having stable rates for the last three years.²²⁷ The CEC also states that it has no issues with respect to the proposed net additions to the RDDA in 2020 of \$440,370, subject to any cost changes arising from its submissions on the Massachusetts formula.²²⁸

The CEC submits that Creative Energy uses a reasonable methodology for the NEFC load forecast and does not object to Creative Energy's load forecast for the NEFC.²²⁹

The CEC does not comment on the 2020 Forecast O&M expenses for the NEFC other than as noted in subsection 3.1 with respect to Creative Energy's proposals regarding the Massachusetts Formula.²³⁰

Panel Determination

The 2016-2017 Core Steam System and NEFC Decision approved a recovery period for the RDDA, commencing in 2020 and ending in 2030. After reflecting the adjustments to the RDDA directed in this decision, the balance in the RDDA now exceeds approximately \$1.4 million. The Panel acknowledges Creative Energy's submission that it currently does not have sufficient information to determine the longer-term levelized rate increases needed to recover the RDDA over a reasonable time frame. The Panel also notes the current hot water demand is approximately 40 percent of the hot water load forecast that formed the basis of Creative Energy's NEFC CPCN Application in 2015 and its RRA in 2016 which could indicate the current levelized rate may not be sufficient to recover capital costs incurred to build the system.

Creative Energy confirms that it is currently reviewing the NEFC rate design and plans to bring forward a comprehensive proposal in 2021. Given the uncertainties outlined, the Panel accepts deferring consideration of the recovery of the RDDA until 2021 to allow Creative Energy time to complete its comprehensive proposal. Creative Energy is directed to file a proposal for how rates should be set for 2021 by October 31, 2020. Creative Energy is also directed to file a comprehensive proposal for an NEFC rate design and for setting 2022 rates by June 30, 2021. Among other things, this application must address adjustments to the levelized rate necessary to reflect an updated hot water load forecast as well as the recovery mechanisms for the RDDA and Variance Deferral Account.

Until a comprehensive rate design application is submitted, the Panel finds it appropriate to maintain the existing rate structure. In line with the Panel's determinations in section 4.0 of this Decision and consistent with previous directives, **Creative Energy is directed to add any 2020 forecast variances related to the Variance Deferral Account, as follows:**

- Revenue shortfalls based on differences between actual revenues received and approved forecasts;
- Variances between forecast versus actual Steam Service Rates and Fuel Cost charged to NEFC;
- Variances between forecast versus actual Distribution expenses; and
- Variances between forecast versus actual Income Tax expense.

²²⁶ Exhibit B-1, p. 46; Exhibit B-4, BCUC IR 31.9.

²²⁷ CEC Final Argument, p. 34.

²²⁸ CEC Final Argument, p. 39.

²²⁹ CEC Final Argument, p. 37.

²³⁰ CEC Final Argument, p. 35.

In addition, the Panel approves the addition of variances between the approved revenue requirements and forecast revenues calculated based on approved rates to the RDDA. Further, the Panel extends approval of the Variance Deferral Account (which otherwise ends on December 31, 2020).

The Panel approves the forecast revenue requirement of \$2,330,926 for 2020 subject to the adjustments directed below. Based on its review and noting that no issues were raised by CEC, the Panel finds the forecast revenue requirement reasonable. As previously discussed in subsection 3.1, the Panel accepts Creative Energy's methodology for its steam load forecast, which by way of extension includes the hot water load. The Panel finds that Creative Energy's methodology is reasonable considering the limited historical data and lack of alternatives. However, the Panel expects that Creative Energy will continue to refine its load forecast as more data is available. Therefore, Creative Energy's request to use a hot water load of 19,162 MWh for revenue requirement purposes is approved.

The Panel also approves the NEFC 2020 Forecast O&M expenses, subject to the recalculation of costs which are allocated to the NEFC using the Massachusetts Formula as approved in subsection 3.1 of this decision. In subsection 3.1, the Panel rejected Creative Energy's proposal for a two-factor Massachusetts formula methodology. The Panel otherwise finds the 2020 Forecast O&M expenses to be reasonable.

Regarding Creative Energy's request for a 3.7 percent rate increase, the Panel agrees with Creative Energy and CEC that a rate increase is reasonable in the current context. Rates have remained at the same level since 2017 and there was an indication that they would increase in 2020 to start to recover the RDDA. However, given the overall rate impact of a 23.7 percent²³¹ increase when including the impact of the FCAC rate rider, and given the Panel's rejection of the FCAC rate rider as set out below, the Panel considers 10 percent to be a more appropriate rate increase for 2020. This level of increase will reduce the addition to the RDDA while keeping the rate below the level typically associated with rate shock. **Accordingly, the Panel approves a 10 percent rate increase for NEFC for 2020.**

5.2 Proposed Fuel Cost Adjustment Charge Rate Rider

Background

The NEFC System is a customer of the Core Steam system and fuel costs are charged to NEFC at the same rate as other customers of the Core Steam system.

Creative Energy's Core Steam service area recovers its fuel expenses from its customers using two methods: i) a Fuel Cost Adjustment Charge (FCAC); and ii) an FCAC Rate Rider.

The FCAC is approved by the BCUC and is based on forecast annual fuel costs divided by forecast annual load. Positive or negative variances between forecast fuel costs and actual fuel costs are captured in the FCSA. The FCAC Rate Rider distributes or collects from customers, as applicable, the positive or negative balances in the FCSA. The threshold for the application of a rate rider is when balances in the FCSA exceed plus or minus 5 percent of the most recently approved 12-month forecast of fuel costs. Creative Energy reports to the BCUC the balance of the FCSA quarterly.

The FCAC and the mechanism to recover excess balances in the FCSA were first approved by the BCUC by Order G-167-16 and accompanying Decision. Although the mechanism was approved in late 2016, the first FCAC Rate Rider was not approved until March 2019, at which time the balance in the FCSA exceeded the threshold criteria.

²³¹ 20 percent from the FCAC rate rider plus 3.7 percent.

The current FCAC of \$10.70 per thousand pounds of steam and FCAC Rate Rider of \$4.40 per thousand pounds of steam, were approved by the BCUC, effective November 1, 2019.²³²

Although Core Steam System and NEFC customers are charged the same rate, how the fuel costs are recovered are distinct. This process can be generally described as follows:

- NEFC customers' fuel costs are included within the revenue requirements and are recovered through hot water rates.
- Core Steam System customers' fuel costs are excluded from the revenue requirements and instead, the FCAC and FCAC Rate Rider are established in a separate process as described above and are recovered through a separate charge on Core Steam System customer bills.

2020 NEFC Fuel Costs and FCAC Rate Rider Proposal

For 2020, Creative Energy requests approval to include in the NEFC revenue requirement \$1,061,666 of fuel cost calculated based on the currently approved FCAC and a forecast hot water load.²³³ In addition to the FCAC-related fuel costs, Creative Energy request approval to charge an FCAC Rate Rider directly to NEFC customers, effective January 1, 2020, to recover from these customers the amounts the BCUC approved to be recovered from Core Steam customers though the FCAC Rate Rider approved by Order G-226-19.

The proposed FCAC Rate Rider of \$16.15/MWh is determined based on the NEFC load forecast and the currently approved FCAC Rate Rider and is calculated as follows:

FCAC Rate Rider \$4.40/M# * 2020 forecast NEFC steam load transfer 70,309 M# 2020 Forecast hot water load of 19,162 MWh

Creative Energy estimates the average expected bill impact of the NEFC FCAC Rate Rider in 2020 is 20 percent. Creative Energy provides further details of the customer impacts as follows:²³⁴

Incremental Annual Bill Impact (\$)	FCAC Rate Rider Impact (20%)
Customer 1	\$22,683
Customer 2	\$142,038
Customer 3	\$68,614
Customer 4	\$76,024
Sum	\$309,359
Average	\$77,340

Creative Energy notes that the 20 percent expected bill impact to NEFC is approximately the same expected bill impact for Core Steam customers under the \$4.40 per thousand pounds of steam FCAC Rate Rider currently in effect.²³⁵

Creative Energy also states that the impact in 2020 of the implementation of the direct charging of the FCAC Rate Rider to NEFC customers is a reduction of \$309,360 to the amount that would otherwise be proposed to be added to the RDDA.²³⁶ Creative Energy states that adding the amount to the RDDA would be contrary to the

²³² Order G-259-19.

²³³ Exhibit B-1-1, Evidentiary Update, Table 5 Update, p. 4.

²³⁴ Exhibit B-4, BCUC IR 58.1

²³⁵ Exhibit B-4, BCUC IR 28.1

²³⁶ Exhibit B-1, p. 12.

BCUC's findings in the Order G-226-19 Decision in regard to the appropriate time period for recovery of the excess fuel costs incurred to provide service in 2018/19.

Position of the Parties

The position of the CEC regarding the request of a direct charge for the FCAC Rate Rider to NEFC customer is unclear. CEC states that it "does not object to the addition of the Fuel Cost Adjustment Charge to the RDDA,"²³⁷ but does not indicate its view on whether a direct charge (which would mean the amount is not added to the RDDA) is appropriate.

Panel Determination

In setting Hot Water Rates for 2019 above, the Panel determines that fuel cost variances should continue to be recorded in the Variance Deferral Account. In subsection 5.1, the Panel also finds it appropriate to maintain the existing rate structure until a comprehensive rate design application is submitted for 2021 Hot Water Rates and approves Creative Energy's 2020 revenue requirement. The approved revenue requirement includes a total fuel cost forecast of \$1,061,666, of which \$309,359 relate to the Creative Energy FCAC Rate Rider to be paid by NEFC.

The Panel notes that Creative Energy's request for an NEFC FCAC Rate Rider payable by NEFC customers for a specific portion of fuel costs is inconsistent with the approved treatment for fuel costs and fuel costs variances in the existing rate structure and with the Panel's determinations and findings noted above. In the Panel's view, it would be more appropriate to make changes to the treatment of fuel costs as part of a comprehensive review of rate design.

The Panel acknowledges that the rate increase related to the proposed NEFC FCAC Rate Rider is similar in magnitude to the increase experienced by Core Steam customers. However, the rate design for Core Steam Rates and NEFC Hot Water Rates are dissimilar. NEFC rates are predicated on a levelized rate mechanism that is designed to record the impact of timing differences between the costs incurred to install the required infrastructure and the revenues from a buildout of customer load over time.

Further, fuel costs are embedded in the NEFC revenue requirements and thus are included in this levelized rate mechanism. As discussed above, NEFC also has a mechanism through the Variance Deferral Account to capture the difference between the forecast fuel costs, as included in the revenue requirements, and the actual fuel cost charged to the NEFC.

In the Panel's view, Creative Energy has not provided sufficient evidence, in the absence of a full comprehensive rate design, to demonstrate that a change to its treatment of fuel cost mechanism is necessary or desirable, especially in light of the expected 20 percent bill impact resulting from that change.²³⁸ Accordingly, the Panel denies Creative Energy's request for an NEFC FCAC Rate Rider. Creative Energy is directed to provide, in its Compliance Filing, the calculation of and proposed timing, by which customers will be refunded for the interim amounts collected in 2020, plus interest, by way of bill credits.

6.0 Additional BCUC Directives and Recommendations

6.1 Historical Pension Accounting Discrepancies

2015-2017 Pension Expense Deferral Account Variances

During the evidentiary phase of the proceeding, Creative Energy stated that it did not review any of the entries made with respect to the Pension Expense Deferral Account for the 2015-2017 period when it prepared the

²³⁷ CEC Final Argument, p. 36.

²³⁸ Exhibit B-4, BCUC IR 58.1

Application because those entries were prepared by previous management. However, based on current management's subsequent review of these entries, Creative Energy states:²³⁹

... it appears that no additions were made to the [Pension Expense] deferral [account] in Schedule 12 [Non-Rate Base Deferred Expenses] for 2015... Creative Energy agrees that the variance for 2015 should have been added to the deferral account balance.

A breakdown of the entry that should have been made with respect to deferral treatment of pension expenses for 2015 is shown in Table 26 below:

Description	Actual	Approved	Variance
Current Service Costs	\$219,400	\$214,300	\$5,100
Remeasurement (Gain)/Loss	\$95,200	\$0	\$95,200
Total	\$314,600	\$214,300	\$100,300

Table 26 – 2015 Pension Expense Variances²⁴⁰

In addition, Creative Energy stated that there is a difference of \$8,963 between what current management calculates as the total pension expense variance for the 2015 to 2017 period (\$100,300) and what previous management prepared (\$91,337). Creative Energy submitted that it appears that the difference relates to 2016 additions.²⁴¹ Table 27 below shows the difference between the two calculations for 2016:

Table 27 – 2016 Pension Expense Variances²⁴²

	New Mai	nagement Calc	ulation	Previous Management Calculation			
Description	Actual	Approved	Variance (A)	Actual	Approved	Variance (B)	
Current Service Costs	\$194,500	\$219,100	(\$24,600)	\$203,463	\$219,100	(\$15,637)	
Remeasurement (Gain)/Loss	(\$59,600)	\$0	(\$59,600)	(\$59,600)	\$0	(\$59,600)	
Total	\$134,900	\$219,100	(\$84,200)	\$143,863	\$219,100	(\$75,237)	
Difference (A-B)						(\$8,963)	

Position of the Parties

The CEC does not raise any issues regarding the Pension Expense Deferral Account.²⁴³

Panel Determination

While the Panel acknowledges that there have been discrepancies in pension calculations related to amounts determined in previous periods, the rates have already been set on a permanent basis for the years impacted.

²³⁹ Exhibit B-9, BCUC IR 54.6.

²⁴⁰ Exhibit B-9, BCUC IR 54.6.

²⁴¹ Exhibit B-9, BCUC IR 54.6.

 ²⁴² Exhibit B-9, BCUC IRs 54.6, 54.8; Creative Energy 2018-2022 Core Steam System Decision, p. 47. Current service cost amounts under the previous management's calculation in this table are calculated from the difference between total 2016 pension expense stated in the Creative Energy 2018-2022 Core Steam System Decision and the 2016 pension remeasurement gain stated in response to BCUC IR 54.8.
 ²⁴³ CEC Final Argument, p. 22.

The Panel finds that if adjustments were made to current or future rates for these amounts this would result in retroactive ratemaking, which as noted previously in this Decision is contrary to well-established regulatory principles. Accordingly, the Panel does not approve the 2015-2017 adjustments related to pension accounting discrepancies.

6.2 Panel Recommendations for Future Filings

After reviewing this Application, IR responses and various submissions, the Panel has identified several recommended focus areas for Creative Energy's future filings that should result in improvements to the efficiency and effectiveness of its proceedings. In connection with the focus areas identified below, the Panel also discusses a few examples of quality issues identified in this proceeding.

Adherence to Fundamental Regulatory Principles

In future filings, the BCUC would find it helpful if Creative Energy could explain how its proposals conform with fundamental ratemaking principles, the legislative framework and the approach taken by the BCUC and other comparable regulatory agencies in similar circumstances. Where applicable, Creative Energy should address how its proposals are consistent with the requirements to setting rates prospectively, the regulatory principle against retroactive ratemaking, cost causation principles, considerations regarding rate shock and other fundamental regulatory principles.

Following its review of this Application, the Panel has denied a number of Creative Energy's proposals that it finds would result in retroactive ratemaking, including proposals related to the Fuel Switch Study and LTRP Deferral Account²⁴⁴ and corrections related to 2015-2017 Pension Expense Deferral Account variances.²⁴⁵ Among other reasons for the Panel's denial of Creative Energy's proposal for the proposed Core Stream DARR mechanism, Creative Energy has not provided any evidence that such mechanism is an accepted means of recovering pension, regulatory and other recurring non-controllable O&M cost variances.²⁴⁶

Compliance with BCUC Orders and Directives

The Panel recommends Creative Energy develop a process and dedicate sufficient resources to ensure BCUC orders are tracked and addressed in a complete and timely manner. If clarification of an order is needed or a request for variance from an order is sought, it should be done in a thorough and transparent way.

The Panel notes instances in this proceeding where significant follow-up on previous BCUC orders was required, including:

- Not addressing the After-tax Regulatory Pension Asset Account as directed.²⁴⁷
- Not initially addressing the Fuel Switch Study and LTRP Deferral Account in the Application as directed;²⁴⁸
- Recording NEFC forecast variances that were directed to be included in the Variance Deferral Account in the RDDA without prior BCUC approval;²⁴⁹ and
- Not addressing the Panel's request for submissions on whether forecast deferral treatment to capture steam load variances for the Core Steam System may be warranted considering recent variances between forecasts and actuals.²⁵⁰

²⁴⁴ Subsection 3.3.

²⁴⁵ Subsection 6.1

²⁴⁶ Subsection 3.2.5.

²⁴⁷ Subsection 2.2.

²⁴⁸ Subsection 1.6.

²⁴⁹ Subsection 4.1.

Holistically and Comprehensively Considering Proposed Rate Design Changes

In the absence of a comprehensive rate design application, if Creative Energy proposes incremental rate design changes in future filings, it would be helpful to the BCUC and other parties if Creative Energy considers and communicates the regulatory justification for the proposal having considered cost causation and other rate design principles (eg., Bonbright) holistically, and taking into account the existing approved rate structure that the BCUC has already found to be not unjust, unreasonable, unduly discriminatory or unduly preferential.

In this Decision, the Panel finds Creative Energy has not provided sufficient justification for the proposed incremental rate design changes that fundamentally alter the rate treatment of certain Core Steam System O&M cost variances²⁵¹ and the treatment of NEFC fuel costs.²⁵² Furthermore, such rate design changes are not appropriate for consideration within the context of a revenue requirements application.

Ensuring the Accuracy, Completeness and Transparency of Accounting, Financial Schedules and Other Information

To ensure an effective and efficient review of an application, it is incumbent on Creative Energy to ensure the accuracy, completeness and transparency of the accounting, financial and other information included in an application.

The Panel notes instances in this proceeding where significant regulatory time and resources were necessary to follow-up on previous BCUC orders. These include:

- The calculation and analysis of accounts related to pension expenses;²⁵³
- The analysis of Fuel Switch Study and LTRP costs. The lack of transparency and clarity as to the nature and amount of the costs noted by the BCUC in its decision on the 2018-2022 RRA²⁵⁴ resulted in the BCUC declining to make any determination regarding their recoverability in that proceeding and instead directing that the costs claimed be put into Fuel Switch Study and LTRP Deferral Account pending the filing of an updated LTRP or RRA. While Creative Energy has been more forthcoming and has provided more clarity relating to these costs in this proceeding, the examination of the issue relating to the proper disposition of these costs has taken longer than warranted; and
- Corrections to capital additions related to the Remote Metering Project. During the proceeding and only in response to BCUC IRs, Creative Energy removed \$293,666 of construction in process amounts from rate base and updated related calculations related to that project.²⁵⁵

Timeliness of Filings

By ensuring the timeliness of filings, Creative Energy can assist the BCUC and other parties in the effective and efficient review of applications. Timely filings allow all parties the opportunity to consider proposals prospectively and to ensure a utility is held accountable for its utility operations.

In addition to the comments above, in making its determination on the 2019 Core Steam Rates, the Panel finds, among other things, an unreasonable amount of time passed between Creative Energy's filing for interim rates (December 14, 2018) and final rates (December 19, 2019).²⁵⁶

²⁵⁰ Subsection 1.6.

²⁵¹ Subsection 3.2.

²⁵² Subsection 5.2.

²⁵³ Subsection 6.1.

²⁵⁴ Creative Energy 2018-2022 Core Steam System Decision, p. 33.

²⁵⁵ Exhibit B-9, BCUC IR 52.2.

In consideration of the Panel's comments in subsection 2.2 that rates should be set prospectively based on forecasts, the Panel directs Creative Energy to file its next two-year revenue requirements application for the Core Steam System by November 1, 2020 in order for the BCUC to review the application before the start of those years. As directed in subsection 5.1, Creative Energy must file a proposal for how rates should be set for 2021 for the NEFC by October 31, 2020 and a comprehensive proposal for NEFC rate design and for setting 2022 rates by June 30, 2021.

7.0 Summary of Approvals and Directives

This summary is provided for the convenience of readers. In the event of any difference between the directions in this summary and those in the body of the decision, the wording in the decision shall prevail.

	Directive	Page
1.	For the reasons outlined above, the Panel approves the 2019 interim rates as	13
	permanent.	
2.	The Panel approves the continuation of the non-rate base Third Party Regulatory	13
	Cost Deferral Account for 2019.	
3.	Therefore, the Panel directs that the following forecast variance amounts be	14
	recognized as additions to the TPRCDA:	
	• 2018 forecast variance of \$207,659 plus interest; and	
	• 2019 forecast variance of \$177,340 plus interest.	
4.	For 2019, Creative Energy is directed to recognize amortization of 50 percent of	14
	the 2018 TPRCDA account balance or \$103,830, plus interest.	
5.	Accordingly, the Panel approves the continuation of the Pension Expense Deferral	14
	Account with a carrying cost equal to Creative Energy's short-term debt rate, until	
	otherwise determined by the BCUC. The Panel directs that the following forecast	
	variance amounts be recognized as additions to the Pension Expense Deferral	
	Account:	
	• The 2018 forecast variance of \$377,966 plus interest; and	
	• The 2019 forecast variance, based on the 2018 approved forecast, plus	
	interest, as set out below.	
	Table 7 – 2019 Pension Expense Variances	
	Description Amount	
	Current Service Costs \$ 56,232	
	Remeasurement Gain/(Loss) \$ 446,200	
	Total \$ 502,432	
6.	Therefore, Creative Energy is directed to record the following adjustments to the	15
	Pension Expense Deferral Account, effective 2019:	
	Table 8 – Offset of the 2018 and 2019 Remeasurement amounts	

²⁵⁶ Subsection 2.2.

	Description Amount	
	2018 Remeasurement Loss \$ (403,700)	
	2019 Remeasurement Gain \$ 446,200	
	Difference 2018 and 2019 \$ 42,500	
7.	Accordingly, Creative Energy is directed to amortize the 2018 current service costs	15
	variance of \$25,735 recorded in the Pension Expense Deferral Account, plus	
	interest, in 2019.	
8.	Creative Energy is directed to file information in the next RRA detailing how other	15
	BC and Canadian regulated utilities treat pension remeasurement gains and losses	
	for ratemaking purposes.	
9.	Creative Energy is directed to reduce the allowed return for 2019 by \$14,862	16
5.	based on a mid-year After-tax Regulatory Pension Asset Account balance of	10
	\$362,670.	
10.	Accordingly, the Panel makes no determination on the Remote Metering Project or	16
	any of the costs associated with this project.	
11.	The Panel approves the 2020 forecast revenue requirement of \$9,086,328 subject	24
	to the adjustments directed below.	
12.	The Panel approves the total 2020 steam load forecast of 1,140,634 thousand	24
	pounds of steam as requested by Creative Energy.	
13.	The Panel approves the inclusion of NEFC in the Massachusetts Formula cost	25
	allocation.	
14.	Accordingly, the Panel denies Creative Energy's request for a two-factor	25
	Massachusetts Formula and approves the use of a three-factor Massachusetts	
	Formula, beginning in 2020, based on the following factors: the average gross book	
	value of capital assets or property, plant and equipment; salaries or direct labour	
	expenses; and operating revenues.	
15.	Creative Energy is directed to recalculate the costs allocated to the Core Steam	25
	System and NEFC, respectively, based on the three-factor Massachusetts Formula	
	approved above in a compliance filing due to the BCUC within 30 days of the date	
16.	of this decision (Compliance Filing), including the rate impact for each service area.	25
10.	Therefore, the Panel approves the 2020 forecast for outside services costs of \$105,466 as requested by Creative Energy.	25
17.	The Panel approves the 2020 forecast steam production overtime expense.	25
17.	Creative Energy is directed to reduce the allowed return for 2020 by \$21,503	26
10.	based on a mid-year After-tax Regulatory Pension Asset Account balance of	20
	\$268,202.	
19.	The Panel approves the establishment of a non-rate base Water Cost Deferral	29
-	Account to capture water cost forecast variances for the 2020 test year only.	-
20.	Creative Energy is directed to address the continued need for the Water Cost	29
	Deferral Account and to re-evaluate its water cost forecast methodology in its next	
	revenue requirement application. Creative Energy must provide in that application	
	information comparing the forecast, approved and actual historical water	
	expenses and volumes to assist in the timely review of the application.	
21.	The Panel also approves Creative Energy's proposal for a one-year amortization	29
	period on the Water Cost Deferral Account commencing January 1, 2021, and a	
	carrying cost equal to Creative Energy's short-term debt rate, unless otherwise	
	determined by the BCUC.	
22.	The Panel approves the establishment of a non-rate base Property Tax Deferral	30
	Account to capture property tax forecast variances. The Panel also approves	

Cost Deferral Account for 2020 to capture third-party regulatory costs forecast variances. 24. Accordingly, for 2020, the Panel approves a one-year amortization period for the TPRCDA, and Creative Energy is directed to recognize amortization of \$281,170 plus interest for 2020. 25. The Panel approves the continuation of the non-rate base Pension Expense Deferral Account for 2020 to capture pension expense forecast variances. 3 26. Accordingly, for 2020, the Panel approves a one-year amortization period for the Pension Expense Deferral Account, and Creative Energy is directed to recognize amortization of \$98,732 plus interest. 3 27. For the reasons outlined above, the Panel denies Creative Energy's request for permanent approval of its proposed DARR mechanism. Creative Energy is directed to provide, in its Compliance Filing the calculation of and proposed timing by which customers will be refunded the amounts collected in 2020 on an interim basis, plus interest, by way of bill credits. Creative Energy is also directed to record the amortization of the Pension Expense Deferral Account and Third Party Regulatory Costs Deferral Accounts in the revenue requirement for 2020 Steam Rates. 28. Accordingly, the Panel denies Creative Energy's request for approval to recover from customers of the Core Steam System the \$214,185 spent in 2016, plus interest , along with Creative Energy's request to defer the mechanism and timing for such recovery for determination in the 2021 RRA. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account as Creative Energy proposes. 30. Accordingly, the Panel den	g January 1, 2021 and a carrying cost equal to debt rate on the mid-year value, unless otherwise31nuation of the non-rate base Third Party Regulatory t0 to capture third-party regulatory costs forecast31all approves a one-year amortization period for the is directed to recognize amortization of \$281,17031nuation of the non-rate base Pension Expense capture pension expense forecast variances.31rel approves a one-year amortization period for the is directed to recognize amortization period for the ount, and Creative Energy is directed to recognize interest.32ve, the Panel denies Creative Energy's request for oposed DARR mechanism. Creative Energy is directed Filing the calculation of and proposed timing by which he amounts collected in 2020 on an interim basis, plus . Creative Energy is also directed to record the txpense Deferral Account and Third Party Regulatory e revenue requirement for 2020 Steam Rates.40Creative Energy's request for approval to recover team System the \$214,185 spent in 2016, plus Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferral Account as40Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account ad to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the n be made. The Panel directs Creative Energy to close the41nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:41
Creative Energy's short-term debt rate on the mid-year value, unless otherwise determined by the BCUC. 23. The Panel approves the continuation of the non-rate base Third Party Regulatory Cost Deferral Account for 2020 to capture third-party regulatory costs forecast variances. 3 24. Accordingly, for 2020, the Panel approves a one-year amortization period for the TPRCDA, and Creative Energy is directed to recognize amortization of \$281,170 plus interest for 2020. 3 25. The Panel approves the continuation of the non-rate base Pension Expense Deferral Account for 2020 to capture pension expense forecast variances. 3 26. Accordingly, for 2020, the Panel approves a one-year amortization period for the Pension Expense Deferral Account, and Creative Energy is directed to recognize amortization of \$98,732 plus interest. 3 27. For the reasons outlined above, the Panel denies Creative Energy's request for permanent approval of its proposed DARR mechanism. Creative Energy is directed to provide, in its Compliance Filing the calculation of and proposed timing by which customers will be refunded the amounts collected in 2020 on an interim basis, plus interest, by way of bill credits. Creative Energy's request for approval to recover from customers of the Core Steam System the \$214,185 spent in 2016, plus interest, along with Creative Energy's request to defer the mechanism and timing for such recovery for determination in the 2021 RRA. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account. 4 29. [The Panel] directs Creative Energy to write	debt rate on the mid-year value, unless otherwise31nuation of the non-rate base Third Party Regulatory 10 to capture third-party regulatory costs forecast31nel approves a one-year amortization period for the is directed to recognize amortization of \$281,17031nuation of the non-rate base Pension Expense capture pension expense forecast variances. nel approves a one-year amortization period for the count, and Creative Energy is directed to recognize interest. <i>re</i> , the Panel denies Creative Energy's request for posed DARR mechanism. Creative Energy is directed Filing the calculation of and proposed timing by which ne amounts collected in 2020 on an interim basis, plus . Creative Energy is also directed to record the xpense Deferral Account and Third Party Regulatory e revenue requirement for 2020 Steam Rates.40Creative Energy's request for approval to recover team System the \$214,185 spent in 2016, plus Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferral Account as41Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account and to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write er and further directs Creative Energy to close the41nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:41
determined by the BCUC. 23. The Panel approves the continuation of the non-rate base Third Party Regulatory Cost Deferral Account for 2020 to capture third-party regulatory costs forecast variances. 3 24. Accordingly, for 2020, the Panel approves a one-year amortization period for the TPRCDA, and Creative Energy is directed to recognize amortization of \$281,170 plus interest for 2020. 3 25. The Panel approves the continuation of the non-rate base Pension Expense Deferral Account for 2020 to capture pension expense forecast variances. 3 26. Accordingly, for 2020, the Panel approves a one-year amortization period for the Pension Expense Deferral Account, and Creative Energy is directed to recognize amortization of \$98,732 plus interest. 3 27. For the reasons outlined above, the Panel denies Creative Energy's request for permanent approval of its proposed DARR mechanism. Creative Energy is directed to provide, in its Compliance Filing the calculation of and proposed timing by which customers will be refunded the amounts collected in 2020 on an interim basis, plus interest, by way of bill credits. Creative Energy's request for approval to recover from customers of the Core Steam System the \$214,185 spent in 2016, plus interest, along with Creative Energy's request for approval to recover from customers of the Core Steam System the \$214,185 spent in 2016, plus interest from the balance in the Fuel Switch Study and LTRP Deferral Account. 4 29. [The Panel] directs Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account as Creative Energy proposes.	nuation of the non-rate base Third Party Regulatory 10 to capture third-party regulatory costs forecast3110 to capture third-party regulatory costs forecast3111 rel approves a one-year amortization period for the is directed to recognize amortization of \$281,1703111 nuation of the non-rate base Pension Expense capture pension expense forecast variances.3112 nuation of the non-rate base Pension Expense capture pension expense forecast variances.3113 rel approves a one-year amortization period for the count, and Creative Energy is directed to recognize interest.3210 rel approves a one-year amortization period for the count, and Creative Energy's request for posed DARR mechanism. Creative Energy is directed Filing the calculation of and proposed timing by which he amounts collected in 2020 on an interim basis, plus . Creative Energy is also directed to record the xpense Deferral Account and Third Party Regulatory e revenue requirement for 2020 Steam Rates.4010 creative Energy's request for approval to recover team System the \$214,185 spent in 2016, plus Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferral Account as4011 creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account and to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write er and further directs Creative Energy to close the41nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, i
 The Panel approves the continuation of the non-rate base Third Party Regulatory Cost Deferral Account for 2020 to capture third-party regulatory costs forecast variances. Accordingly, for 2020, the Panel approves a one-year amortization period for the TPRCDA, and Creative Energy is directed to recognize amortization of \$281,170 plus interest for 2020. The Panel approves the continuation of the non-rate base Pension Expense Deferral Account for 2020 to capture pension expense forecast variances. Accordingly, for 2020, the Panel approves a one-year amortization period for the Pension Expense Deferral Account, and Creative Energy is directed to recognize amortization of \$98,732 plus interest. For the reasons outlined above, the Panel denies Creative Energy's request for permanent approval of its proposed DARR mechanism. Creative Energy is directed to provide, in its Compliance Filing the calculation of and proposed timing by which customers will be refunded the amounts collected in 2020 on an interim basis, plus interest, by way of bill credits. Creative Energy is also directed to recover the amortization of the Pension Expense Deferral Account and Third Party Regulatory Costs Deferral Accounts in the revenue requirement for 2020 Steam Rates. Accordingly, the Panel denies Creative Energy's request for approval to recover from customers of the Core Steam System the \$214,185 spent in 2016, plus interest , along with Creative Energy's request to defer the mechanism and timing for such recovery for determination in the 2021 RRA. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account. If Panel] directs Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account at to defer for consideration the disposition of such costs until such time as a CPCN for a low-carbon energy project is submitted and asses	20 to capture third-party regulatory costs forecast31hel approves a one-year amortization period for the is directed to recognize amortization of \$281,17031nuation of the non-rate base Pension Expense capture pension expense forecast variances.31hel approves a one-year amortization period for the roount, and Creative Energy is directed to recognize interest.32ve, the Panel denies Creative Energy's request for oposed DARR mechanism. Creative Energy is directed Filing the calculation of and proposed timing by which he amounts collected in 2020 on an interim basis, plus . Creative Energy is also directed to record the xpense Deferral Account and Third Party Regulatory e revenue requirement for 2020 Steam Rates.40Creative Energy's request for approval to recover team System the \$214,185 spent in 2016, plus Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferral40Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account as41nergy to write off the amount of \$103,536 plus he Fuel Switch Study and LTRP Deferral Account as41nergy to write off the amount of \$103,536 plus he fuel Switch Study and LTRP Deferral Account as41not submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write er and further directs Creative Energy to close the41nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:41
Cost Deferral Account for 2020 to capture third-party regulatory costs forecast variances.324.Accordingly, for 2020, the Panel approves a one-year amortization period for the TPRCDA, and Creative Energy is directed to recognize amortization of \$281,170 plus interest for 2020.325.The Panel approves the continuation of the non-rate base Pension Expense Deferral Account for 2020 to capture pension expense forecast variances.326.Accordingly, for 2020, the Panel approves a one-year amortization period for the Pension Expense Deferral Account, and Creative Energy is directed to recognize amortization of \$98,732 plus interest.327.For the reasons outlined above, the Panel denies Creative Energy's request for permanent approval of its proposed DARR mechanism. Creative Energy is directed to provide, in its Compliance Filing the calculation of and proposed timing by which customers will be refunded the amounts collected in 2020 on an interim basis, plus interest, by way of bill credits. Creative Energy is also directed to record the amortization of the Pension Expense Deferral Account and Third Party Regulatory Costs Deferral Accounts in the revenue requirement for 2020 Steam Rates.428.Accordingly, the Panel denies Creative Energy's request for approval to recover from customers of the Core Steam System the \$214,185 spent in 2016, plus interest, along with Creative Energy to write off the amount of \$103,536 plus interest from the balance in the Fuel Switch Study and LTRP Deferral Account and to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account and to defer for consideration the fuel Switch Study and LTRP Deferral Account and to defer for consideration the fuel Switch Study and LTRP Deferral Account to the accoun	20 to capture third-party regulatory costs forecast31hel approves a one-year amortization period for the is directed to recognize amortization of \$281,17031nuation of the non-rate base Pension Expense capture pension expense forecast variances.31hel approves a one-year amortization period for the roount, and Creative Energy is directed to recognize interest.32 <i>ve</i> , the Panel denies Creative Energy's request for popsed DARR mechanism. Creative Energy is directed Filing the calculation of and proposed timing by which he amounts collected in 2020 on an interim basis, plus . Creative Energy is also directed to record the expense Deferral Account and Third Party Regulatory e revenue requirement for 2020 Steam Rates.40Creative Energy's request for approval to recover team System the \$214,185 spent n2016, plus Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferral40Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account as41nergy to write off the amount of \$103,536 plus he Fuel Switch Study and LTRP Deferral Account as41no be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the41
variances.24.Accordingly, for 2020, the Panel approves a one-year amortization period for the TPRCDA, and Creative Energy is directed to recognize amortization of \$281,170 plus interest for 2020.325.The Panel approves the continuation of the non-rate base Pension Expense Deferral Account for 2020 to capture pension expense forecast variances.326.Accordingly, for 2020, the Panel approves a one-year amortization period for the Pension Expense Deferral Account, and Creative Energy is directed to recognize amortization of \$98,732 plus interest.327.For the reasons outlined above, the Panel denies Creative Energy's request for permanent approval of its proposed DARR mechanism. Creative Energy is directed to provide, in its Compliance Filing the calculation of and proposed timing by which customers will be refunded the amounts collected in 2020 on an interim basis, plus interest, by way of bill credits. Creative Energy is request for approval to recover from customers of the Core Steam System the \$214,185 spent in 2016, plus interest, along with Creative Energy's request to defer the mechanism and timing for such recovery for determination in the 2021 RRA. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account as Creative Energy propose.430.Accordingly, the Panel denies Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account and to defer for consideration the Huel Switch Study and LTRP Deferral Account at to defer for consideration the fuel Switch Study and LTRP Deferral Account to the saccount thereafter.430.Accordingly, the Panel denies Creative Energy's request to flow such of the ba	nel approves a one-year amortization period for the is directed to recognize amortization of \$281,17031nuation of the non-rate base Pension Expense capture pension expense forecast variances. nel approves a one-year amortization period for the sount, and Creative Energy is directed to recognize interest.32ve, the Panel denies Creative Energy's request for oposed DARR mechanism. Creative Energy is directed Filing the calculation of and proposed timing by which ne amounts collected in 2020 on an interim basis, plus . Creative Energy is also directed to record the xxpense Deferral Account and Third Party Regulatory e revenue requirement for 2020 Steam Rates.40Creative Energy's request for approval to recover team System the \$214,185 spent in 2016, plus Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferral Account as40Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account and to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the41nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:41
24. Accordingly, for 2020, the Panel approves a one-year amortization period for the TPRCDA, and Creative Energy is directed to recognize amortization of \$281,170 plus interest for 2020. 3 25. The Panel approves the continuation of the non-rate base Pension Expense Deferral Account, and Creative Energy is directed to recognize amortization period for the Pension Expense Deferral Account, and Creative Energy is directed to recognize amortization of \$98,732 plus interest. 3 26. Accordingly, for 2020, the Panel approves a one-year amortization period for the Pension Expense Deferral Account, and Creative Energy is directed to recognize amortization of \$98,732 plus interest. 3 27. For the reasons outlined above, the Panel denies Creative Energy is directed to provide, in its Compliance Filing the calculation of and proposed timing by which customers will be refunded the amounts collected in 2020 on an interim basis, plus interest, by way of bill credits. Creative Energy is also directed to recover the amortization of the Pension Expense Deferral Account and Third Party Regulatory Costs Deferral Accounts in the revenue requirement for 2020 Steam Rates. 28. Accordingly, the Panel denies Creative Energy's request for approval to recover for determination in the 2021 RRA. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account as Creative Energy proposes. 4 30. Accordingly, the Panel denies Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account as Creative Energy roposes. 4 30. <td< td=""><td>is directed to recognize amortization of \$281,170 nuation of the non-rate base Pension Expense capture pension expense forecast variances. nel approves a one-year amortization period for the count, and Creative Energy is directed to recognize interest. ve, the Panel denies Creative Energy's request for oposed DARR mechanism. Creative Energy is directed Filing the calculation of and proposed timing by which ne amounts collected in 2020 on an interim basis, plus . Creative Energy is also directed to record the fixpense Deferral Account and Third Party Regulatory e revenue requirement for 2020 Steam Rates. Creative Energy's request for approval to recover team System the \$214,185 spent in 2016, plus Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferral nergy to write off the amount of \$103,536 plus he Fuel Switch Study and LTRP Deferral Account as Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account and to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:</td></td<>	is directed to recognize amortization of \$281,170 nuation of the non-rate base Pension Expense capture pension expense forecast variances. nel approves a one-year amortization period for the count, and Creative Energy is directed to recognize interest. ve, the Panel denies Creative Energy's request for oposed DARR mechanism. Creative Energy is directed Filing the calculation of and proposed timing by which ne amounts collected in 2020 on an interim basis, plus . Creative Energy is also directed to record the fixpense Deferral Account and Third Party Regulatory e revenue requirement for 2020 Steam Rates. Creative Energy's request for approval to recover team System the \$214,185 spent in 2016, plus Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferral nergy to write off the amount of \$103,536 plus he Fuel Switch Study and LTRP Deferral Account as Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account and to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:
TPRCDA, and Creative Energy is directed to recognize amortization of \$281,170 plus interest for 2020.25.The Panel approves the continuation of the non-rate base Pension Expense Deferral Account for 2020 to capture pension expense forecast variances.326.Accordingly, for 2020, the Panel approves a one-year amortization period for the Pension Expense Deferral Account, and Creative Energy is directed to recognize amortization of \$98,732 plus interest.327.For the reasons outlined above, the Panel denies Creative Energy's request for permanent approval of its proposed DARR mechanism. Creative Energy is directed to provide, in its Compliance Filing the calculation of and proposed timing by which customers will be refunded the amounts collected in 2020 on an interim basis, plus interest, by way of bill credits. Creative Energy is also directed to record the amortization of the Pension Expense Deferral Account and Third Party Regulatory Costs Deferral Accounts in the revenue requirement for 2020 Steam Rates.28.Accordingly, the Panel denies Creative Energy's request for approval to recover from customers of the Core Steam System the \$214,185 spent in 2016, plus interest , along with Creative Energy's request to defer the mechanism and timing for such recovery for determination in the 2021 RRA. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account.429.[The Panel] directs Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account as Creative Energy proposes.430.Accordingly, the Panel denies Creative Energy's request to find a low-carbon energy project is submitted and assessment of how much of the<	is directed to recognize amortization of \$281,170 nuation of the non-rate base Pension Expense capture pension expense forecast variances. nel approves a one-year amortization period for the count, and Creative Energy is directed to recognize interest. <i>ve</i> , the Panel denies Creative Energy's request for oposed DARR mechanism. Creative Energy is directed Filing the calculation of and proposed timing by which ne amounts collected in 2020 on an interim basis, plus . Creative Energy is also directed to record the <i>x</i> pense Deferral Account and Third Party Regulatory e revenue requirement for 2020 Steam Rates. Creative Energy's request for approval to recover team System the \$214,185 spent in 2016, plus Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferral nergy to write off the amount of \$103,536 plus he Fuel Switch Study and LTRP Deferral Account as Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account as Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account as Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:
plus interest for 2020.25.The Panel approves the continuation of the non-rate base Pension Expense Deferral Account for 2020 to capture pension expense forecast variances.326.Accordingly, for 2020, the Panel approves a one-year amortization period for the Pension Expense Deferral Account, and Creative Energy is directed to recognize amortization of \$98,732 plus interest.327.For the reasons outlined above, the Panel denies Creative Energy is directed to provide, in its Compliance Filing the calculation of and proposed timing by which customers will be refunded the amounts collected in 2020 on an interim basis, plus interest, by way of bill credits. Creative Energy is also directed to record the amortization of the Pension Expense Deferral Account and Third Party Regulatory Costs Deferral Accounts in the revenue requirement for 2020 Steam Rates.428.Accordingly, the Panel denies Creative Energy is request for aron customers of the Core Steam System the \$214,185 spent in 2016, plus interest , along with Creative Energy is request to defer the mechanism and timing for such recovery for determination in the 2021 RRA. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account.429.[The Panel] directs Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account as Creative Energy proposes.430.Accordingly, the Panel denies Creative Energy's request to maintain the balance of the transition of such costs until such time as a CPCN for a low-carbon energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to write off the remain	nuation of the non-rate base Pension Expense capture pension expense forecast variances.31nuation of the non-rate base Pension Expense capture pension expense forecast variances.32nuation of the approves a one-year amortization period for the count, and Creative Energy is directed to recognize interest.32ve, the Panel denies Creative Energy's request for oposed DARR mechanism. Creative Energy is directed Filing the calculation of and proposed timing by which ne amounts collected in 2020 on an interim basis, plus . Creative Energy is also directed to record the expense Deferral Account and Third Party Regulatory e revenue requirement for 2020 Steam Rates.40Creative Energy's request for approval to recover team System the \$214,185 spent in 2016, plus Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferral Account as40Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account as41Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account at soposition of such costs until such time as a CPCN for a submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the41nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:41
25.The Panel approves the continuation of the non-rate base Pension Expense Deferral Account for 2020 to capture pension expense forecast variances.326.Accordingly, for 2020, the Panel approves a one-year amortization period for the Pension Expense Deferral Account, and Creative Energy is directed to recognize amortization of \$98,732 plus interest.327.For the reasons outlined above, the Panel denies Creative Energy's request for permanent approval of its proposed DARR mechanism. Creative Energy is directed to provide, in its Compliance Filing the calculation of and proposed timing by which customers will be refunded the amounts collected in 2020 on an interim basis, plus interest, by way of bill credits. Creative Energy is also directed to record the amortization of the Pension Expense Deferral Account and Third Party Regulatory Costs Deferral Accounts in the revenue requirement for 2020 Steam Rates.428.Accordingly, the Panel denies Creative Energy's request for approval to recover from customers of the Core Steam System the \$214,185 spent in 2016, plus interest , along with Creative Energy's request to defer the mechanism and timing for such recovery for determination in the 2021 RRA. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account.429.[The Panel] directs Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account and to defer for consideration the disposition of such costs until such time as a CPCN for a low-carbon energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Acco	capture pension expense forecast variances.nel approves a one-year amortization period for the sount, and Creative Energy is directed to recognize interest.32ve, the Panel denies Creative Energy's request for oposed DARR mechanism. Creative Energy is directed Filing the calculation of and proposed timing by which he amounts collected in 2020 on an interim basis, plus . Creative Energy is also directed to record the expense Deferral Account and Third Party Regulatory e revenue requirement for 2020 Steam Rates.40Creative Energy's request for approval to recover team System the \$214,185 spent in 2016, plus Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferral Account as40Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account as41Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account to asubmitted and assessment of how much of the n be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the41nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:41
Deferral Account for 2020 to capture pension expense forecast variances.26.Accordingly, for 2020, the Panel approves a one-year amortization period for the Pension Expense Deferral Account, and Creative Energy is directed to recognize amortization of \$98,732 plus interest.327.For the reasons outlined above, the Panel denies Creative Energy's request for permanent approval of its proposed DARR mechanism. Creative Energy is directed to provide, in its Compliance Filing the calculation of and proposed timing by which customers will be refunded the amounts collected in 2020 on an interim basis, plus interest, by way of bill credits. Creative Energy is also directed to record the amortization of the Pension Expense Deferral Account and Third Party Regulatory Costs Deferral Accounts in the revenue requirement for 2020 Steam Rates.428.Accordingly, the Panel denies Creative Energy's request for approval to recover from customers of the Core Steam System the \$214,185 spent in 2016, plus interest , along with Creative Energy's request to defer the mechanism and timing for such recover for determination in the 2021 RRA. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account.429.[The Panel] directs Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account and to defer for consideration the disposition of such costs until such time as a CPCN for a low-carbon energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Account to the account thereafter.431.Based on the Panel/determt	capture pension expense forecast variances.nel approves a one-year amortization period for the sount, and Creative Energy is directed to recognize interest.32ve, the Panel denies Creative Energy's request for oposed DARR mechanism. Creative Energy is directed Filing the calculation of and proposed timing by which he amounts collected in 2020 on an interim basis, plus . Creative Energy is also directed to record the expense Deferral Account and Third Party Regulatory e revenue requirement for 2020 Steam Rates.40Creative Energy's request for approval to recover team System the \$214,185 spent in 2016, plus Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferral40nergy to write off the amount of \$103,536 plus he Fuel Switch Study and LTRP Deferral Account as41Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account to asubmitted and assessment of how much of the n be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the41nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:41
 26. Accordingly, for 2020, the Panel approves a one-year amortization period for the Pension Expense Deferral Account, and Creative Energy is directed to recognize amortization of \$98,732 plus interest. 27. For the reasons outlined above, the Panel denies Creative Energy's request for permanent approval of its proposed DARR mechanism. Creative Energy is directed to provide, in its Compliance Filing the calculation of and proposed timing by which customers will be refunded the amounts collected in 2020 on an interim basis, plus interest, by way of bill credits. Creative Energy is also directed to record the amortization of the Pension Expense Deferral Account and Third Party Regulatory Costs Deferral Accounts in the revenue requirement for 2020 Steam Rates. 28. Accordingly, the Panel denies Creative Energy's request for approval to recover from customers of the Core Steam System the \$214,185 spent in 2016, plus interest , along with Creative Energy's request to defer the mechanism and timing for such recovery for determination in the 2021 RRA. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account as Creative Energy proposes. 30. Accordingly, the Panel denies Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account as Creative Energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to write off the account of the shareholder and further directs Creative Energy to close the deferral account the shareholder and further directs Creative Energy to close the deferral account thereafter. 31. Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel 	nel approves a one-year amortization period for the count, and Creative Energy is directed to recognize interest.32 <i>ye</i> , the Panel denies Creative Energy's request for opposed DARR mechanism. Creative Energy is directed Filing the calculation of and proposed timing by which he amounts collected in 2020 on an interim basis, plus . Creative Energy is also directed to record the expense Deferral Account and Third Party Regulatory e revenue requirement for 2020 Steam Rates.34Creative Energy's request for approval to recover team System the \$214,185 spent in 2016, plus Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferral40Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account and to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the41nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:41
Pension Expense Deferral Account, and Creative Energy is directed to recognize amortization of \$98,732 plus interest.27.For the reasons outlined above, the Panel denies Creative Energy's request for permanent approval of its proposed DARR mechanism. Creative Energy is directed to provide, in its Compliance Filing the calculation of and proposed timing by which customers will be refunded the amounts collected in 2020 on an interim basis, plus interest, by way of bill credits. Creative Energy is also directed to record the amortization of the Pension Expense Deferral Account and Third Party Regulatory Costs Deferral Accounts in the revenue requirement for 2020 Steam Rates.428.Accordingly, the Panel denies Creative Energy's request for approval to recover from customers of the Core Steam System the \$214,185 spent in 2016, plus interest , along with Creative Energy's request to defer the mechanism and timing for such recovery for determination in the 2021 RRA. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account.429.[The Panel] directs Creative Energy to write off the amount of \$103,536 plus interest from the balance in the Fuel Switch Study and LTRP Deferral Account as Creative Energy proposes.430.Accordingly, the Panel denies Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account and to defer for consideration the disposition of such costs until such time as a CPCN for a low-carbon energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Account to the account of the shareholder and further di	count, and Creative Energy is directed to recognize interest.34ve, the Panel denies Creative Energy's request for oposed DARR mechanism. Creative Energy is directed Filing the calculation of and proposed timing by which he amounts collected in 2020 on an interim basis, plus . Creative Energy is also directed to record the expense Deferral Account and Third Party Regulatory e revenue requirement for 2020 Steam Rates.34Creative Energy's request for approval to recover team System the \$214,185 spent in 2016, plus Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferral40Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account and to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write mations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:41
amortization of \$98,732 plus interest.27.For the reasons outlined above, the Panel denies Creative Energy's request for permanent approval of its proposed DARR mechanism. Creative Energy is directed to provide, in its Compliance Filing the calculation of and proposed timing by which customers will be refunded the amounts collected in 2020 on an interim basis, plus interest, by way of bill credits. Creative Energy is also directed to record the amortization of the Pension Expense Deferral Account and Third Party Regulatory Costs Deferral Accounts in the revenue requirement for 2020 Steam Rates.428.Accordingly, the Panel denies Creative Energy's request for approval to recover from customers of the Core Steam System the \$214,185 spent in 2016, plus interest , along with Creative Energy's request to defer the mechanism and timing for such recovery for determination in the 2021 RRA. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account.429.[The Panel] directs Creative Energy to write off the amount of \$103,536 plus interest from the balance in the Fuel Switch Study and LTRP Deferral Account as Creative Energy proposes.430.Accordingly, the Panel denies Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account and to defer for consideration the disposition of such costs until such time as a CPCN for a low-carbon energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Account to the account of the shareholder and further directs Creative Energy to close the deferral account thereafter.4<	interest.34ve, the Panel denies Creative Energy's request for opposed DARR mechanism. Creative Energy is directed Filing the calculation of and proposed timing by which he amounts collected in 2020 on an interim basis, plus . Creative Energy is also directed to record the expense Deferral Account and Third Party Regulatory e revenue requirement for 2020 Steam Rates.34Creative Energy's request for approval to recover team System the \$214,185 spent in 2016, plus Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferral40nergy to write off the amount of \$103,536 plus he Fuel Switch Study and LTRP Deferral Account as41Creative Energy's request to maintain the balance of submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the41nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:41
 27. For the reasons outlined above, the Panel denies Creative Energy's request for permanent approval of its proposed DARR mechanism. Creative Energy is directed to provide, in its Compliance Filing the calculation of and proposed timing by which customers will be refunded the amounts collected in 2020 on an interim basis, plus interest, by way of bill credits. Creative Energy is also directed to record the amortization of the Pension Expense Deferral Account and Third Party Regulatory Costs Deferral Accounts in the revenue requirement for 2020 Steam Rates. 28. Accordingly, the Panel denies Creative Energy's request for approval to recover from customers of the Core Steam System the \$214,185 spent in 2016, plus interest , along with Creative Energy's request to defer the mechanism and timing for such recovery for determination in the 2021 RRA. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account as Creative Energy proposes. 30. Accordingly, the Panel denies Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account as Creative Energy proposes. 30. Accordingly, the Panel denies Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account as low-carbon energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Account to the account of the shareholder and further directs Creative Energy to close the deferral account thereafter. 31. Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel 	<i>ve</i> , the Panel denies Creative Energy's request for poposed DARR mechanism. Creative Energy is directed Filing the calculation of and proposed timing by which he amounts collected in 2020 on an interim basis, plus . Creative Energy is also directed to record the expense Deferral Account and Third Party Regulatory e revenue requirement for 2020 Steam Rates.34Creative Energy's request for approval to recover team System the \$214,185 spent in 2016, plus Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferral40nergy to write off the amount of \$103,536 plus he Fuel Switch Study and LTRP Deferral Account as41Creative Energy's request to maintain the balance of submitted and assessment of how much of the m be made. The Panel directs Creative Energy to write er and further directs Creative Energy to close the41nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:41
permanent approval of its proposed DARR mechanism. Creative Energy is directed to provide, in its Compliance Filing the calculation of and proposed timing by which customers will be refunded the amounts collected in 2020 on an interim basis, plus interest, by way of bill credits. Creative Energy is also directed to record the amortization of the Pension Expense Deferral Account and Third Party Regulatory Costs Deferral Accounts in the revenue requirement for 2020 Steam Rates.28.Accordingly, the Panel denies Creative Energy's request for approval to recover from customers of the Core Steam System the \$214,185 spent in 2016, plus interest , along with Creative Energy's request to defer the mechanism and timing for such recovery for determination in the 2021 RRA. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account.429.[The Panel] directs Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account and to defer for consideration the disposition of such costs until such time as a CPCN for a low-carbon energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to close the deferral account thereafter.431.Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel4	oposed DARR mechanism. Creative Energy is directedFiling the calculation of and proposed timing by which he amounts collected in 2020 on an interim basis, plus a. Creative Energy is also directed to record the expense Deferral Account and Third Party Regulatory e revenue requirement for 2020 Steam Rates.Creative Energy's request for approval to recover team System the \$214,185 spent in 2016, plus Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferralnergy to write off the amount of \$103,536 plus he Fuel Switch Study and LTRP Deferral Account asCreative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account and to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close thenations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:
permanent approval of its proposed DARR mechanism. Creative Energy is directed to provide, in its Compliance Filing the calculation of and proposed timing by which customers will be refunded the amounts collected in 2020 on an interim basis, plus interest, by way of bill credits. Creative Energy is also directed to record the amortization of the Pension Expense Deferral Account and Third Party Regulatory Costs Deferral Accounts in the revenue requirement for 2020 Steam Rates.28.Accordingly, the Panel denies Creative Energy's request for approval to recover from customers of the Core Steam System the \$214,185 spent in 2016, plus interest , along with Creative Energy's request to defer the mechanism and timing for such recovery for determination in the 2021 RRA. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account.429.[The Panel] directs Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account and to defer for consideration the disposition of such costs until such time as a CPCN for a low-carbon energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to close the deferral account thereafter.431.Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel4	oposed DARR mechanism. Creative Energy is directedFiling the calculation of and proposed timing by which he amounts collected in 2020 on an interim basis, plus a. Creative Energy is also directed to record the expense Deferral Account and Third Party Regulatory e revenue requirement for 2020 Steam Rates.Creative Energy's request for approval to recover team System the \$214,185 spent in 2016, plus Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferralnergy to write off the amount of \$103,536 plus he Fuel Switch Study and LTRP Deferral Account asCreative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account and to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close thenations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:
 to provide, in its Compliance Filing the calculation of and proposed timing by which customers will be refunded the amounts collected in 2020 on an interim basis, plus interest, by way of bill credits. Creative Energy is also directed to record the amortization of the Pension Expense Deferral Account and Third Party Regulatory Costs Deferral Accounts in the revenue requirement for 2020 Steam Rates. 28. Accordingly, the Panel denies Creative Energy's request for approval to recover from customers of the Core Steam System the \$214,185 spent in 2016, plus interest , along with Creative Energy's request to defer the mechanism and timing for such recovery for determination in the 2021 RRA. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account as Creative Energy proposes. 30. Accordingly, the Panel denies Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account as Creative Energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Account to the account of the shareholder and further directs Creative Energy to close the deferral account thereafter. 31. Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel 	Filing the calculation of and proposed timing by which he amounts collected in 2020 on an interim basis, plus b. Creative Energy is also directed to record the expense Deferral Account and Third Party Regulatory e revenue requirement for 2020 Steam Rates.Creative Energy's request for approval to recover team System the \$214,185 spent in 2016, plus Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferral40nergy to write off the amount of \$103,536 plus he Fuel Switch Study and LTRP Deferral Account as40Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account and to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the41nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:41
customers will be refunded the amounts collected in 2020 on an interim basis, plus interest, by way of bill credits. Creative Energy is also directed to record the amortization of the Pension Expense Deferral Account and Third Party Regulatory Costs Deferral Accounts in the revenue requirement for 2020 Steam Rates.28.Accordingly, the Panel denies Creative Energy's request for approval to recover from customers of the Core Steam System the \$214,185 spent in 2016, plus interest , along with Creative Energy's request to defer the mechanism and timing for such recovery for determination in the 2021 RRA. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account.429.[The Panel] directs Creative Energy to write off the amount of \$103,536 plus interest from the balance in the Fuel Switch Study and LTRP Deferral Account as Creative Energy proposes.430.Accordingly, the Panel denies Creative Energy's request to maintain the balance of balance can be capitalized can be made. The Panel directs Creative Energy to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Account and to defer for consideration the disposition of such costs until such time as a CPCN for a low-carbon energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Account to the account of the shareholder and further directs Creative Energy to close the deferral account thereafter.431.Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel4	ne amounts collected in 2020 on an interim basis, plus. Creative Energy is also directed to record the expense Deferral Account and Third Party Regulatory e revenue requirement for 2020 Steam Rates.Creative Energy's request for approval to recover team System the \$214,185 spent in 2016, plus Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferralnergy to write off the amount of \$103,536 plus he Fuel Switch Study and LTRP Deferral Account asCreative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account and to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close thenations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:
 interest, by way of bill credits. Creative Energy is also directed to record the amortization of the Pension Expense Deferral Account and Third Party Regulatory Costs Deferral Accounts in the revenue requirement for 2020 Steam Rates. Accordingly, the Panel denies Creative Energy's request for approval to recover from customers of the Core Steam System the \$214,185 spent in 2016, plus interest , along with Creative Energy's request to defer the mechanism and timing for such recovery for determination in the 2021 RRA. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account. [The Panel] directs Creative Energy to write off the amount of \$103,536 plus interest from the balance in the Fuel Switch Study and LTRP Deferral Account as Creative Energy proposes. Accordingly, the Panel denies Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account and to defer for consideration the disposition of such costs until such time as a CPCN for a low-carbon energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Account to the account of the shareholder and further directs Creative Energy to close the deferral account thereafter. Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel 	Creative Energy is also directed to record the expense Deferral Account and Third Party Regulatory e revenue requirement for 2020 Steam Rates.40Creative Energy's request for approval to recover team System the \$214,185 spent in 2016, plus Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferral40nergy to write off the amount of \$103,536 plus he Fuel Switch Study and LTRP Deferral Account as40Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account and to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the41nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:41
amortization of the Pension Expense Deferral Account and Third Party Regulatory Costs Deferral Accounts in the revenue requirement for 2020 Steam Rates.28.Accordingly, the Panel denies Creative Energy's request for approval to recover from customers of the Core Steam System the \$214,185 spent in 2016, plus interest , along with Creative Energy's request to defer the mechanism and timing for such recovery for determination in the 2021 RRA. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account.429.[The Panel] directs Creative Energy to write off the amount of \$103,536 plus interest from the balance in the Fuel Switch Study and LTRP Deferral Account as Creative Energy proposes.430.Accordingly, the Panel denies Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account and to defer for consideration the disposition of such costs until such time as a CPCN for a low-carbon energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Account to the account of the shareholder and further directs Creative Energy to close the deferral account thereafter.431.Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel4	Expense Deferral Account and Third Party Regulatory e revenue requirement for 2020 Steam Rates.40Creative Energy's request for approval to recover team System the \$214,185 spent in 2016, plus Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferral40nergy to write off the amount of \$103,536 plus he Fuel Switch Study and LTRP Deferral Account as40Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account and to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the m be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the41nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:41
Costs Deferral Accounts in the revenue requirement for 2020 Steam Rates.28.Accordingly, the Panel denies Creative Energy's request for approval to recover from customers of the Core Steam System the \$214,185 spent in 2016, plus interest , along with Creative Energy's request to defer the mechanism and timing for such recovery for determination in the 2021 RRA. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account.29.[The Panel] directs Creative Energy to write off the amount of \$103,536 plus interest from the balance in the Fuel Switch Study and LTRP Deferral Account as Creative Energy proposes.430.Accordingly, the Panel denies Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account and to defer for consideration the disposition of such costs until such time as a CPCN for a low-carbon energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Account to the account of the shareholder and further directs Creative Energy to close the deferral account thereafter.431.Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel4	e revenue requirement for 2020 Steam Rates.40Creative Energy's request for approval to recover team System the \$214,185 spent in 2016, plus Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferral40nergy to write off the amount of \$103,536 plus he Fuel Switch Study and LTRP Deferral Account as40Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account and to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the m be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the41nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:41
 28. Accordingly, the Panel denies Creative Energy's request for approval to recover from customers of the Core Steam System the \$214,185 spent in 2016, plus interest, along with Creative Energy's request to defer the mechanism and timing for such recovery for determination in the 2021 RRA. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account. 29. [The Panel] directs Creative Energy to write off the amount of \$103,536 plus interest from the balance in the Fuel Switch Study and LTRP Deferral Account as Creative Energy proposes. 30. Accordingly, the Panel denies Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account and to defer for consideration the disposition of such costs until such time as a CPCN for a low-carbon energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Account to the account of the shareholder and further directs Creative Energy to close the deferral account thereafter. 31. Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel 	Creative Energy's request for approval to recover team System the \$214,185 spent in 2016, plus Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferral40nergy to write off the amount of \$103,536 plus he Fuel Switch Study and LTRP Deferral Account as40Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account and to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the41nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:41
 from customers of the Core Steam System the \$214,185 spent in 2016, plus interest , along with Creative Energy's request to defer the mechanism and timing for such recovery for determination in the 2021 RRA. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account. [The Panel] directs Creative Energy to write off the amount of \$103,536 plus interest from the balance in the Fuel Switch Study and LTRP Deferral Account as Creative Energy proposes. Accordingly, the Panel denies Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account and to defer for consideration the disposition of such costs until such time as a CPCN for a low-carbon energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Account to the account of the shareholder and further directs Creative Energy to close the deferral account thereafter. Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel 	team System the \$214,185 spent in 2016, plus Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferralnergy to write off the amount of \$103,536 plus he Fuel Switch Study and LTRP Deferral Account as40Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account and to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the41nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:41
 interest , along with Creative Energy's request to defer the mechanism and timing for such recovery for determination in the 2021 RRA. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account. [The Panel] directs Creative Energy to write off the amount of \$103,536 plus interest from the balance in the Fuel Switch Study and LTRP Deferral Account as Creative Energy proposes. Accordingly, the Panel denies Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account and to defer for consideration the disposition of such costs until such time as a CPCN for a low-carbon energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Account to the account of the shareholder and further directs Creative Energy to close the deferral account thereafter. Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel 	Energy's request to defer the mechanism and timing nation in the 2021 RRA. Creative Energy is directed to balance in the Fuel Switch Study and LTRP Deferralnergy to write off the amount of \$103,536 plus he Fuel Switch Study and LTRP Deferral Account as40Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account and to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the41nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:41
for such recovery for determination in the 2021 RRA. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account.29.[The Panel] directs Creative Energy to write off the amount of \$103,536 plus interest from the balance in the Fuel Switch Study and LTRP Deferral Account as Creative Energy proposes.430.Accordingly, the Panel denies Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account and to defer for consideration the disposition of such costs until such time as a CPCN for a 	nation in the 2021 RRA. Creative Energy is directed to e balance in the Fuel Switch Study and LTRP Deferralnergy to write off the amount of \$103,536 plus he Fuel Switch Study and LTRP Deferral Account as40Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account and to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the41nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:41
write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account.29.[The Panel] directs Creative Energy to write off the amount of \$103,536 plus interest from the balance in the Fuel Switch Study and LTRP Deferral Account as Creative Energy proposes.430.Accordingly, the Panel denies Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account and to defer for consideration the disposition of such costs until such time as a CPCN for a low-carbon energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Account to the account of the shareholder and further directs Creative Energy to close the deferral account thereafter.431.Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel4	e balance in the Fuel Switch Study and LTRP Deferralnergy to write off the amount of \$103,536 plus40he Fuel Switch Study and LTRP Deferral Account as40Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account and to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the41nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:41
Account.29.[The Panel] directs Creative Energy to write off the amount of \$103,536 plus interest from the balance in the Fuel Switch Study and LTRP Deferral Account as Creative Energy proposes.430.Accordingly, the Panel denies Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account and to defer for consideration the disposition of such costs until such time as a CPCN for a low-carbon energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Account to the account of the shareholder and further directs Creative Energy to close the deferral account thereafter.31.31.Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel4	nergy to write off the amount of \$103,536 plus40he Fuel Switch Study and LTRP Deferral Account as41Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account and to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the41nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:41
 29. [The Panel] directs Creative Energy to write off the amount of \$103,536 plus interest from the balance in the Fuel Switch Study and LTRP Deferral Account as Creative Energy proposes. 30. Accordingly, the Panel denies Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account and to defer for consideration the disposition of such costs until such time as a CPCN for a low-carbon energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Account to the account of the shareholder and further directs Creative Energy to close the deferral account thereafter. 31. Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel 	he Fuel Switch Study and LTRP Deferral Account as Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account and to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the in be made. The Panel directs Creative Energy to write in the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:
 interest from the balance in the Fuel Switch Study and LTRP Deferral Account as Creative Energy proposes. 30. Accordingly, the Panel denies Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account and to defer for consideration the disposition of such costs until such time as a CPCN for a low-carbon energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Account to the account of the shareholder and further directs Creative Energy to close the deferral account thereafter. 31. Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel 	he Fuel Switch Study and LTRP Deferral Account as Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account and to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the in be made. The Panel directs Creative Energy to write in the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:
Creative Energy proposes.430.Accordingly, the Panel denies Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account and to defer for consideration the disposition of such costs until such time as a CPCN for a low-carbon energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Account to the account of the shareholder and further directs Creative Energy to close the deferral account thereafter.31.31.Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel4	Creative Energy's request to maintain the balance of Fuel Switch Study and LTRP Deferral Account and to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:
 30. Accordingly, the Panel denies Creative Energy's request to maintain the balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account and to defer for consideration the disposition of such costs until such time as a CPCN for a low-carbon energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Account to the account of the shareholder and further directs Creative Energy to close the deferral account thereafter. 31. Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel 	Fuel Switch Study and LTRP Deferral Account and to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the41nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:41
 \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account and to defer for consideration the disposition of such costs until such time as a CPCN for a low-carbon energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Account to the account of the shareholder and further directs Creative Energy to close the deferral account thereafter. Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel 	Fuel Switch Study and LTRP Deferral Account and to sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the41nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:41
 defer for consideration the disposition of such costs until such time as a CPCN for a low-carbon energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Account to the account of the shareholder and further directs Creative Energy to close the deferral account thereafter. 31. Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel 	sposition of such costs until such time as a CPCN for a submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the nations above, the Panel denies Creative Energy's 41 ease in 2020 Steam Rates. Instead, the Panel sive of the impact of its decision above, including:
 low-carbon energy project is submitted and assessment of how much of the balance can be capitalized can be made. The Panel directs Creative Energy to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Account to the account of the shareholder and further directs Creative Energy to close the deferral account thereafter. Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel 	submitted and assessment of how much of the n be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the nations above, the Panel denies Creative Energy's ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:
 balance can be capitalized can be made. The Panel directs Creative Energy to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Account to the account of the shareholder and further directs Creative Energy to close the deferral account thereafter. Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel 	n be made. The Panel directs Creative Energy to write m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the nations above, the Panel denies Creative Energy's 41 ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:
off the remaining balance from the Fuel Switch Study and LTRP Deferral Account to the account of the shareholder and further directs Creative Energy to close the deferral account thereafter.31.Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel4	m the Fuel Switch Study and LTRP Deferral Account to er and further directs Creative Energy to close the nations above, the Panel denies Creative Energy's 41 ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:
the account of the shareholder and further directs Creative Energy to close the deferral account thereafter.31.Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel4	er and further directs Creative Energy to close the nations above, the Panel denies Creative Energy's 41 ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:
deferral account thereafter.31.Based on the Panel's determinations above, the Panel denies Creative Energy's4request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel4	nations above, the Panel denies Creative Energy's 41 ease in 2020 Steam Rates. Instead, the Panel usive of the impact of its decision above, including:
31.Based on the Panel's determinations above, the Panel denies Creative Energy's request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel4	ease in 2020 Steam Rates. Instead, the Panel Isive of the impact of its decision above, including:
request for a 4.2 percent increase in 2020 Steam Rates. Instead, the Panel	ease in 2020 Steam Rates. Instead, the Panel Isive of the impact of its decision above, including:
	isive of the impact of its decision above, including:
	·
approves a rate increase inclusive of the impact of its decision above, including:	te the costs allocated to the Core Steam System and
Directive to recalculate the costs allocated to the Core Steam System and	•
NEFC, respectively, based on the three-factor Massachusetts Formula	
approved above in its Compliance Filing and including the rate impact for	Compliance Filing and including the rate impact for
each service area;	
 Directive to reduce the allowed return for 2020 by \$21,503, based on a 	ne allowed return for 2020 by \$21,503, based on a
mid-year balance of \$268,202 in the After-tax Regulatory Pension Asset	
	268,202 in the After-tax Regulatory Pension Asset
Account.	5268,202 in the After-tax Regulatory Pension Asset
Account.Denial of the request for permanent approval of its proposed DARR	

	Directive to record the amortization of the Pension Expense Deferral	
	Account and Third Party Regulatory Costs Deferral Accounts in the revenue	
	requirement for 2020 Steam Rates.	
	Based on these adjustments, Creative Energy is directed to recalculate the rate	
	increase and include the updated financial schedules in its Compliance Filing.	
32.	The Panel approves Creative Energy's request for permanent approval of NEFC's	44
~~	2019 interim hot water rates.	
33.	Accordingly, the Panel approves the addition of a \$373,900 revenue deficiency plus	44
	interest to the RDDA for 2018, consistent with the amount determined by the	
	BCUC in 2017. For 2018, the Panel directs Creative Energy to file as part of its	
34.	Compliance Filing the amount to be added to the Variance Deferral Account.	44
54.	Accordingly, the Panel denies Creative Energy's request to transfer \$661,861 from the Variance Deferral Account to the RDDA in 2019.	44
35.	Therefore, the Panel denies Creative Energy's proposal to add the difference	44
55.	between the 2019 Actual revenue requirement and the revenues based on existing	
	rates to the existing RDDA.	
36.	Creative Energy is directed to recalculate the allocation of variances between the	44
50.	Variance Deferral Account and the RDDA based on the 2017 approved forecast. As	
	part of its Compliance Filing, Creative Energy is also directed to include an updated	
	schedule of the revised amounts in the Variance and Revenue Deficiency Deferral	
	Accounts, reflecting the above and other determinations in this Decision.	
37.	For the above-stated reasons, Creative Energy is directed to account for the NEFC	46
	underreported fuel cost charges of \$497,728 due to the PARQ metering issue in	
	the Core Steam System's FCSA.	
38.	Creative Energy is also directed to record the additional fuel costs in the NEFC	46
	Variance Deferral Account, consistent with other fuel cost variances.	
39.	Creative Energy is directed to include in its Compliance Filing the detailed	47
	calculations supporting the 2019 Variance Deferral Account balance, including the	
	details of the amount attributed to the PARQ metering issue.	
40.	Creative Energy is directed to file a proposal for how rates should be set for 2021	50
	by October 31, 2020. Creative Energy is also directed to file a comprehensive	
	proposal for an NEFC rate design and for setting 2022 rates by June 30, 2021.	
	Among other things, this application must address adjustments to the levelized	
	rate necessary to reflect an updated hot water load forecast as well as the	
	recovery mechanisms for the RDDA and Variance Deferral Account.	
41.	Creative Energy is directed to add any 2020 forecast variances related to the	50
	Variance Deferral Account, as follows:	
	Revenue shortfalls based on differences between actual revenues received	
	and approved forecasts;	
	Variances between forecast versus actual Steam Service Rates and Fuel	
	Cost charged to NEFC;	
	Variances between forecast versus actual Distribution expenses; and	
10	Variances between forecast versus actual Income Tax expense.	Г1
42.	In addition, the Panel approves the addition of variances between the approved revenue requirements and forecast revenues calculated based on approved rates	51
	to the RDDA. Further, the Panel extends approval of the Variance Deferral Account (which otherwise ends on December 31, 2020).	
43.	The Panel approves the forecast revenue requirement of \$2,330,926 for 2020	51
т Ј.	subject to the adjustments directed below.	JT
44.	Creative Energy's request to use a hot water load of 19,162 MWh for revenue	51
		<u> </u>

45.	The Panel also approves the NEFC 2020 Forecast O&M expenses, subject to the recalculation of costs which are allocated to the NEFC using the Massachusetts Formula as approved in Subsection 3.1 of this decision.	51
46.	Accordingly, the Panel approves a 10 percent rate increase for NEFC for 2020.	51
47.	Accordingly, the Panel denies Creative Energy's request for an NEFC FCAC Rate Rider. Creative Energy is directed to provide, in its Compliance Filing, the calculation of and proposed timing, by which customers will be refunded for the interim amounts collected in 2020, plus interest, by way of bill credits.	53
48.	Accordingly, the Panel does not approve the 2015-2017 adjustments related to pension accounting discrepancies.	55
49.	The Panel directs Creative Energy to file its next two-year revenue requirements application for the Core Steam System by November 1, 2020.	57
50.	As directed in subsection 5.1, Creative Energy must file a proposal for how rates should be set for 2021 for the NEFC by October 31, 2020 and a comprehensive proposal for NEFC rate design and for setting 2022 rates by June 30, 2021.	57

Original signed by:

A. K. Fung, QC Panel Chair / Commissioner

<u>Original signed by:</u> K. A. Keilty Commissioner

<u>Original signed by:</u> E. B. Lockhart Commissioner



Suite 410, 900 Howe Street Vancouver, BC Canada V6Z 2N3 bcuc.com P: 604.660.4700 TF: 1.800.663.1385 F: 604.660.1102

ORDER NUMBER G-227-20

IN THE MATTER OF the Utilities Commission Act, RSBC 1996, Chapter 473

and

Creative Energy Vancouver Platforms Inc. 2019-2020 Revenue Requirements Application for the Core Steam System and Northeast False Creek Service Areas

BEFORE:

A. K. Fung, QC, Panel Chair K. A. Keilty, Commissioner E. B. Lockhart, Commissioner

on September 2, 2020

ORDER

WHEREAS:

- A. On December 19, 2019, Creative Energy Vancouver Platforms Inc. (Creative Energy) filed with the British Columbia Utilities Commission (BCUC) its 2019-2020 Revenue Requirements Application (RRA) for the core steam system (Core Steam System) and Northeast False Creek (NEFC) service areas (Application);
- B. In the Application, Creative Energy requests for the Core Steam System, among other things, the following approvals:
 - 1. Permanent approval to maintain rates for the Core Steam System (Steam Rates) at 2019 interim approved rates, effective January 1, 2019;
 - 2. Interim and permanent approval of an increase in Steam Rates of 4.2 percent, effective January 1, 2020; and
 - 3. Interim and permanent approval to establish a Deferral Account Rate Rider (DARR) to recover specific non-rate base regulatory and pension-related deferred costs from Core Steam System customers over a two-year amortization period, commencing January 1, 2020;
- C. Creative Energy also requests for the NEFC service area, among other things, the following approvals:
 - (i) Permanent approval to maintain NEFC's fixed and variable rates (Hot Water Rates) at 2019 interim approved rates, effective January 1, 2019;
 - (ii) Interim and permanent approval of an increase in Hot Water Rates of 3.7 percent, effective January 1, 2020; and

- (iii) Interim and permanent approval to charge a Fuel Cost Adjustment Charge (FCAC) Rate Rider to NEFC customers to recover the existing FCAC Rate Rider in effect for Core Steam System customers, effective January 1, 2020 and ending February 28, 2021;
- D. By Order G-7-20A, the BCUC approved the following on an interim and refundable basis, effective January 1, 2020: a 4.2 percent increase in Steam Rates; a 3.7 percent increase in Hot Water Rates; a DARR of \$0.29 per thousand pounds of steam for Core Steam System customers; and a FCAC Rate Rider of \$16.15 per megawatt hour for NEFC customers;
- E. By Orders G-7-20A, G-29-20 and G-103-20, the BCUC established and amended the regulatory timetable for the review of the Application, which included Creative Energy filing an evidentiary update and a proposal for the potential recovery of the balance in its Fuel Switch Study and Long term Resource Plan (LTRP) Deferral Account, two rounds of information requests (IRs) by the BCUC and interveners, and the filing of written final and reply arguments;
- F. The following interveners registered in the proceeding;
 - FortisBC Alternative Energy Services Inc.; and
 - Commercial Energy Consumers Association of British Columbia;
- G. The BCUC has reviewed the Application, the evidence and submissions by all parties in this proceeding and makes the following determinations.

NOW THEREFORE pursuant to sections 58 to 61 of the *Utilities Commission Act*, and for the reasons provided in the decision issued concurrently (Decision) with this Order, the BCUC orders as follows:

- 1. Creative Energy's 2019 interim Steam Rates for the Core Steam System are approved as permanent, effective January 1, 2019, incorporating the following determinations set out in Section 2 of the Decision:
 - a) The continuation of the non-rate base Third Party Regulatory Cost Deferral Account (TPRCDA) is approved, including the following:
 - i. The addition of \$207,659 plus interest related to 2018 forecast variances;
 - ii. The addition of \$177,340 plus interest related to 2019 forecast variances; and
 - iii. The amortization of 50 percent of the 2018 TPRCDA balance plus interest for 2019;
 - b) The continuation of the Pension Expense Deferral Account (PEDA) is approved, including the following:
 - i. The addition of \$377,966 plus interest related to 2018 forecast variances;
 - ii. The addition of \$502,432 plus interest related to 2019 forecast variances; and
 - iii. The offset of 2019 pension remeasurement gains against 2018 pension remeasurement losses; and
 - iv. The amortization of the 2018 current service cost variance of \$25,735 plus interest for 2019.

- c) Creative Energy is directed to reduce the allowed return for 2019 by \$14,862 based on a mid-year After-tax Regulatory Pension Asset Account balance of \$362,670.
- 2. Creative Energy is approved to include NEFC in its Massachusetts Formula cost allocation, commencing in 2020.
- 3. Creative Energy is approved to use a three-factor Massachusetts Formula, commencing in 2020, to include: the average gross book value of capital assets or property, plant and equipment; salaries or direct labour expenses; and operating revenues.
- 4. Creative Energy is approved to establish 2020 Steam Rates on a permanent basis, effective January 1, 2020, using a 2020 forecast revenue requirement of \$9,086,328, subject to the following determinations set out in Section 3 of the Decision:
 - a) The 2020 steam load forecast of 1,140,634 thousand pounds of steam is approved;
 - b) Recalculate the Massachusetts Formula-allocated costs for the Core Steam System;
 - c) The 2020 Forecast for outside services costs of \$105,466 is approved;
 - d) The 2020 Forecast for steam production overtime expenses is approved;
 - e) The request for permanent approval of the proposed DARR mechanism is denied;
 - f) The continuation of the non-rate base TPRCDA for 2020 is approved, including the following:
 - i. The TPRCDA must recognize amortization of \$281,170 plus interest to be amortized over a one-year period in 2020; and
 - ii. The amortization of the TPRCDA is to be recorded in the revenue requirements for 2020 Steam Rates.
 - g) The continuation of the PEDA for 2020 is approved, including the following:
 - i. The PEDA must recognize the amortization of \$98,732 plus interest to be amortized over a one-year period in 2020; and
 - ii. The amortization of the PEDA is to be recorded in the revenue requirements for 2020 Steam Rates.
 - h) Creative Energy is directed to reduce the allowed return for 2020 by \$21,503 based on a mid-year After-tax Regulatory Pension Asset Account balance of \$268,202.
- Creative Energy is approved to establish a non-rate base Water Cost Deferral Account (WCDA) to capture forecast variances for the 2020 test year only. The WCDA is to be amortized over one-year commencing January 1, 2021, with a carrying cost equal to Creative Energy's short-term debt rate, unless otherwise determined by the BCUC.
- 6. Creative Energy is directed to address the continued need for the WCDA and to re-evaluate its water cost forecast methodology in its next RRA for the Core Steam System. Creative Energy must provide in that application information comparing the forecast, approved and actual historical water expenses and volumes to assist in the timely review of the application.

- Creative Energy is approved to establish a non-rate base Property Tax Deferral Account (PTDA) to capture property tax forecast variances. The PTDA is to be amortized over one year commencing January 1, 2021, with a carrying cost equal to Creative Energy's short-term debt rate on the mid-year value, unless otherwise determined by the BCUC.
- 8. Creative Energy is denied recovery of \$214,185 plus interest from Core Steam System customers, which it incurred in 2016 and recorded in the Fuel Switch Study and LTRP Deferral Account. Creative Energy is directed to write off these costs from the balance in the Fuel Switch Study and LTRP Deferral Account.
- 9. Creative Energy is approved to write off \$103,536 plus interest from the balance in the Fuel Switch Study and LTRP Deferral Account.
- 10. Creative Energy is denied its request to maintain a balance of \$417,502 plus interest in the Fuel Switch Study and LTRP Deferral Account and to defer for consideration the disposition of such costs until such time as a Certificate for Public Convenience and Necessity for a low-carbon energy project is submitted and an assessment of how much of the balance can be capitalized can be made. Creative Energy is directed to write off the remaining balance from the Fuel Switch Study and LTRP Deferral Account to the account of the shareholder and to close the deferral account thereafter.
- 11. Creative Energy is approved to establish 2019 interim Hot Water Rates for the NEFC on a permanent basis, effective January 1, 2019, incorporating the following determinations as set out Section 4 of the Decision:
 - a) The addition of a revenue deficiency of \$373,900 plus interest to the Revenue Deficiency Deferral Account (RDDA) for 2018;
 - b) The amount to be added to the Variance Deferral Account for 2018, which must be filed as part of a compliance filing to the BCUC within 30 days from the date of this Order, attached as Appendix A to this Order;
 - c) The request to transfer \$661,861 from the Variance Deferral Account to the RDDA in 2019 is denied;
 - d) The request to add the difference between the 2019 Actual revenue requirement and the revenues collected based on existing rates to the RDDA is denied;
 - e) Creative Energy is directed to recalculate the allocation of variances between the Variance Deferral Account and the RDDA for 2019 based on the 2017 approved forecast;
 - f) Creative Energy is to account for underreported fuel cost charges of \$497,728 arising from the PARQ metering issue in the Core Steam System's Fuel Cost Stabilization Account; and
 - g) Creative Energy is to account for the underreported fuel cost charges of \$497,728 arising from the PARQ metering issue in the Variance Deferral Account.
- 12. Creative Energy is approved to increase 2020 Hot Water Rates in the NEFC service area by 10 percent, effective January 1, 2020, using a 2020 forecast revenue requirement of \$2,330,926, subject to the following determinations set out in Section 5 of the Decision:
 - a) Recalculate the Massachusetts Formula-allocated costs for the NEFC;
 - b) The addition of any 2020 forecast variance related to the Variance Deferral Account from the following:

- Revenue shortfalls based on differences between actual revenues received and approved forecasts;
- ii. Variances between forecast versus actual Steam Service Rate and Fuel Cost charged to NEFC;
- iii. Variances between forecast versus actual Distribution expenses; and
- iv. Variance between forecast versus actual Income Tax expense;
- c) The addition of variances between the approved revenue requirements and forecast revenues calculated on approved rates to the RDDA;
- d) Creative Energy is approved to continue the Variance Deferral Account beyond December 31, 2020;
- e) The request to use a hot water load of 19,162 MWh for revenue requirement purposes is approved;
- f) The 2020 Forecast O&M expenses, subject to the recalculation of costs which are allocated to the NEFC using the Massachusetts Formula, as approved in directive #3 of this Order, are approved; and
- g) The request for permanent approval of the proposed NEFC FCAC Rate Rider is denied;
- 13. For the NEFC, Creative Energy is directed to do the following:
 - a) File a proposal for how rates should be set for 2021 by October 31, 2020; and
 - b) File a comprehensive proposal for an NEFC rate design and for setting 2022 rates by June 30, 2021.
- 14. Creative Energy is directed to file a two-year RRA for the Core Steam System by November 1, 2020, which must include:
 - a) Information detailing how other BC and Canadian regulated utilities treat pension remeasurement gains and losses for ratemaking purposes, as set out in Section 2 of the Decision;
- 15. Creative Energy is directed to file a compliance filing and amended tariff pages with the BCUC for endorsement within 30 days of the date of this Order in accordance with the requirements outlined in Appendix A. This compliance filing must include finalized financial schedules for the Core Steam System and NEFC, respectively, for 2019 and 2020.
- 16. Creative Energy is directed to comply with all other directives identified in the Decision and Appendix A to this Order.

DATED at the City of Vancouver, in the Province of British Columbia, this 2nd day of September 2020.

BY ORDER

Original signed by:

A. K. Fung, QC Commissioner

File 62558 | Creative Energy 2019-2020 RRA - Final Order

Creative Energy Vancouver Platforms Inc. 2019-2020 Revenue Requirements Application for the Core Steam System and Northeast False Creek Service Areas

COMPLIANCE FILING REQUIREMENTS

- Creative Energy is directed to recalculate the costs allocated to the Core Steam System and NEFC, respectively, for 2020 based on the approved three-factor Massachusetts Formula established in Directive #3 of this Order and include the rate impact for each service area.
- 2. Creative Energy is directed to provide the calculation of and proposed timing by which customers will be refunded the DARR amounts collected in 2020 on an interim basis, plus interest, by way of bill credits.
- 3. Creative Energy is directed to recalculate the 2020 Steam Rate increase in accordance with Directive #4 of this Order and include updated financial schedules for the Core Steam System.
- 4. Creative Energy is directed to file the amount to be added to the NEFC's Variance Deferral Account related to 2018.
- 5. Creative Energy is directed to recalculate the allocation of variances between the Variance Deferral Account and RDDA for 2019 based on 2017 approved forecast and to include updated financial schedules for the NEFC showing the revised amounts in these two deferral accounts reflecting all the determinations in this Decision.
- 6. Creative Energy is directed to include detailed calculations supporting the 2019 Variance Deferral Account balance, including the details of the amount attributed to the PARQ metering issue.
- 7. Creative Energy is directed to provide the calculation of and proposed timing by which customers will be refunded the NEFC FCAC Rate Riders amounts collected in 2020 on an interim basis, plus interest, by way of bill credits.

Creative Energy Vancouver Platforms Inc. 2019-2020 Revenue Requirements Application for the Core Steam System and Northeast False Creek Service Areas

LIST OF ACRONYMS

2015-2017 Core Steam	Creative Energy 2015-2017 Revenue Requirements Application
System Decision	Decision and Order G-98-15 dated June 9, 2015
2015-2017 RRA	Creative Energy 2015-2017 Revenue Requirements Application
2016-2017 Core Steam	Creative Energy 2016-2017 Revenue Requirements Application and
System and NEFC Decision	Rate Design for NEFC Hot Water Service Decision and Order G-167-16 dated November 18, 2016
2016-2017 RRA	Creative Energy 2016-2017 Revenue Requirements Application and
	Rate Design for NEFC Hot Water Service Application
2018-2022 Core Steam	Creative Energy 2018-2022 Revenue Requirements Application
System Decision	Decision and Order G-205-18 dated October 26, 2018
2018-2022 RRA	Creative Energy 2018-2022 Revenue Requirements Application
Application	2019-2020 Revenue Requirements Application for the Core Steam
	System and Northeast False Creek Service Areas
ΑΤCΟ	ATCO Gas and Pipelines v. Alberta (Utilities Commission)
BCUC	British Columbia Utilities Commission
Beatty Plant	Beatty Street Plant
Beatty-Expo CPCN	Creative Energy Application for a Certificate of Public Convenience
<i>,</i> ,	and Necessity for Beatty-Expo Plants and Reorganization
CPCN	Certificate of Public Convenience and Necessity
CEC	Commercial Energy Consumers Association of British Columbia
CHDL	Central Heat Distribution Ltd.
Compliance Filing	Compliance filing due to the BCUC within 30 days of the date of this
buorise - 11118	decision
Core Steam System	Core steam system
Creative Energy	Creative Energy Vancouver Platforms Inc.

Creative Energy Corp.	Creative Energy Platforms Canada Corporation
DARR	Deferral Account Rate Rider
FAES	FortisBC Alternative Energy Services Inc.
FBC	FortisBC Inc.
FCAC	Fuel Cost Adjustment Charge
FCSA	Fuel Cost Stabilization Account
HDD	Heating Degree Days
IR	Information Request(s)
LTRP	Long-Term Resource Plan
LTRP Application	2017 LTRP Application
M#	Thousand pounds
MW	Megawatt(s)
MWh	Megawatt hour(s)
NEFC	Northeast False Creek
O&M	Operating and Maintenance
RDDA	Revenue Deficiency Deferral Account
Reshape	Reshape Infrastructure Strategies
RFP	Request for Proposal
ROE	Return on Equity
RRA	Revenue Requirements Application
TES	Thermal Energy Systems
TES Regulatory Framework Guidelines	Thermal Energy Systems Regulatory Framework Guidelines established by Order G-127-14 and revised by Order G-27-15
TPRCDA	Third Party Regulatory Costs Deferral Account
UCA	Utilities Commission Act

IN THE MATTER OF the *Utilities Commission Act*, RSBC 1996, Chapter 473

and

Creative Energy Vancouver Platforms Inc. 2019–2020 Revenue Requirements Application for the Core Steam System and Northeast False Creek Service Areas

EXHIBIT LIST

Exhibit No.

Description

COMMISSION DOCUMENTS

A-1	1. Letter dated January 2, 2020 - Appointing the Panel for the review of Creative Energy Vancouver Platforms Inc. 2019–2020 Revenue Requirements Application for the Core Steam System and Northeast False Creek Service Areas
A-2	Exhibit removed and replaced by Exhibit A-2-1
A-2-1	Letter dated January 22, 2020 – Amended BCUC Order G-7-20A with reasons for decision establishing interim rates and regulatory timetable for the review of the Application
A-3	Letter dated February 21, 2020 – BCUC Order G-29-20 amending the regulatory timetable
A-4	Letter dated March 5, 2020 – BCUC issuing Information Request No. 1 to Creative Energy
A-5	Letter dated March 12, 2020 – BCUC issuing Information Request No. 1 for Fuel Switch Study and LTRP Deferral Account to Creative Energy
A-6	Letter dated April 21, 2020 – BCUC issuing Information Request No. 2 to Creative Energy
A-7	Letter dated May 1, 2020 – BCUC Order G-103-20 amending regulatory timetable
A-8	Letter dated May 12, 2020 – BCUC requesting information from Creative Energy regarding Final Arguments

COMMISSION STAFF DOCUMENTS

A2-1 Letter dated March 5, 2020 – BCUC staff filing Creative Energy Notes to Financial Statements dated December 31, 2018

APPLICANT DOCUMENTS

B-1	CREATIVE ENERGY VANCOUVER PLATFORMS INC. (CREATIVE ENERGY) - 2019–2020 Revenue Requirements Application for the Core Steam System and Northeast False Creek Service Areas dated December 19, 2019
B-1-1	Letter dated February 21, 2020 – Creative Energy Submitting Evidentiary Update to the Application
B-2	Letter dated February 20, 2020 – Creative Energy request extension to file proposal regarding Fuel Switch Study and LTRP Deferral Account
B-3	Letter dated February 27, 2020 – Creative Energy Submitting proposal regarding Fuel Switch Study and LTRP Deferral Account
B-4	Letter dated April 2, 2020 – Creative Energy Submitting responses to BCUC Information Request No. 1
B-5	Letter dated April 2, 2020 – Creative Energy Submitting responses to BCUC Information Request No. 1 on Fuel Switch Study and LTRP Deferral Account Proposal
B-6	Letter dated April 2, 2020 – Creative Energy Submitting responses to CEC Information Request No. 1
B-7	Letter dated April 2, 2020 – Creative Energy Submitting responses to CEC Information Request No. 1 on Fuel Switch Study and LTRP Deferral Account Proposal
B-8	Letter dated April 30, 2020 – Creative Energy Submitting an extension request to file Final Argument
B-9	Letter dated May 5, 2020 – Creative Energy Submitting responses to BCUC Information Request No. 2
B-10	Letter dated May 5, 2020 – Creative Energy Submitting responses to CEC Information Request No. 2

INTERVENER DOCUMENTS

C1-1	FORTISBC ALTERNATIVE ENERGY SERVICES INC. (FAES) – Letter dated January 23, 2020 request for Intervener Status by Grant Bierlmeier
C2-1	COMMERCIAL ENERGY CONSUMERS ASSOCIATION OF BRITISH COLUMBIA (CEC) – Letter dated February 12, 2020 request for Intervener Status by David Craig and Christopher Weafer
C2-2	Letter dated March 12, 2020 – CEC Submitting Information Request No. 1 to Creative Energy
C2-3	Letter dated March 19, 2020 – CEC Submitting Information Request No. 1 to Creative Energy on Fuel Switch Study and LTRP Deferral Account Proposal
C2-4	Letter dated April 21, 2020 – CEC Submitting Information Request 2 to Creative Energy

INTERESTED PARTY DOCUMENTS

D-1 FORTISBC ENERGY INC. (FEI) - Submission dated January 23, 2020 Request for Interested Party Status

LETTERS OF COMMENT

E-1